LARGE FILING SEPARATOR SHEET

CASE NUMBER 10-1363-GA-AGG

FILE DATE 8/15/2014

SECTION: 4 of 4

NUMBER OF PAGES: 456 154

DESCRIPTION OF DOCUMENT: Renewal Application

contract, assignment or other instrument pursuant to which the Seller has obtained any joint or sole ownership interest in or to each item of Seller Owned Intellectual Property. To the Knowledge of Seller, no third party inventions, methods, services, materials, processes or Software are included in or required to Exploit the Customer Offerings or Internal Systems. None of the Customer Offerings or Internal Systems includes "shareware," "freeware" or other Software or other material that was obtained by the Seller from third parties other than pursuant to the license agreements listed in Section 2.13(d) of the Disclosure Schedule.

- (e) Employee and Contractor Assignments. Each employee and each independent contractor of the Seller has executed a valid and binding written agreement expressly assigning to the Seller all right, title and interest in any inventions and works of authorship, whether or not patentable, invented, created, developed, conceived and/or reduced to practice during the term of such employee's employment or such independent contractor's work for the Seller, and all Intellectual Property rights therein, and has waived all moral rights therein to the extent legally permissible.
- (f) Infringement by Seller. To the Knowledge of Seller, none of the Customer Offerings, or the Exploitation thereof by the Seller or by any reseller, distributor, customer or user thereof, or any other activity of the Seller, infringes or violates, or constitutes a misappropriation of, any Intellectual Property rights of any third party. To the Knowledge of Seller, none of the Internal Systems, or the Seller's past, current or currently contemplated Exploitation thereof, or any other activity undertaken by them in connection with the Business, infringes or violates, or constitutes a misappropriation of, any Intellectual Property rights of any third party. Section 2.13(f) of the Disclosure Schedule lists any complaint, claim or notice, or threat of any of the foregoing (including any notification that a license under any patent is or may be required), received by the Seller alleging any such infringement, violation or misappropriation and any request or demand for indemnification or defense received by the Seller from any reseller, distributor, customer, user or any other third party; and the Seller has provided to the Buyer copies of all such complaints, claims, notices, requests, demands or threats, as well as any legal opinions, studies, market surveys and analyses relating to any alleged or potential infringement, violation or misappropriation.

2.14 Contracts.

- (a) Section 2.14 of the Disclosure Schedule lists the following agreements (written or oral) to which the Seller is a party as of the date of this Agreement:
- (i) any agreement (or group of related agreements) for the lease of personal property from or to third parties providing for lease payments in excess of \$10,000.00 per annum or having a remaining term longer than three months;
- (ii) any agreement (or group of related agreements) for the purchase or sale of products or for the furnishing or receipt of services (A) which calls for performance over a period of more than one year, (B) which involves more than the sum of \$10,000.00, or (C) in which the Seller has granted manufacturing rights, "most favored nation" pricing provisions or marketing or distribution rights relating to any products or territory or has agreed to purchase a minimum quantity of goods or services or has agreed to purchase goods or services exclusively from a certain party;

- (iii) any agreement concerning the establishment or operation of a partnership, joint venture or limited liability company;
- (iv) any agreement (or group of related agreements) under which it has created, incurred, assumed or guaranteed (or may create, incur, assume or guarantee) indebtedness (including capitalized lease obligations) involving more than \$10,000 or under which it has imposed (or may impose) a Security Interest on any of its assets, tangible or intangible;
- (v) any agreement for the disposition of any significant portion of the assets or business of the Seller (other than sales of products in the Ordinary Course of Business) or any agreement for the acquisition of the assets or business of any other entity (other than purchases of inventory or components in the Ordinary Course of Business);
 - (vi) any agreement concerning exclusivity or confidentiality;
 - (vii) any employment or consulting agreement;
 - (viii) any agreement involving any current or former officer, manager or member of the Seller;
- (ix) any agreement which contains any provisions requiring the Seller to indemnify any other party (excluding indemnities contained in agreements for the purchase, sale or license of products entered into in the Ordinary Course of Business);
- (x) any agreement under which the Seller is restricted from selling, licensing or otherwise distributing any of its technology or products, or providing services to, customers or potential customers or any class of customers, in any geographic area, during any period of time or any segment of the market or line of business;
- (xi) any agreement which would entitle any third party to receive a license or any other right to intellectual property of the Buyer or any of the Buyer's Affiliates following the Closing; and
- (xii) any other agreement (or group of related agreements) either involving more than \$10,000.00 or not entered into in the Ordinary Course of Business.
- (b) The Seller has delivered to the Buyer a complete and accurate copy of each agreement listed in <u>Section 2.13</u> or <u>Section 2.14</u> of the Disclosure Schedule. Except as otherwise provided in this Agreement, with respect to each agreement so listed: (i) the agreement is legal, valid, binding and enforceable and in full force and effect subject to applicable bankruptcy, insolvency, reorganization, moratorium and similar laws affecting creditors' rights and remedies generally, and subject, as to enforceability, to general principles of equity, including principles of commercial reasonableness, good faith and fair dealing (regardless of whether enforcement is sought in a proceeding at law or in equity); and (ii) neither the Seller nor, to the

Knowledge of the Seller, any other party, is in breach or violation of, or default under, any such agreement, and no event has occurred, is pending or, to the Knowledge of the Seller, is threatened, which, after the giving of notice, with lapse of time, or otherwise, would constitute a breach or default by the Seller or, to the Knowledge of the Seller, any other party under such agreement.

- 2.15 <u>Backlog</u>. To the Knowledge of Seller, all Backlog of the Seller are valid receivables and the Seller is not aware of any setoffs or counterclaims. A complete and accurate list of the Backlog was provided to Buyer by Russ Monroe on September 6, 2012 and a summary by supplier of the Backlog is included in <u>Schedule 2.15</u> of the Disclosure Schedule. The Seller has not received any written notice from an account debtor stating that any Backlog is subject to any contest, claim or setoff by such account debtor.
- 2.16 <u>Insurance</u>. Section 2.16 of the Disclosure Schedule lists each insurance policy (including fire, theft, casualty, comprehensive general liability, workers compensation, business interruption, environmental, product liability, errors and omissions, professional liability and automobile insurance policies and bond and surety arrangements) to which the Seller is a party, all of which are in full force and effect. There is no material claim pending under any such policy as to which coverage has been questioned, denied or disputed by the underwriter of such policy. All premiums due and payable under all such policies have been paid, neither the Seller nor the Buyer will be liable for retroactive premiums or similar payments, and the Seller is otherwise in compliance in all material respects with the terms of such policies. To the Knowledge of the Seller there is no threatened termination of, or premium increase with respect to, any such policy. To the Knowledge of Seller, Seller's insurance is adequate in terms of scope and coverage to pay all claims arising or made prior to the Closing.
- 2.17 <u>Litigation</u>. Except as set forth in <u>Section 2.17 of the Disclosure Schedule</u>, there is no Legal Proceeding which is pending or has been threatened in writing against the Seller which (a) seeks either damages or equitable relief in any way relating to the Seller or its business or (b) in any manner challenges or seeks to prevent, enjoin, alter or delay the transactions contemplated by this Agreement. There are no judgments, orders or decrees outstanding against the Seller.
- 2.18 <u>Warranties</u>. No product or service manufactured, sold, leased, licensed or delivered by the Seller is subject to any guaranty, warranty, right of return, right of credit or other indemnity other than third-party manufacturers' warranties for which the Seller has no liability.

2.19 Employees.

(a) Section 2.19 of the Disclosure Schedule contains a list of all employees of the Seller, along with the position and the annual rate of compensation of each such person. Each current or past employee of the Seller has entered into a confidentiality and assignment of inventions agreement with the Seller, a copy or form of which has previously been delivered to the Buyer. Section 2.19 of the Disclosure Schedule contains a list of all employees of the Seller who are a party to a non-competition agreement with the Seller; copies of such agreements have previously been delivered to the Buyer. Section 2.19 of the Disclosure Schedule contains a list of all employees of the Seller who are not citizens of the United States. To the Knowledge of Seller,

no key employee or group of employees has any plans to terminate employment with the Seller (other than for the purpose of accepting employment with the Buyer following the Closing) or not to accept employment with the Buyer. Seller is in compliance with all applicable laws relating to the hiring and employment of employees.

(b) The Seller is not a party to or bound by any collective bargaining agreement, nor has it experienced any strikes, grievances, claims of unfair labor practices or other collective bargaining disputes. To the Knowledge of Seller, there is no organizational effort made or threatened, either currently or within the past two years, by or on behalf of any labor union with respect to employees of the Seller.

2.20 Employee Benefits.

- (a) Section 2.20(a) of the Disclosure Schedule contains a complete and accurate list of all Seller Plans. Complete and accurate copies of (i) all Seller Plans which have been reduced to writing, (ii) written summaries of all unwritten Seller Plans, (iii) all related trust agreements, insurance contracts and summary plan descriptions, and (iv) all annual reports filed on IRS Form 5500, 5500C or 5500R and (for all funded plans) all plan financial statements for the last five plan years for Seller Plan, have been delivered to the Buyer.
- (b) Each Seller Plan has been administered in all material respects in accordance with its terms and Seller and the ERISA Affiliates has in all material respects met its obligations with respect to each Seller Plan and has made all required contributions thereto. The Seller, each ERISA Affiliate and each Seller Plan are in compliance in all material respects with the currently applicable provisions of ERISA and the Code and the regulations thereunder (including Section 4980 B of the Code, Subtitle K, Chapter 100 of the Code and Sections 601 through 608 and Section 701 et seq. of ERISA). All filings and reports as to each Seller Plan required to have been submitted to the Internal Revenue Service or to the United States Department of Labor have been duly submitted. No Seller Plan has assets that include securities issued by the Seller or any ERISA Affiliate.
- (c) There are no Legal Proceedings (except claims for benefits payable in the normal operation of the Seller Plans and proceedings with respect to qualified domestic relations orders) against or involving any Seller Plan or asserting any rights or claims to benefits under any Seller Plan that could give rise to any material liability.
- (d) All the Seller Plans that are intended to be qualified under Section 401(a) of the Code have received determination letters from the Internal Revenue Service to the effect that such Seller Plans are qualified and the plans and the trusts related thereto are exempt from federal income taxes under Sections 401(a) and 501(a), respectively, of the Code, no such determination letter has been revoked and revocation has not been threatened, and no such Seller Plan has been amended since the date of its most recent determination letter or application therefor in any respect, and no act or omission has occurred, that would adversely affect its qualification or materially increase its cost. Each Seller Plan which is required to satisfy Section 401(k)(3) or Section 401(m)(2) of the Code has been tested for compliance with, and satisfies the requirements of Section 401(k)(3) and Section 401(m)(2) of the Code for each plan year ending prior to the Closing Date.

- (e) Neither the Seller nor any ERISA Affiliate has ever maintained an Employee Benefit Plan subject to Section 412 of the Code or Title IV of ERISA.
- (f) At no time has the Seller or any ERISA Affiliate been obligated to contribute to any "multiemployer plan" (as defined in Section 4001(a)(3) of ERISA).
- (g) There are no unfunded obligations under any Seller Plan providing benefits after termination of employment to any employee of the Seller (or to any beneficiary of any such employee), including but not limited to retiree health coverage and deferred compensation, but excluding continuation of health coverage required to be continued under Section 4980B of the Code or other applicable law and insurance conversion privileges under state law. The assets of each Seller Plan which is funded are reported at their fair market value on the books and records of such Seller Plan.
- (h) No act or omission has occurred and no condition exists with respect to any Seller Plan that would subject the Seller or any ERISA Affiliate to (i) any material fine, penalty, tax or liability of any kind imposed under ERISA or the Code or (ii) any contractual indemnification or contribution obligation protecting any fiduciary, insurer or service provider with respect to any Seller Plan.
- (i) Each Seller Plan is funded by, associated with or related to a "voluntary employee's beneficiary association" within the meaning of Section 501(c)(9) of the Code.
- (j) No Seller Plan is amendable and terminable unilaterally by the Seller at any time without liability or expense to the Seller or such Seller Plan as a result thereof (other than for benefits accrued through the date of termination or amendment and reasonable administrative expenses related thereto) and, no Seller Plan, plan documentation or agreement, summary plan description or other written communication distributed generally to employees by its terms prohibits the Seller from amending or terminating any such Seller Plan.
- (k) Section 2.20(k) of the Disclosure Schedule discloses each: (i) agreement with any member, director, executive officer or other key employee of the Seller (A) the benefits of which are contingent, or the terms of which are altered, upon the occurrence of a transaction involving the Seller of the nature of any of the transactions contemplated by this Agreement, (B) providing any term of employment or compensation guarantee or (C) providing severance benefits or other benefits after the termination of employment of such director, executive officer or key employee; (ii) agreement, plan or arrangement under which any person may receive payments from the Seller that may be subject to the tax imposed by Section 4999 of the Code or included in the determination of such person's "parachute payment" under Section 280G of the Code; and (iii) agreement or plan binding the Seller, including any stock option plan, stock appreciation right plan, restricted stock plan, stock purchase plan, severance benefit plan or Seller Plan, any of the benefits of which will be increased, or the vesting of the benefits of which will be calculated on the basis of any of the transactions contemplated by this Agreement or the value of any of the benefits of which will be calculated on the basis of any of the transactions contemplated by this Agreement.

- (1) Section 2.20(1) of the Disclosure Schedule sets forth the policy of the Seller with respect to accrued vacation, accrued sick time and earned time off and the amount of such liabilities as of the date hereof.
- (m) Each Seller Plan that is a "nonqualified deferred compensation plan" (as defined in Code Section 409A(d)(1)) has been operated since January 1, 2005 in good faith compliance with Code Section 409A and IRS Notice 2005-1. No Seller Plan that is a "nonqualified deferred compensation plan" has been materially modified (as determined under Notice 2005-1) after October 3, 2004. No event has occurred that would be treated by Code Section 409A(b) as a transfer of property for purposes of Code Section 83. No stock option or equity unit option granted under any Seller Plan has an exercise price that has been or may be less than the fair market value of the underlying stock or equity units (as the case may be) as of the date such option was granted or has any feature for the deferral of compensation other than the deferral of recognition of income until the later of exercise or disposition of such option.

2.21 Environmental Matters.

- (a) The Seller has complied with all applicable Environmental Laws. There is no pending nor, to the Knowledge of Seller, threatened civil or criminal litigation, written notice of violation, formal administrative proceeding, or investigation, inquiry or information request by any Governmental Entity, relating to any Environmental Law involving the Seller.
- (b) To the Knowledge of Seller, the Seller has no liabilities or obligations arising from the release of any Materials of Environmental Concern into the environment.
- (c) The Seller is not a party to or bound by any court order, administrative order, consent order or other agreement with any Governmental Entity entered into in connection with any legal obligation or liability arising under any Environmental Law.
- 2.22 <u>Legal Compliance</u>. The Seller has, at all times, conducted its business in compliance with each applicable law (including rules and regulations thereunder) of any federal, state, local or foreign government, or any Governmental Entity except where such non-compliance would not reasonably be expected to have a Seller Material Adverse Effect. The Seller has not received any notice or communication from any Governmental Entity alleging noncompliance with any applicable law, rule or regulation except where such non-compliance would not reasonably be expected to have a Seller Material Adverse Effect.
- 2.23 <u>Customers and Suppliers</u>. <u>Section 2.23</u> of the Disclosure Schedule sets forth a list of (a) each customer that accounted for more than 1% of the consolidated revenues of the Seller during the last full fiscal year or the interim period through the Most Recent Balance Sheet Date and the amount of revenues accounted for by such customer during each such period and (b) each supplier that is the sole supplier of any significant product or service to the Seller. No such customer or supplier has indicated within the past year that it will stop, or decrease the rate of, buying products or supplying products, as applicable, to the Seller.
- 2.24 Permits . Section 2.24 of the Disclosure Schedule sets forth a list of all Permits issued to or held by the Seller. To the Knowledge of Seller, each such Permit is in full force and effect; the Seller is in compliance with the terms of each such Permit except where such non-compliance

would not reasonably be expected to have a Seller Material Adverse Effect and, to the Knowledge of Seller, no suspension or cancellation of such Permit is threatened and there is no basis for believing that such Permit will not be renewable upon expiration.

- 2.25 Certain Business Relationships With Affiliates. Seller has no Affiliates.
- 2.26 <u>Brokers' Fees</u>. The Seller has no liability or obligation to pay any fees or commissions to any broker, finder or agent with respect to the transactions contemplated by this Agreement.
- 2.27 <u>Books and Records</u>. The books and records of the Seller accurately reflect the assets, liabilities, business, financial condition and results of operations of the Seller and have been maintained in accordance with reasonable business and bookkeeping practices. <u>Section 2.27</u> of the Disclosure Schedule contains a list of all bank accounts and safe deposit boxes of the Seller and the names of persons having signature authority with respect thereto or access thereto.
 - 2.28 Left Intentionally Blank.
- 2.29 Controls and Procedures. The Seller maintains accurate books and records reflecting its assets and liabilities and maintains proper and adequate internal control over financial reporting which provide assurance that (i) transactions are executed with management's authorization, (ii) transactions are recorded as necessary to permit preparation of the consolidated financial statements of the Seller and to maintain accountability for the Seller's consolidated assets, (iii) access to assets of the Seller is permitted only in accordance with management's authorization, (iv) the reporting of assets of the Seller is compared with existing assets at regular intervals and (v) accounts, notes and other receivables and inventory were recorded accurately, and proper and adequate procedures are implemented to effect the collection thereof on a current and timely basis.
- 2.30 Government Contracts. The Seller has not been suspended or debarred from bidding on contracts or subcontracts with any Governmental Entity; to the Knowledge of Seller, no such suspension or debarment has been threatened or initiated; and to the Knowledge of Seller, the consummation of the transactions contemplated by this Agreement will not result in any such suspension or debarment of the Seller or the Buyer (assuming that no such suspension or debarment will result solely from the identity of the Buyer). The Seller has not been or is now being audited or investigated by the United States Government Accounting Office, the United States Department of Defense or any of its agencies, the Defense Contract Audit Agency, the contracting or auditing function of any Governmental Entity with which it is contracting, the United States Department of Justice, the Inspector General of the United States Governmental Entity, or any prime contractor with a Governmental Entity; nor, to the Knowledge of Seller, has any such audit or investigation been threatened. To the Knowledge of Seller, there is no valid basis for (i) the suspension or debarment of the Seller from bidding on contracts or subcontracts with any Governmental Entity or (ii) any claim (including any claim for return of funds to the Government) pursuant to an audit or investigation by any of the entities named in the foregoing sentence. The Seller has no agreements, contracts or commitments which require it to obtain or maintain a security clearance with any Governmental Entity.

2.31 Securities Laws.

- (a) The Seller and the Members have been furnished all of the materials relating to the Buyer, and its payment of the Purchase Price, that have been requested and each of them has been afforded an opportunity to ask questions of, and receive answers from, management of the Buyer in connection with the payment of the Purchase Price. The Seller and the Members have not been furnished with any oral or written representation in connection with the payment of the Purchase Price by or on behalf of the Buyer that it has relied on that is not contained in this Agreement.
- (b) Each of the Seller and the Members: (i) is an "accredited investor" as defined in Rule 501 of Regulation D under the Securities Act of 1933, as amended; (ii) has obtained, in its judgment, sufficient information to evaluate the merits and risks of the payment of the Earnout with securities of the Buyer; (iii) has sufficient knowledge and experience in financial and business matters to evaluate the merits and risks associated with such payment of the Earnout with securities of the Buyer and to make an informed investment decision with respect thereto, and (iv) has consulted with his or its own advisors with respect to the receipt of securities as part of the Earnout.
- (c) The securities being acquired hereunder are being acquired for each of the Seller and the Members' own account for investment and not for the benefit or account of any other person and not with a view to, or in connection with, any unlawful resale or distribution thereof. Each of the Seller and the Members fully understands and agrees that it must bear the economic risk of the investment in securities received hereunder for an indefinite period of time because, among other reasons, such securities received hereunder have not been registered under the Securities Act of 1933, as amended or under the securities laws of any states, and, therefore, the securities are "restricted securities" and cannot be resold, pledged, assigned or otherwise disposed of unless they are subsequently registered under the Securities Act of 1933, amended and under the applicable securities laws of such states or an exemption from such registration is otherwise available. Each of the Seller and the Members understands that the Buyer is not under any obligation to register such securities on the Seller and the Members' behalf or to assist such Seller and the Members in complying with any exemption from registration under the Securities Act or applicable state securities laws.
- (d) Each of the Seller and the Members intends that the applicable state securities law will apply to its receipt of the securities hereunder. Each of the Seller and the Members meets all suitability standards imposed by the state securities laws relating to the receipt of the securities as part of the Earnout hereunder without registering any of the Buyer's securities under the securities laws of such state.
- (e) Buyer shall assist Seller with an opinion of counsel with respect to any transfer of the shares if there is no material question as to the availability of Rule 144 promulgated under the Securities Act of 1933, as amended.

ARTICLE III

REPRESENTATIONS AND WARRANTIES OF THE BUYER

The Buyer represents and warrants to the Seller that the statements contained in this Article III are true and correct as of the date of this Agreement.

- 3.1 <u>Organization and Corporate Power</u>. The Buyer is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware. The Buyer has all requisite corporate power and authority to carry on the businesses in which it is engaged and to own and use the properties owned and used by it.
- 3.2 <u>Authorization of the Transaction</u>. The Buyer has all requisite power and authority to execute and deliver this Agreement and the Ancillary Agreements and to perform its obligations hereunder and thereunder and all other agreements contemplated by this Agreement. The execution and delivery by the Buyer of this Agreement and the Ancillary Agreements and the consummation by the Buyer of the transactions contemplated hereby and thereby have been duly and validly authorized by all necessary corporate action on the part of the Buyer. This Agreement has been duly and validly executed and delivered by the Buyer and constitutes a valid and binding obligation of the Buyer, enforceable against it in accordance with its terms.
- 3.3 Noncontravention. Neither the execution and delivery by the Buyer of this Agreement or the Ancillary Agreements, nor the consummation by the Buyer of the transactions contemplated hereby or thereby, will (a) conflict with or violate any provision of the Certificate of Incorporation or by-laws of the Buyer, (b) require on the part of the Buyer any filing with, or permit, authorization, consent or approval of, any Governmental Entity, (c) conflict with, result in breach of, constitute (with or without due notice or lapse of time or both) a default under, result in the acceleration of obligations under, create in any party any right to terminate, modify or cancel, or require any notice, consent or waiver under, any contract or instrument to which the Buyer is a party or by which it is bound or to which any of its assets is subject, or (d) violate any order, writ, injunction, decree, statute, rule or regulation applicable to the Buyer or any of its properties or assets except, in each case of (a) through (d), where such breach or violation would not reasonably be expected to have a Buyer Material Adverse Effect.
- 3.4 Equity Consideration. The Shares which may be issued to Seller under this Agreement are and/or will be duly and validly authorized by the Buyer when and if issued to Seller and the potential issuance of any such Shares to Buyer do not and will not violate at the time of the issuance to Seller any applicable rule or regulation applicable to Buyer or by which Buyer is subject.
 - 3.5 Buyer SEC Documents; No Undisclosed Liabilities.
- (a) Buyer has filed all reports, schedules, forms, statements and other documents (including exhibits and other information incorporated therein) with the SEC required to be filed by Buyer since January 1, 2011 (such documents, the "Buyer SEC Documents"). As of their respective dates, the Buyer SEC Documents complied in all material respects with the requirements of the Securities Act or the Securities and Exchange Act of 1934, as amended, as

the case may be, and the rules and regulations of the SEC promulgated thereunder applicable to such Buyer SEC Documents, and, as of their respective dates, none of the Buyer SEC Documents contained any untrue statement of a material fact or omitted to state a material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading. The financial statements of Buyer included in the Buyer SEC Documents complied as to form in all material respects with applicable accounting requirements and the published rules and regulations of the SEC with respect thereto as of their respective dates, were prepared in accordance with GAAP (except, in the case of unaudited statements, as permitted by Form 10-Q of the SEC) applied on a consistent basis during the periods involved (except as may be indicated in the notes thereto) and fairly presented in all material respects the financial position of Buyer and its consolidated Subsidiaries as of the dates thereof and the consolidated results of their operations and cash flows for the periods then ended (subject, in the case of unaudited statements, to normal year-end audit adjustments).

(b) Except as set forth in the most recent financial statements included in the Buyer SEC Documents filed by Buyer and publicly available prior to the date of this Agreement or for liabilities incurred in connection with this Agreement or in the ordinary course of business since the date of the most recent financial statements included in the Buyer SEC Documents, Buyer has no liabilities or obligations of any nature (whether accrued, absolute, contingent or otherwise) that would be required by GAAP to be reflected in, or reserved against or otherwise described in the consolidated balance sheet of Buyer (including the notes thereto) which, individually or in the aggregate, have had or would reasonably be expected to have a Buyer Material Adverse Effect.

ARTICLE IV

CONDITIONS PRECEDENT

- 4.1 <u>Conditions to Obligations of the Buyer</u>. The obligation of the Buyer to consummate the transactions contemplated by this Agreement to be consummated at the Closing is subject to the satisfaction of the following additional conditions:
- (a) the Seller shall have obtained at its own expense (and shall have provided copies thereof to the Buyer) all of the waivers, permits, consents, approvals or other authorizations, and effected all registrations, filings and notices, which are required on the part of the Seller except with respect to the tax clearance letter issued by the Connecticut Department of Revenue Services;
- (b) the representations and warranties of the Seller set forth in Sections 2.1 (first sentence), 2.2 and 2.3 and any representations and warranties of the Seller set forth in this Agreement that are qualified as to materiality shall be true and correct in all respects, and all other representations and warranties of the Seller set forth in this Agreement shall be true and correct in all material respects, in each case as of the date of this Agreement and as of the Closing as though made as of the Closing, except to the extent such representations and warranties are specifically made as of a particular date (in which case such representations and warranties shall be true and correct as of such date);

- (c) the Seller shall have performed or complied with its agreements and covenants required to be performed or complied with under this Agreement as of or prior to the Closing;
- (d) no Legal Proceeding shall be pending or threatened; and no judgment, order, decree, stipulation or injunction shall be pending, threatened, or in effect which would (i) prevent consummation of the transactions contemplated by this Agreement, (ii) cause the transactions contemplated by this Agreement to be rescinded following consummation, or (iii) affect adversely the right of the Buyer to own, operate or control any of the Acquired Assets, or to conduct the business of the Seller as currently conducted, following the Closing;
 - (e) the Seller shall have delivered to the Buyer the Seller Certificate;
- (f) the Seller shall have delivered to the Buyer documents evidencing the release or termination of all Security Interests on the Acquired Assets, and copies of filed UCC termination statements with respect to all UCC financing statements evidencing Security Interests or written statements from the lien holders evidencing repayment in full of all outstanding principal loan amounts and interest with respect to any such loans evidencing such Security Interests;
 - (g) the Buyer shall have secured financing for the transactions contemplated herein on terms acceptable to the Buyer;
 - (h) the Buyer shall have entered into a lease of the premises at 174 South Road, Enfield, Connecticut on terms acceptable to Buyer;
- (i) Russ Monroe has entered into an employment agreement with the Buyer on terms satisfactory to each party to such agreement; and
- (j) the Buyer shall have received such other certificates and instruments (including certificates of good standing of the Seller in its jurisdiction of organization and the various foreign jurisdictions in which it is qualified, certified organizational and operational documents, certificates as to the incumbency of officers and the adoption of authorizing resolutions) as it shall reasonably request in connection with the Closing.
- 4.2 <u>Conditions to Obligations of the Seller</u>. The obligation of the Seller to consummate the transactions contemplated by this Agreement to be consummated at the Closing is subject to the satisfaction of the following additional conditions:
- (a) the representations and warranties of the Buyer set forth in the first sentence of Section 3.1 and in Section 3.2 and any representations and warranties of the Buyer set forth in this Agreement that are qualified as to materiality shall be true and correct in all respects, and all other representations and warranties of the Buyer set forth in this Agreement shall be true

and correct in all material respects, in each case as of the date of this Agreement and as of the Closing as though made as of the Closing, except to the extent such representations and warranties are specifically made as of a particular date (in which case such representations and warranties shall be true and correct as of such date);

- (b) the Buyer shall have performed or complied with in all material respects its agreements and covenants required to be performed or complied with under this Agreement as of or prior to the Closing;
- (c) no Legal Proceeding shall be pending or threatened wherein an unfavorable judgment, order, decree, stipulation or injunction would (i) prevent consummation of the transactions contemplated by this Agreement or (ii) cause the transactions contemplated by this Agreement to be rescinded following consummation, and no such judgment, order, decree, stipulation or injunction shall be in effect;
 - (d) the Buyer shall have delivered to the Seller the Buyer Certificate;
- (e) the Seller shall have received such other certificates and instruments (including certificates of good standing of the Buyer in its jurisdiction of organization, certificates as to the incumbency of officers and the adoption of authorizing resolutions) as it shall reasonably request in connection with the Closing;
- (f) Russ Monroe has entered into an employment agreement with the Buyer on terms satisfactory to each party to each such employment agreement; and

ARTICLE V

POST-CLOSING COVENANTS

- 5.1 <u>Proprietary Information</u>. From and after the Closing, Seller, unless required by legal process, shall not disclose or make use of (except to pursue its rights under this Agreement or the Ancillary Agreements and with respect to providing any information under this Agreement to its attorneys, accountants and other professionals), any knowledge, information or documents of a confidential nature or not generally known to the public with respect to Acquired Assets, the Seller's business or the Buyer or its business (including the financial information, technical information or data relating to the Seller's products and names of customers of the Seller), except to the extent that such knowledge, information or documents shall have become public knowledge other than through improper disclosure by the Seller and except to the extent such information (a) is known to the public and did not become so known through any violation of legal obligation on the part of any of the Seller; (b) is required to be disclosed under the provisions of any applicable law, or by any stock exchange or similar body; or (c) is required to be disclosed by a rule or order of any court of competent jurisdiction. Notwithstanding the foregoing, Buyer acknowledges and agrees that the Seller, and its owners, have developed relationships with the customers of the business of Seller, and their owners may maintain such relationships for purposes of commercial activities that do not violate Section 5.3.
- 5.2 Solicitation and Hiring. For a period of five (5) years after the Closing Date, Seller and each of the Members shall not, either directly or indirectly (including through an

Affiliate), (a) solicit or attempt to induce any Restricted Employee to terminate his or her employment with the Buyer or any subsidiary of the Buyer or (b) hire or attempt to hire any Restricted Employee; <u>provided</u>, that this clause (b) shall not apply to any individual whose employment with the Buyer or a subsidiary of the Buyer has been terminated for a period of six months or longer.

5.3 Non-Competition.

- (a) For a period of five (5) years after the Closing Date, Seller and each of the Members shall not, either directly or indirectly as an owner, partner, officer, employee, director, investor, lender, consultant, independent contractor or otherwise, (i) design, develop, manufacture, market, sell or license any product or provide any service anywhere in the United States which is competitive with any product designed, developed (or under development), manufactured, sold or licensed or any service provided by the Seller within the three-year period prior to the Closing Date or (ii) engage anywhere in the United States in any business competitive with the business of the Seller as conducted as of the Closing Date or during the three-year period prior to the Closing Date. Notwithstanding anything contained in this Agreement to the contrary, in no event shall the Seller or any Member (i) have any obligation to ensure compliance by any other party with respect to this Section 5.3 or (ii) have any liability whatsoever for the breach of this Section 5.3 by any other party.
- (b) Seller and each of the Members agree that the duration and geographic scope of the non-competition provision set forth in this <u>Section 5.3</u> are reasonable. In the event that any court determines that the duration or the geographic scope, or both, are unreasonable and that such provision is to that extent unenforceable, the Parties agree that the provision shall remain in full force and effect for the greatest time period and in the greatest area that would not render it unenforceable. The Parties intend that this non-competition provision shall be deemed to be a series of separate covenants, one for each and every county of each and every state of the United States of America and each and every political subdivision of each and every country outside the United States of America where this provision is intended to be effective.

5.4 Tax Matters.

- (a) All transfer taxes, deed excise stamps and similar charges related to the sale of the Acquired Assets contemplated by this Agreement shall be paid by the Seller.
 - (b) The Seller shall be responsible for all Tax liabilities attributable to its business prior to and including the Closing Date.
 - (c) The Buyer shall be responsible for all Taxes attributable to the conduct of its business after the Closing Date.
- (d) The Buyer shall make available to the Seller and its representatives all records and materials reasonably required by the Seller to prepare, pursue or contest any Tax matters related to taxable periods (or portions thereof) ending on or before the Closing Date and shall provide reasonable cooperation to the Seller in such case. The Seller shall make available to the Buyer and its representatives all records and materials reasonably required by the Buyer to prepare, pursue or contest any Tax matters arising after the Closing which have factual reference to the pre-closing tax period and shall provide reasonable cooperation to the Buyer in such case.

5.5 Sharing of Data.

- (a) The Seller shall have the right for a period of seven years following the Closing Date to have reasonable access to such books, records and accounts, including financial and tax information, correspondence, production records, employment records and other records that are transferred to the Buyer pursuant to the terms of this Agreement for the limited purposes of concluding its involvement in the business conducted by the Seller prior to the Closing Date and for complying with its obligations under applicable securities, tax, environmental, employment or other laws and regulations. The Buyer shall have the right for a period of seven years following the Closing Date to have reasonable access to those books, records and accounts, including financial and accounting records (and to the extent available, the work papers of the Seller's independent accountants), tax records, correspondence, production records, employment records and other records that are retained by the Seller pursuant to the terms of this Agreement to the extent that any of the foregoing is needed by the Buyer for the purpose of conducting the business of the Seller after the Closing and complying with its obligations under applicable securities, tax, environmental, employment or other laws and regulations. Neither the Buyer nor the Seller shall destroy any such books, records or accounts retained by it without first providing the other Party with the opportunity to obtain or copy such books, records, or accounts at such other Party's expense.
- (b) Seller will cooperate with the Buyer's auditors and Seller's auditors to produce the financial information and statements necessary so that Buyer may comply with its federal, state and regulatory reporting requirements. Seller will respond to any reasonable requests made by the Buyer, the Buyer's auditors or the Seller's auditors promptly. Promptly upon request by the Buyer, the Seller shall authorize the release to the Buyer of all files pertaining to the Seller, the Acquired Assets or the business or operations of the Seller held by any federal, state, county or local authorities, agencies or instrumentalities. Seller understands that Buyer will suffer harm if it does not meet regulatory requirements regarding reporting of this transaction.
- 5.6 <u>Use of Name</u>. Seller shall not use the name Northeast Energy Partners or any name reasonably similar thereto after the Closing Date. Within 10 days following the Closing, the Seller shall amend its organizational documents and state filings, where appropriate, and other corporate records, if necessary, to comply with this provision.
- 5.7 Cooperation in Litigation. From and after the Closing Date, each Party shall fully cooperate with the other in the defense or prosecution of any litigation or proceeding already instituted or which may be instituted hereafter against or by such other Party relating to or arising out of the conduct of the business of the Seller or the Buyer prior to or after the Closing Date and related to the transaction contemplated by this Agreement (other than litigation among the Parties and/or their Affiliates arising out the transactions contemplated by this Agreement). The Party requesting such cooperation shall pay the reasonable out-of-pocket expenses incurred in providing such cooperation (including legal fees and disbursements) by the Party providing such cooperation and by its officers, directors, employees and agents.

5.8 Assignment of Supplier Agreements.

- (a) Notwithstanding anything to the contrary herein, only those Supplier Agreements that may be assigned or transferred without the consent of a third party are being assigned to the Buyer as of the Closing Date. Instead, the Seller will use its Commercially Reasonable Efforts to collect any and all amounts due to the Seller pursuant to the Supplier Agreements. The Seller will direct all suppliers making payments pursuant to the Supplier Agreements to remit payment directly to a lockbox that will be established with Silicon Valley Bank pursuant to a lockbox agreement (the "Supplier Agreement Lockbox"). The Seller agrees that it shall forward promptly to the Buyer any monies, checks or instruments received by the Seller after the Closing Date (a) pursuant to Supplier Agreements and (b) related to items earned, invoiced or to be invoiced after October 1, 2012. The Seller shall provide to the Buyer such reasonable assistance as the Buyer may request with respect to the collection of any such accounts receivable, provided the Buyer pays the reasonable out-of-pocket expenses of the Seller and its officers, directors and employees incurred in providing such assistance. Effective as of the Closing, the Seller hereby grants to the Buyer a power of attorney for the sole purpose of endorsing and cashing any checks or instruments payable or endorsed to the Seller or its order which are received by the Buyer and which relate to accounts receivable purchased by the Buyer from the Seller.
- (b) If and to the extent that after the Closing Date, Buyer collects revenue pursuant to Any Supplier Agreement that relates to an invoice generated prior to October 1, 2012, Buyer shall remit such amounts to Seller within thirty (30) days after the end of the calendar month in which such amounts were received along with such schedules it has prepared to support the determination of such amount. Seller shall have the right at its expense to examine those financial records of Buyer as may be reasonably necessary to confirm the accuracy of the schedules provided to Seller and any calculations or payments made by Buyer hereunder. The Buyer will afford Seller access to the records during normal business hours, upon reasonable advance notice given by the Seller, and subject to such reasonable limitations as the Buyer may impose to delete competitively sensitive or privileged information.
- 5.9 Employees. Effective as of the Closing, the Seller shall terminate the employment of each of its employees. The Buyer shall be permitted to offer employment to each such employee, terminable at the will of the Buyer. The Seller hereby consents to the hiring of any such employees by the Buyer and waives, with respect to the employment by the Buyer of such employees, any claims or rights the Seller may have against the Buyer or any such employee under any non-competition, confidentiality or employment agreement.

5.10 Third-Party Notices and Consents.

- (a) The Seller shall use its Reasonable Best Efforts to obtain, at its expense, all such waivers, consents or approvals from third parties, and to give all such notices to third parties, as listed or are required to be listed in the Disclosure Schedule.
- (b) If (i) any of the Assigned Contracts or other assets or rights constituting Acquired Assets may not be assigned and transferred by the Seller to the Buyer (as a result of either the provisions thereof or applicable law) without the consent or approval of a third party, (ii) the Seller, after using its Reasonable Best Efforts, is unable to obtain such consent or approval

prior to the Closing and (iii) the Closing occurs nevertheless, then (A) such Assigned Contracts and/or other assets or rights shall not be assigned and transferred by the Seller to the Buyer at the Closing and the Buyer shall not assume the Seller's liabilities with respect thereto at the Closing, (B) the Seller shall continue to use its Reasonable Best Efforts to obtain the necessary consent or approval as soon as practicable after the Closing, and (C) upon the obtaining of such consent or approval, the Buyer and the Seller shall execute such further instruments of conveyance (in substantially the form executed at the Closing) as may be necessary to assign and transfer such Assigned Contracts and/or other assets or rights (and the associated liabilities and obligations of the Seller) to the Buyer.

5.11 Curb Energy and Rate Droppers Asset Purchase Agreement

The Buyer and Curb Energy, LLC and Rate Droppers, LLC shall negotiate in good faith to enter into an asset purchase agreement acceptable to Buyer, Seller and Curb Energy, LLC and Rate Droppers, LLC within thirty (30) days of the Closing Date on mutually agreeable terms.

5.12 Insurance Coverage

Seller shall obtain, at Seller's expense, within thirty (30) days of the Closing Date a policy for tail coverage for its existing employee practices insurance policy for a period of twelve (12) months from the Closing Date.

5.13 Connecticut Light and Power

Seller will continue to pay the electricity invoice from Connecticut Light and Power for the Leased Premises, and Buyer will reimburse Buyer for such amounts within ten (10) days of receipt of an invoice from Seller.

ARTICLE VI

INDEMNIFICATION

- 6.1 <u>Indemnification by the Seller</u>. The Seller and its Members shall jointly and severally indemnify Buyer (and its officers, directors and affiliates) in respect of, and hold the Buyer (and its officers, directors and affiliates) harmless against, Damages incurred or suffered by the Buyer or any Affiliate thereof resulting from, relating to or constituting:
- (a) any breach, as of the date of this Agreement, of any representation or warranty of any Seller contained in this Agreement, any Ancillary Agreement or any other agreement or instrument furnished by Seller to the Buyer pursuant to this Agreement;
- (b) any failure to perform any covenant or agreement of Seller contained in this Agreement, any Ancillary Agreement or any agreement or instrument furnished by Seller to the Buyer pursuant to this Agreement;
 - (c) any Retained Liabilities; and/or

- (d) the operation of its business or the Acquired Assets prior to the Closing Date.
- 6.2 <u>Indemnification by the Buyer</u>. The Buyer shall indemnify the Seller in respect of, and hold it harmless against, any and all Damages incurred or suffered by the Seller resulting from, relating to or constituting:
- (a) any breach, as of the date of this Agreement, of any representation or warranty of the Buyer contained in this Agreement, any Ancillary Agreement or any other agreement or instrument furnished by the Buyer to the Seller pursuant to this Agreement;
- (b) any failure to perform any covenant or agreement of the Buyer contained in this Agreement, any Ancillary Agreement or any other agreement or instrument furnished by the Buyer to the Seller pursuant to this Agreement;
 - (c) any Assumed Liabilities; and/or
 - (d) operation of its business or the Acquired Assets after the Closing Date.

6.3 Indemnification Procedure.

(a) In the event that any Legal Proceedings shall be instituted or that any claim or demand ("Claim") shall be asserted by any Person in respect of which payment may be sought under Section 6.1 and 6.2 hereof (regardless of the limitations set forth in Section 6.4), the Indemnified Party shall reasonably and promptly cause written notice of the assertion of any Claim of which it has knowledge which is covered by this indemnity to be forwarded to the Indemnifying Party. The Indemnifying Party shall have the right, at its sole option and expense, to be represented by counsel of its choice, which must be reasonably satisfactory to the Indemnified Party, and to defend against, negotiate, settle or otherwise deal with any Claim which relates to any Damages indemnified against hereunder. If the Indemnifying Party elects to defend against, negotiate, settle or otherwise deal with any Claim which relates to any Damages indemnified against hereunder, it shall within thirty (30) days (or sooner, if the nature of the Claim so requires) notify the Indemnified Party of its intent to do so. If the Indemnifying Party elects not to defend against, negotiate, settle or otherwise deal with any Claim which relates to any Damages indemnified against hereunder, the Indemnified Party may defend against, negotiate, settle or otherwise deal with such Claim. If the Indemnifying Party shall assume the defense of any Claim, the Indemnified Party may participate, at his or its own expense, in the defense of such Claim. The parties hereto agree to cooperate fully with each other in connection with the defense, negotiation or settlement of any such Claim. Notwithstanding anything in this Section 6.3 to the contrary, neither the Indemnifying Party nor the Indemnified Party shall, without the written consent of the other party, settle or compromise any indemnifiable Claim or permit a default or consent to entry of any judgment unless the claimant and such party provide to such other party an unqualified release from all liability in respect of the indemnifiable Claim. Notwithstanding the foregoing, if a settlement offer is made by the applicable third party claimant, and the Indemnifying Party notifies the Indemnified Party in writing of the Indemnifying Party's willingness to accept the settlement offer and, subject to the applicable limitations of Section 6.4, pay the amount called for by such offer, and the Indemnified Party

declines to accept such offer, the Indemnified Party may continue to contest such indemnifiable Claim, free of any participation by the Indemnifying Party, and the amount of any ultimate liability with respect to such Indemnifiable Claim that the Indemnifying Party has an obligation to pay hereunder shall be limited to the amount of the settlement offer.

- (b) After any final judgment or award shall have been rendered by a Governmental Body of competent jurisdiction and the expiration of the time in which to appeal therefrom, or a settlement shall have been consummated, or the Indemnified Party and the Indemnifying Party shall have arrived at a mutually binding agreement with respect to a Claim hereunder, the Indemnified Party shall forward to the Indemnifying Party notice of any sums due and owing by the Indemnifying Party pursuant to this Agreement with respect to such matter.
- 6.4 <u>Survival of Representations and Warranties</u>. The representations and warranties of the Parties shall (a) survive Closing and (b) shall expire eighteen (18) months after the Closing Date, except that (i) the representations and warranties set forth in <u>Sections 2.1 and 2.2</u> shall survive the Closing without limitation, and (ii) the representations and warranties set forth in <u>Section 2.9, 2.10(a), 2.20 and 2.21</u> shall survive until 30 days following the expiration of the statute of limitations applicable to the matters referred to within each such section. If an Indemnified Party delivers to an Indemnifying Party, before expiration of a representation or warranty, a notice of a Claim based upon a breach of such representation or warranty, then the applicable representation or warranty shall survive until, but only for purposes of, the resolution of any claims arising from or related to the matter covered by such notice. The rights to indemnification set forth in this Article VI shall not be affected by (i) any investigation conducted by or on behalf of an Indemnified Party or any knowledge acquired (or capable of being acquired) by an Indemnified Party, whether before or after the date of this Agreement, with respect to the inaccuracy or noncompliance with any representation, warranty, covenant or obligation which is the subject of indemnification hereunder or (ii) any waiver by an Indemnified Party of any closing condition relating to the accuracy of any representations and warranties or the performance of or compliance with agreements and covenants.

6.5 Exclusive Remedy and Set-Off.

- (a) Except with respect to claims based on fraud, after the Closing, the rights of the Indemnified Parties under this Article VI shall be the exclusive remedy of the Indemnified Parties with respect to claims resulting from or relating to any misrepresentation, breach of warranty or failure to perform any covenant or agreement contained in this Agreement.
- (b) The Seller and the Buyer hereby agree that should the Buyer be entitled to indemnification under Section 6.1 of this Agreement, the Buyer shall first set off the outstanding indebtedness evidence by the Seller Note and the Earnout against any and all obligations of the Seller and the Members arising under Section 6.1 of this Agreement. The Buyer shall not attempt to collect any Damages directly from the Seller and/or any of the Members unless and until it has exhausted all set off rights with respect to the Seller Note and the Earnout.
 - 6.6. Certain Limitations on Indemnification.

- (a) Notwithstanding the provisions of this Article VI, neither Seller nor Buyer shall have any indemnification obligations for Damages pursuant to Section 6.1 or 6.2, unless the aggregate amount of all such Damages finally determined for which such party would be liable, but for this paragraph (a) exceeds \$50,000 (the "Basket"), and in such event, the Indemnifying Party shall then be required to pay the total amount of such Damages in excess of the Basket, and subject to the Cap, provided that such limitation shall not apply to Seller's breach of Section 2.9.
- (b) None of the Seller Parties or the Buyer shall be required to indemnify any Person for Damages pursuant to <u>Section 6.1</u> or <u>6.2</u> for an aggregate amount of all such Damages exceeding \$2,500,000.00 provided that such limitation shall not apply to claims based on fraud. (the "Cap").
- 6.7 <u>Treatment of Indemnity Payments</u>. Any payments made to an Indemnified Party pursuant to this Article VI shall be treated as an adjustment to the Purchase Price for tax purposes.

ARTICLE VIII

DEFINITIONS

For purposes of this Agreement, each of the following terms shall have the meaning set forth below.

- "Acquired Assets" shall mean all of the assets, properties and rights of the Seller existing as of the Closing, except to the extent included in the definition of Excluded Assets and to the extent assignable, including:
- (a) all trade and other accounts receivable that are payable to the Seller, and all rights to unbilled amounts for products delivered or services provided, together with any security held by the Seller for the payment thereof;
- (b) all computers, machinery, equipment, tools and tooling, furniture, fixtures, supplies, leasehold improvements, and other tangible personal property;
 - (c) all Intellectual Property, if any;
 - (d) all rights under Assigned Contracts to the extent assignable;
- (e) all claims, prepayments, deposits, refunds, causes of action, chooses in action, rights of recovery, rights of setoff and rights of recoupment;
 - (f) all Permits to the extent assignable:
- (g) all books, records, accounts, ledgers, files, documents, correspondence, lists (including customer and prospect lists), employment records, manufacturing and procedural manuals, Intellectual Property records, sales and promotional materials, studies, reports and other printed or written materials;

- (h) all insurance policies of the Seller, as well as all proceeds which may be payable thereunder; and
- (i) all rights of the Seller in and with respect to the assets associated with its Employee Benefit Plans.
- "Affiliate" shall mean any affiliate, as defined in Rule 12b-2 under the Securities Exchange Act of 1934.
- "Agreement" shall have the meaning set forth in the first paragraph of this Agreement.
- "Ancillary Agreements" shall mean the Bill of Sale and Assignment and Assumption Agreement referred to in Section 1.5.
- "Assigned Contracts" shall mean the agreements listed on Section 2.14 of the Disclosure Schedule (except for those agreements that are specifically indicated on Section 2.14 of the Disclosure Schedule as not being deemed Assigned Contracts or are not capable of being assigned).
 - "Assignment and Assumption Agreement" shall have the meaning set forth in Section 1.5.
- "Assumed Liabilities" shall mean (a) all obligations of the Seller arising after the Closing under the Assigned Contracts, other than any liabilities for any breach, act or omission by the Seller prior to the Closing under any Assigned Contract, (b) any liability for Taxes in accordance with Section 5.4(d), (c) for any commission and/or bonus due to any employee of the Seller for deals booked and eligible for commission for the period from July 1, 2012 to September 30, 2012 per Seller's Sales Incentive Compensation ("SIC") Plan and scheduled to be paid on October 31, 2012, provided such employee is employed or engaged by Buyer after the Closing Date, and (d) for any commission and/or bonus due per the terms of Seller's SIC Plan any employee or independent contractor of the Seller subsequent to the Closing Date provided such employee or independent contractor is employed or engaged by Buyer after the Closing Date; (e) the Settlement and Release Agreement dated March 23, 2009 between Seller and Constellation NewEnergy; (f) the Settlement and Release Agreement dated January 27, 2008 between Seller and Constellation NewEnergy; (g) the Agreement dated March 22, 2010 between Seller and M & W Scoops, Inc.; and (h) any obligation set forth in Schedule 1.2
- "Backlog" shall mean estimated future commissions to be collected by Buyer from energy supply contracts between energy suppliers and energy consumers deals that have been brokered by the Buyer through September 30, 2012.
 - "Bill of Sale" shall have the meaning set forth in Section 1.5 of this Agreement.
 - "Buyer" shall have the meaning set forth in the first paragraph of this Agreement.
- "Buyer Certificate" shall mean a certificate to the effect that each of the conditions specified in clauses (a) through (c) (insofar as clause (c) relates to Legal Proceedings involving the Buyer) of Section 4.11 is satisfied in all respects.

- "Buyer Material Adverse Effect." shall mean a material adverse effect on any of the Assets or on the business, operations, property, financial condition, results of operations or cash flow of Buyer or the business of Buyer. For the avoidance of doubt, the parties agree that the terms "material", "materially" or "materiality" as used in this Agreement with an initial lower case "m" shall have their respective customary and ordinary meanings, without regard to the meaning ascribed to Buyer or Seller Material Adverse Effect.
 - "Cap" has the meaning set forth in Section 5.5(b) of this Agreement.
 - "CERCLA" shall mean the federal Comprehensive Environmental Response, Compensation and Liability Act of 1980, as amended.
- "Change of Control Transaction" shall mean (A) the liquidation, bankruptcy, merger, consolidation, dissolution or winding up of the Buyer; (B) the acquisition of the Buyer by another entity by means of any transaction or series of related transactions (including, without limitation, any reorganization, merger or consolidation) that results in the transfer of fifty percent (50%) or more of the outstanding voting power of the Buyer; (C) a sale of all or substantially all of the assets of the Buyer, (D) any Person or group of related persons for purposes of Section 13(d) of the Exchange Act (a "Group") (other than any Person or Group who owns or has the right to acquire Buyer's common stock on the date hereof) shall become the beneficial owner (within the meaning of Section 13(d) of the Exchange Act), directly or indirectly, of common stock of the Buyer representing more than 40% of the aggregate outstanding voting power of the Buyer and such Person or Group actually has the power to vote such common stock in any election, or (E) a majority of the board of directors of the Buyer shall consist of Persons who was not (i) a member of the board of directors on the Closing Date, or (ii) nominated for election or elected to the board of directors of the Buyer with the affirmative vote of a majority of the members of the board of directors on the Closing Date; that occurs prior to the expiration of the measurement period for the Earnout payment.
 - "Claim" shall have the meaning set forth in Section 6.3(a) of this Agreement.
 - "Closing" shall mean the closing of the transactions contemplated by this Agreement.
 - "Closing Date" shall mean the date of this Agreement.
 - "Code" shall mean the Internal Revenue Code of 1986, as amended.
- "Customer Offerings" shall mean (a) the services that the Seller (i) currently provides or makes available to third parties, or (ii) has provided or made available to third parties within the previous four years, or (iii) currently plans to provide or make available to third parties in the future and (b) the products (including Software and Documentation) that the Seller (i) currently develops, manufactures, markets, distributes, makes available, sells or licenses to or for third parties, or (ii) has developed, manufactured, marketed, distributed, made available, sold or licensed to or for third parties within the previous six years, or (iii) currently plans to develop, manufacture, market, distribute, make available, sell or license to or for third parties in the future. A true and complete list of all Customer Offerings is set forth in Section 2.13(c) of the Disclosure Schedule.

- "<u>Damages</u>" shall mean any and all debts, obligations and other liabilities (whether absolute, accrued, contingent, fixed or otherwise, or whether known or unknown, or due or to become due or otherwise), diminution in value, monetary damages, fines, fees, penalties, interest obligations, deficiencies, losses and expenses (including amounts paid in settlement, interest, court costs, costs of investigators, fees and expenses of attorneys, accountants, financial advisors and other experts, and other expenses of litigation).
- " <u>Disclosure Schedule</u>" shall mean the disclosure schedule provided by the Seller to the Buyer on the date hereof and attached to this Agreement.
- "<u>Dispute</u>" shall mean the dispute resulting if the Indemnifying Party in a Response disputes its liability for all or part of the Claimed Amount.
- "<u>Documentation</u>" shall mean printed, visual or electronic materials, reports, white papers, documentation, specifications, designs, flow charts, code listings, instructions, user manuals, frequently asked questions, release notes, recall notices, error logs, diagnostic reports, marketing materials, packaging, labeling, service manuals and other information describing the use, operation, installation, configuration, features, functionality, pricing, marketing or correction of a product, whether or not provided to end user.
- "EBITDA" or "Earnings before Interest, Taxes Depreciation and Amortization" shall mean income from operations, plus any depreciation and amortization expense reflected within income from operations, all applied on a consistent basis with the Financial Statements of Seller taking into consideration the methodology for calculating the Earnout set forth in Section 1.7(a).
 - " EBITDA Target" shall mean the financial metrics as specified in the table set forth in Section 1.7.
- "Employee Benefit Plan" shall mean any "employee pension benefit plan" (as defined in Section 3(2) of ERISA), any "employee welfare benefit plan" (as defined in Section 3(1) of ERISA), and any other written or oral plan, agreement or arrangement involving direct or indirect compensation, including insurance coverage, severance benefits, disability benefits, deferred compensation, bonuses, stock options, stock purchase, phantom stock, stock appreciation or other forms of incentive compensation or post-retirement compensation.
- "Environmental Law" shall mean any federal, state or local law, statute, rule, order, directive, judgment, Permit or regulation or the common law relating to the environment, occupational health and safety, or exposure of persons or property to Materials of Environmental Concern, including any statute, regulation, administrative decision or order pertaining to: (i) the presence of or the treatment, storage, disposal, generation, transportation, handling, distribution, manufacture, processing, use, import, export, labeling, recycling, registration, investigation or remediation of Materials of Environmental Concern or documentation related to the foregoing; (ii) air, water and noise pollution; (iii) groundwater and soil contamination; (iv) the release, threatened release, or accidental release into the environment, the workplace or other areas of Materials of Environmental Concern, including emissions, discharges, injections, spills, escapes or dumping of Materials of Environmental Concern; (v) transfer of interests in or control of real property which may be contaminated; (vi) community or worker right-to-know disclosures with

respect to Materials of Environmental Concern; (vii) the protection of wild life, marine life and wetlands, and endangered and threatened species; (viii) storage tanks, vessels, containers, abandoned or discarded barrels and other closed receptacles; and (ix) health and safety of employees and other persons. As used above, the term "release" shall have the meaning set forth in CERCLA.

- "ERISA" shall mean the Employee Retirement Income Security Act of 1974, as amended.
- "ERISA Affiliate" shall mean any entity which is, or at any applicable time was, a member of (1) a controlled group of corporations (as defined in Section 414(b) of the Code), (2) a group of trades or businesses under common control (as defined in Section 414(c) of the Code), or (3) an affiliated service group (as defined under Section 414(m) of the Code or the regulations under Section 414(o) of the Code), any of which includes or included the Seller.
 - "Excluded Assets" shall mean the following assets of the Seller:
- (a) the governing documents, qualifications to conduct business as a foreign limited liability company, arrangements with registered agents relating to foreign qualifications, taxpayer and other identification numbers, seals, minute books, membership or security transfer books and other documents relating to the organization and existence of the Seller as a limited liability company;
 - (b) all rights relating to refunds, recovery or recoupment of Taxes;
 - (c) any of the rights of the Seller under this Agreement or under the Ancillary Agreements;
 - (d) prepayments by Seller on insurance policies not assumed;
 - (e) those assets listed on Schedule 1.1(b) attached hereto and/or pursuant to the Disclosure Schedule.
 - (f) all cash, short term investments, deposits, bank accounts and similar assets and all motor vehicles.
- "Exploit" shall mean develop, design, test, modify, make, use, sell, have made, used and sold, import, reproduce, market, distribute, commercialize, support, maintain, correct and create derivative works of.
 - "Financial Statements" shall mean:
- (a) the statement of assets, liabilities and members' equity-income tax basis, statements of revenues, expenses and members' equity-income tax basis and statements of cash flows-income tax basis of the Seller as of the end of July 31, 2012 and for each of the years ended December 31, 2009, December 31, 2010 and December 31, 2011; and

- (b) the Most Recent Balance Sheet and the unaudited profit and loss statements for the seven (7) months ended as of the Most Recent Balance Sheet Date.
- "Force Majeure Event" shall mean a (i) fire, flood, earthquake, hurricane, elements of nature or acts of God; (ii) wars (declared and undeclared), acts of terrorism, sabotage, riots, civil disorders, rebellions or revolutions; or (iii) acts of any governmental authority with respect to any of the foregoing, and provided that such default or delay cannot reasonably be circumvented by the non-performing Party through the use of commercially reasonable alternate sources, workaround plans or other commercially reasonable means.
 - "GAAP" shall mean United States generally accepted accounting principles consistently applied.
- "Governmental Entity" shall mean any court, arbitrational tribunal, administrative agency or commission or other governmental or regulatory authority or agency.
 - "Indemnified Party" shall mean a party entitled, or seeking to assert rights, to indemnification under Article VI of this Agreement.
 - "Indemnifying Party" shall mean the party from whom indemnification is sought by the Indemnified Party.
 - "Intellectual Property" shall mean the following subsisting throughout the world:
 - (a) Patent Rights;
 - (b) Trademarks and all goodwill in the Trademarks;
- (c) copyrights, designs, data and database rights and registrations and applications for registration thereof, including moral rights of authors;
 - (d) mask works and registrations and applications for registration thereof under the laws of any jurisdiction;
- (e) inventions, invention disclosures, statutory invention registrations, trade secrets and confidential business information, know-how, manufacturing and product processes and techniques, research and development information, financial, marketing and business data, pricing and cost information, business and marketing plans and customer and supplier lists and information, whether patentable or nonpatentable, whether copyrightable or noncopyrightable and whether or not reduced to practice; and
- (f) other proprietary rights relating to any of the foregoing (including remedies against infringement thereof and rights of protection of interest therein under the laws of all jurisdictions).
- "Intellectual Property Registrations" means Patent Rights, registered Trademarks, registered copyrights and designs, mask work registrations and applications for each of the foregoing.

- "Internal Systems" shall mean the Software and Documentation and the computer, communications and network systems (both desktop and enterprise-wide), laboratory equipment, reagents, materials and test, calibration and measurement apparatus used by the Seller in its business or operations or to develop, manufacture, fabricate, assemble, provide, distribute, support, maintain or test the Customer Offerings, whether located on the premises of the Seller or hosted at a third party site. All Internal Systems that are material to the business of the Seller is listed and described in Section 2.13(c) of the Disclosure Schedule.
- "Knowledge of Seller" shall mean the (i) actual knowledge of Thomas Lockwood, Lora Monroe, John Hardy, Russell Monroe, Susan DiMare, and/or Tim Lockwood, or (ii) if a reasonable prudent individual similarly situated to the persons listed in clause (i) could reasonably be expected to discover or otherwise become aware of that fact or matter in the course of conducting a reasonably comprehensive investigation regarding the accuracy of any representation or warranty in this Agreement.
 - "Lease" shall mean any lease or sublease pursuant to which the Seller leases or subleases from another party any real property.
 - "Leased Premises" shall mean 174 South Road, Enfield, Connecticut 06082.
- "Legal Proceeding" shall mean any action, suit, proceeding, claim, arbitration or investigation before any Governmental Entity or before any arbitrator.
- "Materials of Environmental Concern" shall mean any: pollutants, contaminants or hazardous substances (as such terms are defined under CERCLA), pesticides (as such term is defined under the Federal Insecticide, Fungicide and Rodenticide Act), solid wastes and hazardous wastes (as such terms are defined under the Resource Conservation and Recovery Act), chemicals, other hazardous, radioactive or toxic materials, oil, petroleum and petroleum products (and fractions thereof), or any other material (or article containing such material) listed or subject to regulation under any law, statute, rule, regulation, order, Permit, or directive due to its potential, directly or indirectly, to harm the environment or the health of humans or other living beings.
 - " Members" shall mean Lora Monroe, John Hardy and Thomas Lockwood.
 - " Most Recent Balance Sheet" shall mean the unaudited balance sheet of the Seller as of the Most Recent Balance Sheet Date.
 - "Most Recent Balance Sheet Date" shall mean July 31, 2012.
- "NASDAQ Price" shall mean the consolidated closing bid price of XWES common stock at the time this Agreement is executed. If this Agreement is executed before 4PM Eastern Standard Time, the NASDAQ Price will be the consolidated closing bid price for the business day prior to the date this Agreement is executed. If this Agreement is executed after 4PM Eastern Standard Time, the NASDAQ Price will be the consolidated closing bid price on the date this Agreement is executed.

- "Open Source Materials" means all Software, Documentation or other material that is distributed as "free software", "open source software" or under a similar licensing or distribution model, including, but not limited to, the GNU General Public License (GPL), GNU Lesser General Public License (LGPL), Mozilla Public License (MPL), or any other license described by the Open Source Initiative as set forth on www.opensource.org.
- "Ordinary Course of Business" shall mean the ordinary course of business consistent with past custom and practice (including with respect to frequency and amount).
 - "Parties" shall mean the Buyer and/or the Seller and Party shall mean any of the Buyer or the Seller.
- "Patent Rights" shall mean all patents, patent applications, utility models, design registrations and certificates of invention and other governmental grants for the protection of inventions or industrial designs (including all related continuations, continuations-in-part, divisionals, reissues and reexaminations).
- "Permits" shall mean all permits, licenses, registrations, certificates, orders, approvals, franchises, variances and similar rights issued by or obtained from any Governmental Entity (including those issued or required under Environmental Laws and those relating to the occupancy or use of owned or leased real property).
- "Person" shall mean any individual, corporation, partnership, joint venture, association, limited liability company, joint stock company, trust, unincorporated organization, or other entity, or any governmental agency, and any officer, department, commission, board, bureau, or instrumentality thereof.
 - "Purchase Price" shall mean the purchase price to be paid by the Buyer for the Acquired Assets.
 - "Reasonable Best Efforts" shall mean best efforts, to the extent commercially reasonable.
 - "Response" shall mean a written response containing the information provided for in Section 6.3(c).
 - "Restricted Employee" shall mean any person who was an employee of the Buyer on the date of this Agreement.
- "Retained Liabilities" shall mean any and all liabilities or obligations (whether known or unknown, absolute or contingent, liquidated or unliquidated, due or to become due and accrued or unaccrued, and whether claims with respect thereto are asserted before or after the Closing) of the Seller which are not Assumed Liabilities t. The Retained Liabilities shall include, without limitation, all liabilities and obligations of the Seller:
- (a) for income, transfer, sales, use or other Taxes arising in connection with the consummation of the transactions contemplated by this Agreement (including any income Taxes arising as a result of (i) the transfer by the Seller to the Buyer of the Acquired Assets, (ii) any deemed transfer by a subsidiary of the Seller of its assets pursuant to an election under

Section 338(h)(10) of the Code, (iii) the Seller having an "excess loss account" (within the meaning of Treasury Regulation §1.1502-19) in the stock of any Subsidiary of the Seller, or (iv) the Seller having deferred gain on any "deferred intercompany transaction" (within the meaning of Treasury Regulation §1.1502-13));

- (b) for costs and expenses incurred in connection with this Agreement or the consummation of the transactions contemplated by this Agreement;
 - (c) under this Agreement or the Ancillary Agreements;
- (d) for any Taxes, including deferred taxes or taxes measured by income of the Seller earned prior to the Closing, any liabilities for federal or state income tax and FICA taxes of employees of the Seller which the Seller is legally obligated to withhold, any liabilities of the Seller for employer FICA and unemployment taxes incurred, and any liabilities of the Seller for sales, use or excise taxes or customs and duties however, to the extent Seller assumes commissions (with respect to earnings taxed as W-2 earnings) there are payroll taxes associated with it and Buyer shall be responsible for such taxes;
- (e) under any agreements, contracts, leases or licenses which are not Assigned Contracts ;arising prior to the Closing under the Assigned Contracts, and all liabilities for any breach, act or omission by the Seller prior to the Closing under any Assigned Contract;
- (f) for any commissions earned by any employee or independent contractor of the Seller on or before the Closing Date, whether or not such employee or independent contractor is employed or engaged by Buyer on or after the Closing Date, except to the extent such commissions are Assumed Liabilities;

(g) Left Intentionally Blank.

- (h) arising out of events, conduct or conditions existing or occurring prior to the Closing that constitute a violation of or non-compliance with any law, rule or regulation (including Environmental Laws), any judgment, decree or order of any Governmental Entity, or any Permit or that give rise to liabilities or obligations with respect to Materials of Environmental Concern;
- (i) to pay severance benefits to any employee of the Seller whose employment is terminated (or treated as terminated) in connection with the consummation of the transactions contemplated by this Agreement, and all liabilities resulting from the termination of employment of employees of the Seller prior to the Closing that arose under any federal or state law or under any Employee Benefit Plan established or maintained by the Seller;
- (j) any unemployment claims filed as a result of Seller's termination of employees following Buyer's purchase of the Acquired Assets that result in any adverse consequences for Buyer including, but not limited to, any adverse effect on Buyer's unemployment insurance experience rating.
- (k) to indemnify any person or entity by reason of the fact that such person or entity was a director, officer, employee, or agent of the Seller or was serving at the request of the

Seller as a partner, trustee, director, officer, employee, or agent of another entity (whether such indemnification is for judgments, damages, penalties, fines, costs, amounts paid in settlement, losses, expenses, or otherwise and whether such indemnification is pursuant to any statute, charter document, bylaw, agreement, or otherwise);

- (l) injury to or death of persons or damage to or destruction of property occurring prior to the Closing (including any workers compensation claim);
- (m) for medical, dental and disability (both long-term and short-term benefits), whether insured or self-insured, owed to employees or former employees of the Seller based upon (A) exposure to conditions in existence prior to the Closing or (B) disabilities existing prior to the Closing (including any such disabilities which may have been aggravated following the Closing);
 - (n) for benefits under any Seller Plan;
- (o) for any retrospective premium increases under any Seller Plan assumed by Buyer that relates to periods before and including the Closing; and
 - (p) Commercial Promissory Note with United Bank for York HVAC unit.
- "Revenue" shall mean net cash received by Buyer from Buyer's operation of the Division from energy suppliers for the brokering of energy services between energy suppliers and end users of energy. Revenue shall be recorded on a basis consistent with revenue as reflected in the statements of revenues, expenses and members' equity included in Seller's historical financial statements for the years ended December 31, 2011 and 2010.
 - "Revenue Target" shall mean the Revenue metric as specified in the table set forth in Section 1.7.
 - "SEC" shall mean the Securities and Exchange Commission.
 - "Securities Act" shall mean the Securities Act of 1933, as amended.
- "Security Interest" shall mean any mortgage, pledge, security interest, encumbrance, charge or other lien (whether arising by contract or by operation of law), other than (i) mechanic's, materialmen's, and similar liens, (ii) liens arising under worker's compensation, unemployment insurance, social security, retirement, and similar legislation and (iii) liens on goods in transit incurred pursuant to documentary letters of credit, in each case arising in the Ordinary Course of Business of the Seller and not material to the Seller.
 - " Seller" shall have the meaning set forth in the first paragraph of this Agreement.
- "Seller Certificate" shall mean a certificate to the effect that each of the conditions specified in clauses (a) through (c) (insofar as clause (c) relates to Legal Proceedings involving the Seller) of Section 4.2 is satisfied in all respects.
 - "Seller Intellectual Property" shall mean shall the Seller Owned Intellectual Property and the Seller Licensed Intellectual Property.

- "Seller Licensed Intellectual Property" shall mean all Intellectual Property that is licensed to the Seller by any third party.
- "Seller Material Adverse Effect" shall mean any material adverse effect on any of the Acquired Assets or on the business, operations, property, financial condition, results of operations or cash flow of Seller or the business of Seller. For the avoidance of doubt, the parties agree that the terms "material", "materially" or "materiality" as used in this Agreement with an initial lower case "m" shall have their respective customary and ordinary meanings, without regard to the meaning ascribed to Seller Material Adverse Effect.
- "Seller Owned Intellectual Property" shall mean all Intellectual Property owned or purported to be owned by the Seller, in whole or in part.
 - "Seller Plan" shall mean any Employee Benefit Plan maintained, or contributed to, by the Seller or any ERISA Affiliate.
- "Seller Registrations" shall mean Intellectual Property Registrations that are registered or filed in the name of the Seller, alone or jointly with others.
- "Seller Source Code" shall mean the source code for any Software included in the Customer Offerings or Internal Systems or other confidential information constituting, embodied in or pertaining to such Software.
- "Shares" shall mean any shares of the Buyer's common stock issued to the Seller or Seller's designees pursuant to the terms of this Agreement.
- "Software" shall mean computer software code, applications, utilities, development tools, diagnostics, databases and embedded systems, whether in source code, interpreted code or object code form.
- "Subsidiary" shall mean any corporation, partnership, trust, limited liability company or other non-corporate business enterprise in which the Seller (or another Subsidiary) holds stock or other ownership interests representing (a) more than 50% of the voting power of all outstanding stock or ownership interests of such entity or (b) the right to receive more than 50% of the net assets of such entity available for distribution to the holders of outstanding stock or ownership interests upon a liquidation or dissolution of such entity.
- "Supplier Agreements" shall mean those agreements by and between Seller, Buyer, and energy suppliers pursuant to which Seller, Buyer, and/or any Affiliate of Buyer, is to be paid commodity brokerage fees with respect to transactions between energy suppliers and end users of energy (customers).
- "Taxes" shall mean any and all taxes, charges, fees, duties, contributions, levies or other similar assessments or liabilities in the nature of a tax, including, without limitation, income, gross receipts, corporation, ad valorem, premium, value-added, net worth, capital stock, capital gains, documentary, recapture, alternative or add-on minimum, disability, estimated, registration, recording, excise, real property, personal property, sales, use, license, lease, service, service use, transfer, withholding, employment, unemployment, insurance, social security, national

insurance, business license, business organization, environmental, workers compensation, payroll, profits, severance, stamp, occupation, windfall profits, customs duties, franchise and other taxes of any kind whatsoever imposed by the United States of America or any state, local or foreign government, or any agency or political subdivision thereof, and any interest, fines, penalties, assessments or additions to tax imposed with respect to such items or any contest or dispute thereof.

- "Tax Returns" shall mean any and all reports, returns, declarations, or statements relating to Taxes, including any schedule or attachment thereto and any related or supporting work papers or information with respect to any of the foregoing, including any amendment thereof.
 - "TD Bank" shall mean TD Bank, N.A.
- "Third Party Action" shall mean any suit or proceeding by a person or entity other than a Party for which indemnification may be sought by a Party under Article VI.
- "<u>Trademarks</u>" shall mean all registered trademarks and service marks, logos, Internet domain names, corporate names and doing business designations and all registrations and applications for registration of the foregoing, common law trademarks and service marks and trade dress.

ARTICLE IX

MISCELLANEOUS

- 9.1 <u>Press Releases and Announcements</u>. No Party shall issue any press release or public announcement relating to the subject matter of this Agreement without the prior written approval of the other Party; <u>provided</u>, <u>however</u>, that either Party may make any public disclosure it believes in good faith is required by applicable law, regulation or stock market rule.
- 9.2 No Third Party Beneficiaries. This Agreement shall not confer any rights or remedies upon any person other than the Parties and their respective successors and permitted assigns.
- 9.3 Entire Agreement. This Agreement (including the documents referred to herein) constitutes the entire agreement between the Parties and supersedes any prior understandings, agreements, or representations by or between the Parties, written or oral, with respect to the subject matter hereof, including, without limitation, that certain letter of intent dated August 8, 2012; provided that the Confidentiality Agreement dated April 19, 2012 between the Buyer and the Seller shall remain in effect in accordance with its terms.
- 9.4 <u>Succession and Assignment</u>. This Agreement shall be binding upon and inure to the benefit of the Parties named herein and their respective successors and permitted assigns. The Buyer may not assign either this Agreement or any of its rights, interests, or obligations hereunder without the prior written approval of the Seller, provided that the Buyer may assign some or all of its rights, interests and/or obligations hereunder to one or more Affiliates of the

Buyer. Seller may not assign either this Agreement or any of its rights, interests, or obligations hereunder without the prior written approval of the Buyer. Any attempted assignment in contravention of this provision shall be void.

- 9.5 <u>Counterparts and Facsimile Signature</u>. This Agreement may be executed in two or more counterparts, each of which shall be deemed an original but all of which together shall constitute one and the same instrument. This Agreement may be executed by facsimile signature or electronic signature.
- 9.6 <u>Headings</u>. The section headings contained in this Agreement are inserted for convenience only and shall not affect in any way the meaning or interpretation of this Agreement.
- 9.7 <u>Notices</u>. All notices, requests, demands, claims, and other communications hereunder shall be in writing. Any notice, request, demand, claim, or other communication hereunder shall be deemed duly delivered four business days after it is sent by registered or certified mail, return receipt requested, postage prepaid, or one business day after it is sent for next business day delivery via a reputable nationwide overnight courier service, in each case to the intended recipient as set forth below:

If to Seller:

NORTHEAST ENERGY PARTNERS, LLC 174 South Road

Enfield, Connecticut 06082

LORA A. MONROE
23 Lauren Lane

Southwick, Massachusetts 01077

THOMAS LOCKWOOD

131 Burr Hill Killingworth, Connecticut 06419

JOHN F. HARDY

9 Scarborough Drive Avon, Connecticut 06001

If to the Buyer:

World Energy Solutions, Inc. 100 Front Street Worcester, MA 01608 Attn: General Counsel Copy to

THE LAW OFFICES OF ROBERT L. IAMONACO & ASSOCIATES, P.C. 150 Trumbull Street Hartford, CT 06103

Copy to:

Mirick O'Connell 100 Front Street Worcester, MA 01608 Attn: Jeff Swaim, Esq. Any Party may give any notice, request, demand, claim, or other communication hereunder using any other means (including personal delivery, expedited courier, messenger service, telecopy, telex, ordinary mail, or electronic mail), but no such notice, request, demand, claim, or other communication shall be deemed to have been duly given unless and until it actually is received by the party for whom it is intended. Each Party may change the address to which notices, requests, demands, claims, and other communications hereunder are to be delivered by giving the other Parties notice in the manner herein set forth.

- 9.8 Governing Law. This Agreement (including the validity and applicability of the arbitration provisions of this Agreement, the conduct of any arbitration of a Dispute, the enforcement of any arbitral award made hereunder and any other questions of arbitration law or procedure arising hereunder) shall be governed by and construed in accordance with the internal laws of the Commonwealth of Massachusetts, without giving effect to any choice or conflict of law provision or rule (whether of the Commonwealth of Massachusetts or any other jurisdiction) that would cause the application of laws of any jurisdictions other than those of the Commonwealth of Massachusetts.
- 9.9 Amendments and Waivers. The Buyer and the Seller may mutually amend any provision of this Agreement at any time prior to the Closing. No amendment of any provision of this Agreement shall be valid unless the same shall be in writing and signed by each of the Buyer and the Seller. No waiver by any Party of any right or remedy hereunder shall be valid unless the same shall be in writing and signed by the Party giving such waiver. No waiver by any Party with respect to any default, misrepresentation, or breach of warranty or covenant hereunder shall be deemed to extend to any prior or subsequent default, misrepresentation, or breach of warranty or covenant hereunder or affect in any way any rights arising by virtue of any prior or subsequent such occurrence.
- 9.10 Severability. Any term or provision of this Agreement that is invalid or unenforceable in any situation in any jurisdiction shall not affect the validity or enforceability of the remaining terms and provisions hereof or the validity or enforceability of the offending term or provision in any other situation or in any other jurisdiction. If the final judgment of a court of competent jurisdiction declares that any term or provision hereof is invalid or unenforceable, the Parties agree that the court making the determination of invalidity or unenforceability shall have the power to limit the term or provision, to delete specific words or phrases, or to replace any invalid or unenforceable term or provision with a term or provision that is valid and enforceable and that comes closest to expressing the intention of the invalid or unenforceable term or provision, and this Agreement shall be enforceable as so modified.
- 9.11 Expenses. Except as set forth in Article VI, each Party shall bear its own costs and expenses (including legal fees and expenses) incurred in connection with this Agreement and the transactions contemplated hereby.
- 9.12 <u>Submission to Jurisdiction</u>. Each Party (a) submits to the jurisdiction of any state or federal court sitting in Worcester, Massachusetts in any action or proceeding arising out of or relating to this Agreement or the Ancillary Agreements, (b) agrees that all claims in respect of such action or proceeding may be heard and determined in any such court, (c) waives any claim of inconvenient forum or other challenge to venue in such court, (d) agrees not to bring any

action or proceeding arising out of or relating to this Agreement or the Ancillary Agreements in any other court and (e) waives any right it may have to a trial by jury with respect to any action or proceeding arising out of or relating to this Agreement or the Ancillary Agreements. Each party agrees to accept service of any summons, complaint or other initial pleading made in the manner provided for the giving of notices in Section 9.7, provided that nothing in this Section 9.12 shall affect the right of either Party to serve such summons, complaint or other initial pleading in any other manner permitted by law.

9.13 Specific Performance. Each Party acknowledges and agrees that the other Party would be damaged irreparably in the event any of the provisions of this Agreement (including Sections 5.1, 5.2 and 5.3) are not performed in accordance with their specific terms or otherwise are breached. Accordingly, each Party agrees that each other Party shall be entitled to an injunction or other equitable relief to prevent breaches of the provisions of this Agreement and to enforce specifically this Agreement and the terms and provisions hereof in any action instituted in any court of the United States or any state thereof having jurisdiction over the Parties and the matter, in addition to any other remedy to which it may be entitled, at law or in equity.

9.14 Construction.

- (a) The language used in this Agreement shall be deemed to be the language chosen by the Parties to express their mutual intent, and no rule of strict construction shall be applied against any Party.
- (b) Any reference to any federal, state, local, or foreign statute or law shall be deemed also to refer to all rules and regulations promulgated thereunder, unless the context requires otherwise.
 - (c) Any reference herein to "including" shall be interpreted as "including without limitation".
- (d) Any reference to any Article, Section or paragraph shall be deemed to refer to an Article, Section or paragraph of this Agreement, unless the context clearly indicates otherwise.

[signature page follows]

N WITNESS WHEREOF, the Parties have executed this Agreement as of the date first above written.	
	Buyer:
	World Energy Solutions, Inc.
	By: /s/ Phil Adams
	Phil Adams, Chief Executive Officer
	Seller:
	Northeast Energy Partners, LLC
	By: /s/ John Hardy
	Name: John Hardy
	Title: Its Manager
	For purposes of Section 5.2, 5.3 and 6.1 Only:
	/s/ John Hardy
	John Hardy
	/s/ Thomas Lockwood
	Thomas Lockwood
	/s/ Lora Monroe
	Lora Monroe

Exhibits

Exhibit A -Exhibit B -Exhibit C -

Form of Promissory Note Bill of Sale Instrument of Assumption

Schedules

Schedule 1.1(b) -Schedule 1.6 -Disclosure Schedule

Excluded Asset

Allocation of Purchase Price

PROMISSORY NOTE

Maturity Date: April 1, 2014 Principal Amount: \$2,000,000

FOR VALUE RECEIVED, World Energy Solutions, Inc. a Delaware corporation having a principal place of business at 100 Front St., Worcester, MA 01608 (the "Maker"), promises to pay to the order of Northeast Energy Partners, LLC (the "Payee"), a Connecticut limited liability company having a principal place of business 174 South Road, Enfield, Connecticut 06082, or such other place as Payee may designate, the sum of TWO MILLION DOLLARS (\$2,000,000) with interest on the unpaid principal balance computed on the basis of a 360-day year at the Note Rate.

The Note Rate is four percent (4.0%) interest per year.

Except as otherwise provided below, on, October 1, 2013, and April 1, 2014 (each, a "Payment Date"), the Maker shall pay Payee, by check payable to Payee and delivered to the Payee's address stated above, \$1,500,000 on October 1, 2013 and \$500,000 on April 1, 2014 of the unpaid principal amount of this Note, together with the accrued but unpaid interest on the unpaid principal balance of this Note with each such payment (each such payment of principal and interest, a "Cash Payment").

Notwithstanding anything to the contrary in this Note, if the Maker has delivered to Payee one or more notices of a Claim (as defined in the Asset Purchase Agreement dated October 3, 2012 by and among the Payee and John Hardy, Thomas Lockwood and Lora Monroe, being all of the members of the Seller (the "Members") (the "Asset Purchase Agreement")) on or before any Payment Date, then Maker, if not already paid, may withhold so much of the unpaid payment of principal and interest otherwise due Payee on such date as may be reasonably required to satisfy the claimed amount in such notice(s) of Claim until such time as the indemnification claim(s) that is the subject of such notice(s) of Claim is finally resolved. Maker may set-off from the Cash Payments, the aggregate amount of any and all Damages (as defined in the Asset Purchase Agreement) that Payee must indemnify Maker for with respect to any and all Claims that are the subject of any notice of Claim delivered to the Payee by the Maker on or before April 1, 2014. Notwithstanding anything to the contrary in this Note, Maker's withholding and, if applicable, set-off, of any Cash Payment pursuant to the terms of this paragraph will not constitute an Event of Default or other breach of this Promissory Note. Additionally, if any portion of a Cash Payment is withheld by Maker pursuant to the terms of this paragraph, no interest shall accrue on such portion of the Cash Payment from the applicable Payment Date until the date all Claims that are subject to Maker's notice of Claim delivered to Payee on or before such Payment Date are finally resolved. Maker shall pay to Payee any portion of the Cash Payments remaining after the satisfaction of Payee's indemnification obligations under the Asset Purchase Agreement.

While no Event of Default exists, each payment under this Note will be applied first to interest then due and then to principal. When an Event of Default exists any payments will be applied to interest and/or principal as determined by the Payee in its sole discretion.

The Maker may, at any time and without penalty, prepay any part or all of the unpaid principal balance of this Note.

The occurrence of any one or more of the following events is an Event of Default under this Note:

(i) Failure of the Maker to pay, perform or observe any of its obligations contained in this Note within seven (7) days of receipt of written notice thereof from Payee; or

- (ii) upon default by the Maker in the performance of any its obligations, covenants or agreements contained in the Asset Purchase Agreement for a period of seven (7) days after receipt of written notice thereof from Payee; or
- (iii) The termination of existence of the Maker or the involvement of the Maker in any financial difficulties as evidenced by:
- (a) an assignment for the benefit of its creditors; or
- (b) the appointment of a receiver, trustee, custodian, liquidator or conservator of it or its assets not vacated or set aside within sixty (60) days; or
- (c) the commencement by it of proceedings under any federal or state law relating to bankruptcy, insolvency or relief of debtors; or
- (d) the commencement against it of proceedings under any federal or state law relating to bankruptcy, insolvency or relief of debtors if the proceedings are not dismissed within sixty (60) days after the date on which commenced.

If an Event of Default occurs, the Payee may, to the extent permitted by law and without notice to the Maker, declare the unpaid principal balance and accrued interest to be due immediately without notice, presentment, demand, protest or other notice of dishonor of any kind, all of which are expressly waived. No course of dealing by the Payee and no delay in exercising any right under this Note will operate as a waiver by the Payee of its rights, and a waiver of a right on one occasion may not be construed as a waiver of the right on a future occasion.

This Note shall be governed by the laws of the Commonwealth of Massachusetts, and shall take effect as an instrument under seal.

This Note will be interpreted and construed under the laws of the Commonwealth of Massachusetts and will be considered to have been made, executed and performed in Massachusetts. All claims, disputes and other matters in question arising out of this agreement will be decided by proceedings instituted and litigated in a court of competent jurisdiction sitting in Massachusetts.

EXECUTED as a sealed instrument as of the 3rd day of October, 2012.

	World Energy Solutions, Inc.
Witness	Ву:
	Name: <u>James Parslow</u> Title: <u>Chief Financial Officer</u>

GENERAL BILL OF SALE

REFERENCE is made to the Asset Purchase Agreement dated of even date herewith (the "Agreement") by and among World Energy Solutions, Inc., a Delaware Corporation (the "Buyer"), Northeast Energy Partners, LLC, a Connecticut limited liability company (the "Seller"), John Hardy, Thomas Lockwood, and Lora Monroe, being all the members of the Seller. Capitalized terms not otherwise defined herein shall have the meaning given to them in the Agreement.

FOR VALUE RECEIVED pursuant to the Agreement, Seller, for itself and its successors and assigns, does hereby sell, convey, assign, transfer and deliver to and vest in Buyer and its successors and assigns all right, title and interest which Seller may have in and to all of the Acquired Assets (collectively, the "Assets").

Seller warrants that Seller hereby transfers to Buyer good, valid and transferable title to all of its Assets, free and clear of all liens, encumbrances, restrictions, agreements and adverse claims of every kind, nature and description, and agrees to defend such title.

Seller further covenants and agrees that, from time to time after the delivery of this instrument, at Buyer's request and without further consideration, Seller will do, execute, acknowledge and deliver, or cause to be done, executed, acknowledged and delivered, all such further acts, conveyances, transfers, assignments, deeds, documents and assurances as reasonably may be requested by Buyer more effectively to convey to, transfer to and vest in Buyer all right, title and interest in and to any of the Assets transferred or assigned hereunder.

Nothing contained in this General Bill of Sale shall supersede, modify, limit, eliminate or otherwise affect any of the representations and warranties, covenants, agreements or indemnities set forth in the Agreement. This General Bill of Sale is executed and delivered pursuant to the terms of the Agreement, and nothing herein shall be construed to modify, terminate or merge any rights any party thereto may have pursuant to the terms thereof. In the event of any inconsistency or conflict between the terms of the Agreement and the terms of this General Bill of Sale, the terms of the Agreement shall prevail.

This General Bill of Sale shall be governed by and construed in accordance with the laws of The Commonwealth of Massachusetts without regard to its conflict of laws provisions.

IN WITNESS WHEREOF, Seller has executed this General Bill of Sale as an instrument under seal as of this 3rd day of October, 2012.

WITNESSED BY:	Northeast Energy Partners, LLC
Signature	By: John F. Hardy Its Member Duly Authorized
Printed Name of Witness	

ASSIGNMENT AND ASSUMPTION OF CONTRACTS

This Assignment and Assumption of Contracts (this "Assignment") dated as of October 3, 2012 (the "Effective Date") is by and between Northeast Energy Partners, LLC ("Assignor") and World Energy Solutions, Inc. ("Assignee").

RECITALS

Assignor, as seller, and Assignee, as buyer, are parties to an Asset Purchase Agreement dated as of October 3, 2012 (the "APA"). Capitalized terms used but not defined in this Assignment have the meanings ascribed in the APA.

Assignor desires to assign all of its right, title and interest in and to all of Assigned Contracts (collectively, the "Contracts") to Assignee and Assignee desires to acquire all of Assignor's right, title and interest in and to the Contracts.

TERMS OF AGREEMENT

For and in consideration of the recitals set forth above, and the covenants and agreements set forth below and other valuable consideration, the receipt of which is hereby acknowledged, Assignor and Assignee agree as follows:

- 1. <u>Assignment</u>. To the extent assignable, Assignor hereby assigns, sets over and transfers to Assignee all of Assignor's right, title and interest in the Contracts. Assignor hereby represents that the Contracts previously delivered to Assignee are complete, accurate and true copies of the Contracts, including all amendments.
- 2. <u>Assumption</u>. Assignee hereby assumes and agrees to perform, fulfill and observe all of the covenants, agreements, obligations and liabilities of Assignor under the Contracts arising on and after the Effective Date. Assignor agrees to pay, perform, fulfill and observe all of the covenants, agreements, obligations and liabilities prior to the Effective Date.
- 3. <u>Assignee Indemnification</u>. Assignee agrees to indemnify and hold Assignor harmless from and against all loss, cost, damage and expense, including, without limitation, reasonable attorneys' fees, arising out of any act, omission or default by Assignee under the Contracts arising after the Effective Date.
- 4. <u>Assignor Indemnification</u>. Except as set forth in the APA, Assignor agrees to indemnify and hold Assignee harmless from and against all loss, cost, damage and expense, including, without limitation, reasonable attorneys' fees, arising out of any act, omission or default by Assignor under the Contracts arising before the Effective Date.

- 5. <u>Notices</u>. All notices, demands, requests and other communications necessary or desirable under this Assignment shall be in writing and shall be deemed properly served if sent, by United States mail, postage prepaid, registered or certified mail, return receipt requested, or national overnight express courier service, addressed as follows:
 - (a) If intended for Assignor:

World Energy Solutions, Inc. 100 Front Street Worcester, MA 01608 Attn: General Counsel

With a copy to:

Mirick O'Connell 100 Front Street Worcester, MA 01608 Attn: Jeff Swaim, Esq.

(b) If intended for Assignee:

Northeast Energy Partners, LLC 174 South Road Enfield, Connecticut 06082

With a copy to:

The Law Offices of Robert L. Iamonaco & Associates, P.C. 150 Trumbull Street Hartford, CT 06103

or at such other address or to such other individual as the party entitled to receive notices shall designate to the other in writing. Any notice shall be effective upon actual receipt or the date of the refusal by the addressee to accept delivery of such notice.

- 6. <u>Binding Effect</u>. The provisions of this Assignment are binding on and inure to the benefit of Assignor, its successors and assigns, and Assignee, its successors and assigns.
- 7. <u>Headings</u>. The section headings used in this Assignment are for reference and convenience only and shall not be used in the interpretation of this Assignment. In the event of any inconsistency or conflict between the terms of the APA and the terms of this Assignment, the terms of the APA shall prevail.
- 8. <u>Counterparts</u>. This Assignment may be signed in several counterparts, each of which is an original, but all of which constitute a single instrument.

EXECUTED under seal as of the date first written above.

Assi	nor:		
Nort	east Energy P	artners, LLC	
Ву:	John Hardy, N	Member	
Assi	nee:		
Wor	l Energy Solu	tions, Inc.	
Ву:			
Ву:	Phil Adams. (CEO	

PROMISSORY NOTE

Maturity Date: April 1, 2014 Principal Amount: \$2,000,000

FOR VALUE RECEIVED, World Energy Solutions, Inc. a Delaware corporation having a principal place of business at 100 Front St., Worcester, MA 01608 (the "Maker"), promises to pay to the order of Northeast Energy Partners, LLC (the "Payee"), a Connecticut limited liability company having a principal place of business 174 South Road, Enfield, Connecticut 06082, or such other place as Payee may designate, the sum of TWO MILLION DOLLARS (\$2,000,000) with interest on the unpaid principal balance computed on the basis of a 360-day year at the Note Rate.

The Note Rate is four percent (4.0%) interest per year.

Except as otherwise provided below, on, October 1, 2013, and April 1, 2014 (each, a "Payment Date"), the Maker shall pay Payee, by check payable to Payee and delivered to the Payee's address stated above, \$1,500,000 on October 1, 2013 and \$500,000 on April 1, 2014 of the unpaid principal amount of this Note, together with the accrued but unpaid interest on the unpaid principal balance of this Note with each such payment (each such payment of principal and interest, a "Cash Payment").

Notwithstanding anything to the contrary in this Note, if the Maker has delivered to Payee one or more notices of a Claim (as defined in the Asset Purchase Agreement dated October 3, 2012 by and among the Payee and John Hardy, Thomas Lockwood and Lora Monroe, being all of the members of the Seller (the "Members") (the "Asset Purchase Agreement")) on or before any Payment Date, then Maker, if not already paid, may withhold so much of the unpaid payment of principal and interest otherwise due Payee on such date as may be reasonably required to satisfy the claimed amount in such notice(s) of Claim until such time as the indemnification claim(s) that is the subject of such notice(s) of Claim is finally resolved. Maker may set-off from the Cash Payments, the aggregate amount of any and all Damages (as defined in the Asset Purchase Agreement) that Payee must indemnify Maker for with respect to any and all Claims that are the subject of any notice of Claim delivered to the Payee by the Maker on or before April 1, 2014. Notwithstanding anything to the contrary in this Note, Maker's withholding and, if applicable, set-off, of any Cash Payment pursuant to the terms of this paragraph will not constitute an Event of Default or other breach of this Promissory Note. Additionally, if any portion of a Cash Payment is withheld by Maker pursuant to the terms of this paragraph, no interest shall accrue on such portion of the Cash Payment from the applicable Payment Date until the date all Claims that are subject to Maker's notice of Claim delivered to Payee on or before such Payment Date are finally resolved. Maker shall pay to Payee any portion of the Cash Payments remaining after the satisfaction of Payee's indemnification obligations under the Asset Purchase Agreement.

While no Event of Default exists, each payment under this Note will be applied first to interest then due and then to principal. When an Event of Default exists any payments will be applied to interest and/or principal as determined by the Payee in its sole discretion.

The Maker may, at any time and without penalty, prepay any part or all of the unpaid principal balance of this Note.

The occurrence of any one or more of the following events is an Event of Default under this Note:

- (i) Failure of the Maker to pay, perform or observe any of its obligations contained in this Note within seven (7) days of receipt of written notice thereof from Payee; or
- (ii) upon default by the Maker in the performance of any its obligations, covenants or agreements contained in the Asset Purchase Agreement for a period of seven (7) days after receipt of written notice thereof from Payee; or
- (iii) The termination of existence of the Maker or the involvement of the Maker in any financial difficulties as evidenced by:
- (a) an assignment for the benefit of its creditors; or
- (b) the appointment of a receiver, trustee, custodian, liquidator or conservator of it or its assets not vacated or set aside within sixty (60) days; or
- (c) the commencement by it of proceedings under any federal or state law relating to bankruptcy, insolvency or relief of debtors; or
- (d) the commencement against it of proceedings under any federal or state law relating to bankruptcy, insolvency or relief of debtors if the proceedings are not dismissed within sixty (60) days after the date on which commenced.

If an Event of Default occurs, the Payee may, to the extent permitted by law and without notice to the Maker, declare the unpaid principal balance and accrued interest to be due immediately without notice, presentment, demand, protest or other notice of dishonor of any kind, all of which are expressly waived. No course of dealing by the Payee and no delay in exercising any right under this Note will operate as a waiver by the Payee of its rights, and a waiver of a right on one occasion may not be construed as a waiver of the right on a future occasion.

This Note shall be governed by the laws of the Commonwealth of Massachusetts, and shall take effect as an instrument under seal.

This Note will be interpreted and construed under the laws of the Commonwealth of Massachusetts and will be considered to have been made, executed and performed in Massachusetts. All claims, disputes and other matters in question arising out of this agreement will be decided by proceedings instituted and litigated in a court of competent jurisdiction sitting in Massachusetts.

EXECUTED as a sealed instrument as of the 3rd day of October, 2012.

world Energy Solutions, Inc.		
By: /s/ James Parslow	 	

/s/ Carolyn Oldenburg
Witness

Name: <u>James Parslow</u> Title: <u>Chief Financial Officer</u> THIS WARRANT AND THE SHARES ISSUABLE HEREUNDER HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "ACT"), OR THE SECURITIES LAWS OF ANY STATE AND, EXCEPT AS SET FORTH IN SECTIONS 5.3 AND 5.4 BELOW, MAY NOT BE OFFERED, SOLD, PLEDGED OR OTHERWISE TRANSFERRED UNLESS AND UNTIL REGISTERED UNDER SAID ACT AND LAWS OR, IN THE OPINION OF LEGAL COUNSEL IN FORM AND SUBSTANCE SATISFACTORY TO THE ISSUER, SUCH OFFER, SALE, PLEDGE OR OTHER TRANSFER IS EXEMPT FROM SUCH REGISTRATION.

WARRANT TO PURCHASE STOCK

Company: World Energy Solutions, Inc., a Delaware corporation

Number of Shares: 45,045, subject to adjustment

Type/Series of Stock: Common Stock, \$0.0001 par value per share

Warrant Price: \$4.44 per Share, subject to adjustment

Issue Date: October 3, 2012

Expiration Date: October 2, 2019 See also Section 5.1(b).

Credit Facility: This Warrant to Purchase Stock ("Warrant") is issued in connection with that certain Fourth Loan Modification Agreement,

of even date herewith, to that certain Loan and Security Agreement dated September 8, 2008, between Silicon Valley Bank and the Company, as amended (collectively, and as further amended and/or modified and in effect from time to time, the "

Loan Agreement").

THIS WARRANT CERTIFIES THAT, for good and valuable consideration, SILICON VALLEY BANK (together with any successor or permitted assignee or transferee of this Warrant or of any shares issued upon exercise hereof, "Holder") is entitled to purchase the number of fully paid and non-assessable shares (the "Shares") of the above-stated Type/Series of Stock (the "Class") of the above-named company (the "Company") at the above-stated Warrant Price, all as set forth above and as adjusted pursuant to Section 2 of this Warrant, subject to the provisions and upon the terms and conditions set forth in this Warrant. Reference is made to Section 5.4 of this Warrant whereby Silicon Valley Bank shall transfer this Warrant to its parent company, SVB Financial Group.

SECTION 1. EXERCISE.

1.1 Method of Exercise. Holder may at any time and from time to time exercise this Warrant, in whole or in part, by delivering to the Company the original of this Warrant together with a duly executed Notice of Exercise in substantially the form attached hereto as Appendix 1 and, unless Holder is exercising this Warrant pursuant to a cashless exercise set forth in Section 1.2, a check, wire transfer of sameday funds (to an account designated by the Company), or other form of payment acceptable to the Company for the aggregate Warrant Price for the Shares being purchased.

1.2 <u>Cashless Exercise</u>. On any exercise of this Warrant, in lieu of payment of the aggregate Warrant Price in the manner as specified in Section 1.1 above, but otherwise in accordance with the requirements of Section 1.1, Holder may elect to receive Shares equal to the value of this Warrant, or portion hereof as to which this Warrant is being exercised. Thereupon, the Company shall issue to the Holder such number of fully paid and non-assessable Shares as are computed using the following formula:

X = Y(A-B)/A

where:

- X = the number of Shares to be issued to the Holder;
- Y = the number of Shares with respect to which this Warrant is being exercised (inclusive of the Shares surrendered to the Company in payment of the aggregate Warrant Price);
- A = the Fair Market Value (as determined pursuant to Section 1.3 below) of one Share; and
- B = the Warrant Price.
- 1.3 Fair Market Value. If shares of the Class are then traded or quoted on a nationally recognized securities exchange, inter-dealer quotation system or over-the-counter market (a "Trading Market"), the fair market value of a Share shall be the closing price or last sale price of a share of the Class reported for the Business Day immediately before the date on which Holder delivers this Warrant together with its Notice of Exercise to the Company. If shares of the Class are not then traded in a Trading Market, the Board of Directors of the Company shall determine the fair market value of a Share in its reasonable good faith judgment.
- 1.4 <u>Delivery of Certificate and New Warrant</u>. Within a reasonable time after Holder exercises this Warrant in the manner set forth in Section 1.1 or 1.2 above, the Company shall deliver to Holder a certificate representing the Shares issued to Holder upon such exercise and, if this Warrant has not been fully exercised and has not expired, a new warrant of like tenor representing the Shares not so acquired.
- 1.5 <u>Replacement of Warrant</u>. On receipt of evidence reasonably satisfactory to the Company of the loss, theft, destruction or mutilation of this Warrant and, in the case of loss, theft or destruction, on delivery of an indemnity agreement reasonably satisfactory in form, substance and amount to the Company or, in the case of mutilation, on surrender of this Warrant to the Company for cancellation, the Company shall, within a reasonable time, execute and deliver to Holder, in lieu of this Warrant, a new warrant of like tenor and amount.
 - 1.6 Treatment of Warrant Upon Acquisition of Company.
- (a) Acquisition. For the purpose of this Warrant, "Acquisition" means any transaction or series of related transactions involving:
 (i) the sale, lease, exclusive license, or other disposition of all or substantially all of the assets of the Company (ii) any merger or consolidation of the Company into or with another person or entity (other than a merger or consolidation effected exclusively to change the Company's domicile), or any other corporate reorganization, in which the stockholders of the Company in their capacity as such immediately prior to such merger, consolidation or reorganization, own less than a majority of the Company's (or the surviving or successor entity's) outstanding voting power immediately after such merger, consolidation or reorganization (or, if such

Company stockholders beneficially own a majority of the outstanding voting power of the surviving or successor entity as of immediately after such merger, consolidation or reorganization, such surviving or successor entity is not the Company); or (iii) any sale or other transfer by the stockholders of the Company of shares representing at least a majority of the Company's then-total outstanding combined voting power.

- (b) <u>Treatment of Warrant at Acquisition</u>. In the event of an Acquisition in which the consideration to be received by the Company's stockholders consists solely of cash, solely of Marketable Securities or a combination of cash and Marketable Securities (a "<u>Cash/Public Acquisition</u>"), either (i) Holder shall exercise this Warrant pursuant to Section 1.1 and/or 1.2 and such exercise will be deemed effective immediately prior to and contingent upon the consummation of such Acquisition or (ii) if Holder elects not to exercise the Warrant, this Warrant will expire immediately prior to the consummation of such Acquisition.
- (c) The Company shall provide Holder with written notice of its request relating to the Cash/Public Acquisition (together with such reasonable information as Holder may reasonably require regarding the treatment of this Warrant in connection with such contemplated Cash/Public Acquisition giving rise to such notice), which is to be delivered to Holder not less than seven (7) Business Days prior to the closing of the proposed Cash/Public Acquisition. In the event the Company does not provide such notice, then if, immediately prior to the Cash/Public Acquisition, the fair market value of one Share (or other security issuable upon the exercise hereof) as determined in accordance with Section 1.3 above would be greater than the Warrant Price in effect on such date, then this Warrant shall automatically be deemed on and as of such date to be exercised pursuant to Section 1.2 above as to all Shares (or such other securities) for which it shall not previously have been exercised, and the Company shall promptly notify the Holder of the number of Shares (or such other securities) issued upon such exercise to the Holder and Holder shall be deemed to have restated each of the representations and warranties in Section 4 of the Warrant as the date thereof.
- (d) Upon the closing of any Acquisition other than a Cash/Public Acquisition, the acquiring, surviving or successor entity shall assume the obligations of this Warrant, and this Warrant shall thereafter be exercisable for the same securities and/or other property as would have been paid for the Shares issuable upon exercise of the unexercised portion of this Warrant as if such Shares were outstanding on and as of the closing of such Acquisition, subject to further adjustment from time to time in accordance with the provisions of this Warrant.
- (e) As used in this Warrant, "Marketable Securities" means securities meeting all of the following requirements: (i) the issuer thereof is then subject to the reporting requirements of Section 13 or Section 15(d) of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), and is then current in its filing of all required reports and other information under the Act and the Exchange Act; (ii) the class and series of shares or other security of the issuer that would be received by Holder in connection with the Acquisition were Holder to exercise this Warrant on or prior to the closing thereof is then traded in Trading Market, and (iii) following the closing of such Acquisition, Holder would not be restricted from publicly re-selling all of the issuer's shares and/or other securities that would be received by Holder in such Acquisition were Holder to exercise or convert this Warrant in full on or prior to the closing of such Acquisition, except to the extent that any such restriction (x) arises solely under federal or state securities laws, rules or regulations, and (y) does not extend beyond six (6) months from the closing of such Acquisition.

SECTION 2. ADJUSTMENTS TO THE SHARES AND WARRANT PRICE.

- 2.1 Stock Dividends, Splits, Etc. If the Company declares or pays a dividend or distribution on the outstanding shares of the Class payable in additional shares of the Class or other securities or property (other than cash), then upon exercise of this Warrant, for each Share acquired, Holder shall receive, without additional cost to Holder, the total number and kind of securities and property which Holder would have received had Holder owned the Shares of record as of the date the dividend or distribution occurred. If the Company subdivides the outstanding shares of the Class by reclassification or otherwise into a greater number of shares, the number of Shares purchasable hereunder shall be proportionately increased and the Warrant Price shall be proportionately decreased. If the outstanding shares of the Class are combined or consolidated, by reclassification or otherwise, into a lesser number of shares, the Warrant Price shall be proportionately increased and the number of Shares shall be proportionately decreased.
- 2.2 Reclassification, Exchange, Combinations or Substitution. Upon any event whereby all of the outstanding shares of the Class are reclassified, exchanged, combined, substituted, or replaced for, into, with or by Company securities of a different class and/or series, then from and after the consummation of such event, this Warrant will be exercisable for the number, class and series of Company securities that Holder would have received had the Shares been outstanding on and as of the consummation of such event, and subject to further adjustment thereafter from time to time in accordance with the provisions of this Warrant. The provisions of this Section 2.2 shall similarly apply to successive reclassifications, exchanges, combinations substitutions, replacements or other similar events.
- 2.3 No Fractional Share. No fractional Share shall be issuable upon exercise of this Warrant and the number of Shares to be issued shall be rounded down to the nearest whole Share. If a fractional Share interest arises upon any exercise of the Warrant, the Company shall eliminate such fractional Share interest by paying Holder in cash the amount computed by multiplying the fractional interest by (i) the fair market value (as determined in accordance with Section 1.3 above) of a full Share, less (ii) the then-effective Warrant Price.
- 2.4 Notice/Certificate as to Adjustments. Upon each adjustment of the Warrant Price, Class and/or number of Shares, the Company, at the Company's expense, shall notify Holder in writing within a reasonable time setting forth the adjustments to the Warrant Price, Class and/or number of Shares and facts upon which such adjustment is based. The Company shall, upon written request from Holder, furnish Holder with a certificate of its Chief Financial Officer, including computations of such adjustment and the Warrant Price, Class and number of Shares in effect upon the date of such adjustment.

SECTION 3. REPRESENTATIONS AND COVENANTS OF THE COMPANY.

- 3.1 Representations and Covenants. The Company represents and warrants to, and agrees with, the Holder as follows:
- (a) All Shares which may be issued upon the exercise of this Warrant shall, upon issuance, be duly authorized, validly issued, fully paid and non-assessable, and free of any liens and encumbrances except for restrictions on transfer provided for herein or under applicable federal and state securities laws.

- (b) The Company covenants that it shall at all times cause to be reserved and kept available out of its authorized and unissued capital stock such number of shares of the Class and other securities as will be sufficient to permit the exercise in full of this Warrant.
 - 3.2 Notice of Certain Events. If the Company proposes at any time to:
- (a) declare any dividend or distribution upon the outstanding shares of the Class, whether in cash, property, stock, or other securities and whether or not a regular cash dividend;
- (b) offer for subscription or sale pro rata to the holders of the outstanding shares of the Class any additional shares of any class or series of the Company's stock (other than pursuant to contractual pre-emptive rights);
- (c) effect any reclassification, exchange, combination, substitution, reorganization or recapitalization of the outstanding shares of the Class; or
 - (d) effect an Acquisition or to liquidate, dissolve or wind up;

then, in connection with each such event, the Company shall give Holder notice thereof at the same time and in the same manner as given to holders of the outstanding shares of the Class.

SECTION 4. REPRESENTATIONS, WARRANTIES OF THE HOLDER.

The Holder represents and warrants to the Company as follows:

- 4.1 <u>Purchase for Own Account.</u> This Warrant and the Shares to be acquired upon exercise of this Warrant by Holder are being acquired for investment for Holder's account, not as a nominee or agent, and not with a view to the public resale or distribution within the meaning of the Act. Holder also represents that it has not been formed for the specific purpose of acquiring this Warrant or the Shares.
- 4.2 <u>Disclosure of Information</u>. Holder is aware of the Company's business affairs and financial condition and has received or has had full access to all the information it considers necessary or appropriate to make an informed investment decision with respect to the acquisition of this Warrant and its underlying securities. Holder further has had an opportunity to ask questions and receive answers from the Company regarding the terms and conditions of the offering of this Warrant and its underlying securities and to obtain additional information (to the extent the Company possessed such information or could acquire it without unreasonable effort or expense) necessary to verify any information furnished to Holder or to which Holder has access.
- 4.3 Investment Experience. Holder understands that the purchase of this Warrant and its underlying securities involves substantial risk. Holder has experience as an investor in securities of companies in the development stage and acknowledges that Holder can bear the economic risk of such Holder's investment in this Warrant and its underlying securities and has such knowledge and experience in financial or business matters that Holder is capable of evaluating the merits and risks of its investment in this Warrant and its underlying securities and/or has a preexisting personal or business relationship with the Company and certain of its officers, directors or controlling persons of a nature and duration that enables Holder to be aware of the character, business acumen and financial circumstances of such persons.

- 4.4 Accredited Investor Status. Holder is an "accredited investor" within the meaning of Regulation D promulgated under the Act.
- 4.5 The Act. Holder understands that this Warrant and the Shares issuable upon exercise hereof have not been registered under the Act in reliance upon a specific exemption therefrom, which exemption depends upon, among other things, the bona fide nature of the Holder's investment intent as expressed herein. Holder understands that this Warrant and the Shares issued upon any exercise hereof must be held indefinitely unless subsequently registered under the Act and qualified under applicable state securities laws, or unless exemption from such registration and qualification are otherwise available. Holder is aware of the provisions of Rule 144 promulgated under the Act.
 - 4.6 No Voting Rights. Holder, as a Holder of this Warrant, will not have any voting rights until the exercise of this Warrant.

SECTION 5. MISCELLANEOUS.

- 5.1 Term; Automatic Cashless Exercise Upon Expiration.
- (a) <u>Term</u>. Subject to the provisions of Section 1.6 above, this Warrant is exercisable in whole or in part at any time and from time to time on or before 6:00 PM, Pacific time, on the Expiration Date and shall be void thereafter.
- (b) <u>Automatic Cashless Exercise upon Expiration</u>. In the event that, upon the Expiration Date, the fair market value of one Share as determined in accordance with Section 1.3 above is greater than the Warrant Price in effect on such date, then this Warrant shall automatically be deemed on and as of such date to be exercised pursuant to Section 1.2 above as to all Shares for which it shall not previously have been exercised, and the Company shall, within a reasonable time, deliver a certificate representing the Shares issued upon such exercise to Holder.
 - 5.2 Legends. Each certificate evidencing Shares shall be imprinted with a legend in substantially the following form: THE SHARES EVIDENCED BY THIS CERTIFICATE HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "ACT"), OR THE SECURITIES LAWS OF ANY STATE AND, EXCEPT AS SET FORTH IN THAT CERTAIN WARRANT TO PURCHASE STOCK ISSUED BY THE ISSUER TO SILICON VALLEY BANK DATED OCTOBER 3, 2012, MAY NOT BE OFFERED, SOLD, PLEDGED OR OTHERWISE TRANSFERRED UNLESS AND UNTIL REGISTERED UNDER SAID ACT AND LAWS OR, IN THE OPINION OF LEGAL COUNSEL IN FORM AND SUBSTANCE SATISFACTORY TO THE ISSUER, SUCH OFFER, SALE, PLEDGE OR OTHER TRANSFER IS EXEMPT FROM SUCH REGISTRATION.
- 5.3 <u>Compliance with Securities Laws on Transfer</u>. This Warrant and the Shares issued upon exercise of this Warrant may not be transferred or assigned in whole or in part except in compliance with applicable federal and state securities laws by the transferor and the transferee (including, without limitation, the delivery of investment representation letters and legal opinions

reasonably satisfactory to the Company, as reasonably requested by the Company). The Company shall not require Holder to provide an opinion of counsel if the transfer is to SVB Financial Group (Silicon Valley Bank's parent company) or any other affiliate of Holder, provided that any such transferee is an "accredited investor" as defined in Regulation D promulgated under the Act and represents to the Company in writing that it is both such an accredited investor and an affiliate of Holder. Additionally, the Company shall also not require an opinion of counsel if there is no material question as to the availability of Rule 144 promulgated under the Act.

5.4 Transfer Procedure. After receipt by Silicon Valley Bank of the executed Warrant, Silicon Valley Bank will transfer all of this Warrant to its parent company, SVB Financial Group. By its acceptance of this Warrant, SVB Financial Group hereby makes to the Company each of the representations and warranties set forth in Section 4 hereof and agrees to be bound by all of the terms and conditions of this Warrant as if the original Holder hereof. Subject to the provisions of Section 5.3 and upon providing the Company with written notice, SVB Financial Group and any subsequent Holder may transfer all or part of this Warrant or the Shares issued upon exercise of this Warrant to any transferee, provided, however, in connection with any such transfer, SVB Financial Group or any subsequent Holder will give the Company notice of the portion of the Warrant and/or Shares being transferred with the name, address and taxpayer identification number of the transferee and Holder will surrender this Warrant to the Company for reissuance to the transferee(s) (and Holder if applicable); and provided further, that any subsequent transferee other than SVB Financial Group shall agree in writing with the Company to be bound by all of the terms and conditions of this Warrant.

5.5 Notices. All notices and other communications hereunder from the Company to the Holder, or vice versa, shall be deemed delivered and effective (i) when given personally, (ii) on the third (3 rd) Business Day after being mailed by first-class registered or certified mail, postage prepaid, (iii) upon actual receipt if given by facsimile or electronic mail and such receipt is confirmed in writing by the recipient, or (iv) on the first Business Day following delivery to a reliable overnight courier service, courier fee prepaid, in any case at such address as may have been furnished to the Company or Holder, as the case may be, in writing by the Company or such Holder from time to time in accordance with the provisions of this Section 5.5. All notices to Holder shall be addressed as follows until the Company receives notice of a change of address in connection with a transfer or otherwise:

SVB Financial Group Attn: Treasury Department 3003 Tasman Drive, HC 215 Santa Clara, CA 95054 Telephone: (408) 654-7400 Facsimile: (408) 988-8317

Email address: derivatives@svb.com

Notice to the Company shall be addressed as follows until Holder receives notice of a change in address:

World Energy Solutions, Inc. Attn: Chief Financial Officer 100 Front Street Worcester, MA 01608 Telephone: (508) 459-8100 Facsimile: (508) 459-8101

Email: jparslow@worldenergy.com

- 5.6 <u>Waiver</u>. This Warrant and any term hereof may be changed, waived, discharged or terminated (either generally or in a particular instance and either retroactively or prospectively) only by an instrument in writing signed by the party against which enforcement of such change, waiver, discharge or termination is sought.
- 5.7 Attorneys' Fees. In the event of any dispute between the parties concerning the terms and provisions of this Warrant, the party prevailing in such dispute shall be entitled to collect from the other party all costs incurred in such dispute, including reasonable attorneys' fees.
- 5.8 <u>Counterparts</u>; <u>Facsimile/Electronic Signatures</u>. This Warrant may be executed in counterparts, all of which together shall constitute one and the same agreement. Any signature page delivered electronically or by facsimile shall be binding to the same extent as an original signature page with regards to any agreement subject to the terms hereof or any amendment thereto.
- 5.9 Governing Law. This Warrant shall be governed by and construed in accordance with the laws of the State of California, without giving effect to its principles regarding conflicts of law.
- 5.10 <u>Headings</u>. The headings in this Warrant are for purposes of reference only and shall not limit or otherwise affect the meaning of any provision of this Warrant.
 - 5.11 Business Days. "Business Day" is any day that is not a Saturday, Sunday or a day on which Silicon Valley Bank is closed.

[Remainder of page left blank intentionally] [Signature page follows] IN WITNESS WHEREOF, the parties have caused this Warrant to Purchase Stock to be executed by their duly authorized representatives effective as of the Issue Date written above.

"COMPANY"

WORLD ENERGY SOLUTIONS, INC.

By: /s/ James Parslow

Name: James Parslow

(Print)

Title: Chief Financial Officer

"HOLDER"

SILICON VALLEY BANK

By: /s/ Darren Gastrock

Name: Darren Gastrock

(Print)

Title: Relationship Manager

APPENDIX 1

NOTICE OF EXERCISE

	he undersigned Holder hereby exercises its right to purchase shares of the Common/Series Preferred [circle one]				
Stock of Warrant I	(the "Company") in accordance with the attached Warrant To Purchase Stock, and tenders payment of the aggregate rice for such shares as follows:				
	check in the amount of \$ payable to order of the Company enclosed herewith				
	Cashless Exercise pursuant to Section 1.2 of the Warrant				
. 🗆	Other [Describe]				
2. P	lease issue a certificate or certificates representing the Shares in the name specified below:				
	Holder's Name				
	(Address)				
3. E	y its execution below and for the benefit of the Company, Holder hereby restates each of the representations and warranties in				
	of the Warrant to Purchase Stock as of the date hereof.				
	HOLDER:				
					
	Ву:				
	Name:				
	Title:				
	(Date):				

Schedule I

WORLD ENERGY SOLUTIONS, INC.

SUBORDINATED NOTE DUE 2020

\$4,000,000 October 3, 2012

For value received, World Energy Solutions, Inc., a Delaware corporation (the "Company"), hereby promises to pay to Massachusetts Capital Resource Company or registered assigns (hereinafter referred to as the "Payee"), on or before September 30, 2020, the principal sum of Four Million Dollars (\$4,000,000) or such part thereof as then remains unpaid pursuant to the terms set forth in that certain Note Purchase Agreement, dated as of October 3, 2012, between the Company and Massachusetts Capital Resource Company (as the same may be amended from time to time, hereinafter referred to as the "Agreement"), and to pay interest from the date hereof on the whole amount of said principal sum remaining from time to time unpaid at the rate of ten and one-half percent (10.5%) per annum, such interest to be payable monthly on the last day of each calendar month in each year, the first such payment to be due and payable on October 31, 2012, until the whole amount of the principal hereof remaining unpaid shall become due and payable, and to pay interest at the rate of fourteen percent (14%) (so far as the same may be legally enforceable) on all overdue principal (including any overdue required redemption), premium and interest. All or a portion of the principal amount of this Note must be redeemed in the amounts and at the times set forth in Section 1.04 of the Agreement. Principal, premium, if any, and interest shall be payable in lawful money of the United States of America, in immediately available funds, at the principal office of the Payee or at such other place as the legal holder may designate from time to time in writing to the Company. Interest shall be computed on the basis of a 360-day year and a 30-day month.

This Note is issued pursuant to and is entitled to the benefits of the Agreement, and each holder of this Note, by his acceptance hereof, agrees to be bound by the provisions of the Agreement, including, without limitation, that (i) this Note is subject to prepayment, in whole or in part, as specified in said Agreement, (ii) the principal of and interest on this Note is subordinated to Senior Debt, as defined in the Agreement and (iii) in case of an Event of Default, as defined in the Agreement, the principal of this Note may become or may be declared due and payable in the manner and with the effect provided in the Agreement.

As further provided in the Agreement, upon surrender of this Note for transfer or exchange, a new Note or new Notes of the same tenor dated the date to which interest has been paid on the surrender Note and in an aggregate principal amount equal to the unpaid principal amount of the Note so surrendered will be issued to, and registered in the name of, the transferee or transferees. The Company may treat the person in whose name this Note is registered as the owner hereof for the purpose of receiving payment and for all other purposes.

In case any payment herein provided for shall not be paid when due, the Company promises to pay all cost of collection, including all reasonable attorney's fees.

This Note shall be governed by, and construed in accordance with, the laws of the Commonwealth of Massachusetts and shall have the effect of a sealed instrument.

The Company and all endorsers and guarantors of this Note hereby waive presentment, demand, notice of nonpayment, protest and all other demands and notices in connection with the delivery, acceptance, performance or enforcement of this Note.

WORLD ENERGY SOLUTIONS, INC.

By /s/ Philip Adams

Philip Adams, Chief Executive Officer

FOURTH LOAN MODIFICATION AGREEMENT

This Fourth Loan Modification Agreement (this "Loan Modification Agreement") is entered into as of October 3, 2012 (the "Fourth Loan Modification Effective Date"), by and between (i) SILICON VALLEY BANK, a California corporation, with its principal place of business at 3003 Tasman Drive, Santa Clara, California 95054 and with a loan production office located at 275 Grove Street, Suite 2-200, Newton, Massachusetts 02466 ("Bank") and (ii) WORLD ENERGY SOLUTIONS, INC., a Delaware corporation with offices located at 100 Front Street, Worcester, Massachusetts 01608, and WORLD ENERGY SECURITIES CORP., a Massachusetts securities corporation with offices located at 100 Front Street, Worcester, Massachusetts 01608 (individually and collectively, jointly and severally, "Borrower").

- 1. <u>DESCRIPTION OF EXISTING INDEBTEDNESS AND OBLIGATIONS</u>. Among other indebtedness and obligations which may be owing by Borrower to Bank, Borrower is indebted to Bank pursuant to a loan arrangement dated as of September 8, 2008, evidenced by, among other documents, a certain Loan and Security Agreement dated as of September 8, 2008, between Borrower and Bank, as amended by a certain First Loan Modification Agreement, dated as of September 30, 2009, as further amended by a certain Second Loan Modification Agreement, dated as of March 8, 2011 and as further amended by a certain Third Loan Modification and Waiver Agreement, dated as of March 2, 2012 (as amended, the "Loan Agreement"). Capitalized terms used but not otherwise defined herein shall have the same meaning as in the Loan Agreement.
- 2. <u>DESCRIPTION OF COLLATERAL</u>. Repayment of the Obligations is secured by the Collateral as described in the Loan Agreement (together with any other collateral security granted to Bank, the "Security Documents").

Hereinafter, the Security Documents, together with all other documents evidencing or securing the Obligations shall be referred to as the "Existing Loan Documents".

3. DESCRIPTION OF CHANGE IN TERMS.

- A. Modifications to Loan Agreement.
 - The Loan Agreement shall be amended by inserting the following new Section 2.1.6 immediately following Section 2.1.5 thereof:

" 2.1.6 Term Loan 2012A.

- (a) Existing Term Loan 2012 Indebtedness. Bank and Borrower acknowledge and agree that Borrower is indebted to Bank pursuant to the Term Loan 2012. As of the Fourth Loan Modification Effective Date, prior to application of the proceeds of the Term Loan 2012A, the outstanding principal balance of the Term Loan 2012 is Two Million Five Hundred Thousand Dollars (\$2,500,000).
- (b) Availability. Bank shall make one (1) term loan available to Borrower in an amount up to the Term Loan 2012A Amount on the Fourth Loan Modification Effective Date, subject to the satisfaction of the terms and conditions of this Agreement.
- (c) Repayment. Commencing on the first day of the month following the month in which the Funding Date occurs and thereafter on the first day of each successive calendar month until the Term Loan 2012A is paid in full, Borrower shall make monthly payments of interest in arrears with respect to the Term Loan 2012A. Commencing on January 1, 2013, Borrower shall repay the principal amount of the Term Loan 2012A in thirty-nine (39) equal installments of principal, based on a thirty-nine (39)

month amortization schedule (each payment of principal and/or interest being a "Term Loan 2012A Payment"). Each Term Loan 2012A Payment shall be payable on the first day of each month. Borrower's final Term Loan 2012A Payment, due on the Term Loan 2012A Maturity Date, shall include all outstanding principal and accrued and unpaid interest under the Term Loan 2012A. Once repaid, the Term Loan 2012A may not be reborrowed.

- (d) <u>Prepayment</u>. The Term Loan 2012A may be prepaid, in whole prior to the Term Loan 2012A Maturity Date by Borrower, effective three (3) Business Days after written notice of such prepayment is given to Bank. Notwithstanding any such prepayment, Bank's lien and security interest in the Collateral shall continue until Borrower fully satisfies its Obligations (other than inchoate indemnity obligations). If such prepayment is at Borrower's election or at Bank's election due to the occurrence and continuance of an Event of Default, Borrower shall pay to Bank, in addition to the payment of any other expenses or fees then-owing, if such prepayment occurs on or before the first anniversary of the Fourth Loan Modification Effective Date, a prepayment premium in an amount equal to one percent (1.00%) of the principal amount of the Term Loan 2012A then outstanding; <u>provided that</u> no prepayment premium shall be charged if the Term Loan 2012A is replaced with a new facility from another division of Silicon Valley Bank. Upon payment in full of the Obligations (other than inchoate indemnity obligations) and at such time as Bank's obligation to make Credit Extensions has terminated, Bank shall terminate and release its liens and security interests in the Collateral and all rights therein shall revert to Borrower.
- (e) <u>Use of Proceeds</u>. Proceeds of the Term Loan 2012A will be used (i) for the repayment in full of the outstanding principal balance of, plus the accrued but unpaid interest on, the Term Loan 2012 (which repayment shall not be subject to any prepayment premium); and (ii) to fund a portion of the purchase price of the NEP Acquisition."
- 2 The Loan Agreement shall be amended by deleting the following text appearing as Section 2.3(a)(ii) thereof:
 - "(ii) <u>Term Loan 2012</u>. Subject to Section 2.3(b) the principal amount outstanding under the Term Loan 2012 shall accrue interest at a floating per annum rate equal to the Prime Rate plus two and one-quarter of one percentage point (2.25%), which interest shall be payable monthly on the first day of each month."

and inserting in lieu thereof the following:

- "(ii) <u>Term Loan 2012A</u>. Subject to Section 2.3(b) the principal amount outstanding under the Term Loan 2012A shall accrue interest at a floating per annum rate equal to the Prime Rate plus two and three-quarters of one percentage point (2.75%), which interest shall be payable monthly on the first day of each month."
- 3 The Loan Agreement shall be amended by deleting the following text appearing as Section 6.9 thereof:

" 6.9 Financial Covenants .

Borrower shall maintain at all times, to be tested as of the last day of each month, unless otherwise noted, on a consolidated basis with respect to Borrower and its Subsidiaries:

(a) Minimum Cash and Availability. Commencing February 28, 2012 and at all times thereafter, maintain unrestricted cash of Borrower at Bank plus unused Availability Amount of not less than One Million Two Hundred Fifty Thousand Dollars (\$1,250,000);

(b) <u>Minimum Fixed Charge Coverage Ratio</u>. Commencing with the first full fiscal quarter following the Third Loan Modification Effective Date, achieve, on a trailing three month basis, measured on the last day of each monthly period, a Fixed Charge Coverage Ratio of not less than 1.25:1.00.

and inserting in lieu thereof the following:

" 6.9 Financial Covenants.

Borrower shall maintain at all times, to be tested as of the last day of each month, unless otherwise noted, on a consolidated basis with respect to Borrower and its Subsidiaries:

- (a) Minimum Cash and Availability. Commencing on the Fourth Loan Modification Effective Date and at all times thereafter, maintain unrestricted cash of Borrower at Bank plus unused Availability Amount of not less than Three Million Dollars (\$3,000,000);
- (b) <u>Minimum Fixed Charge Coverage Ratio</u>. Commencing with the first full fiscal quarter following the Fourth Loan Modification Effective Date, achieve, on a trailing three month basis, measured on the last day of each monthly period, a Fixed Charge Coverage Ratio of not less than 1.25:1.00.
- 4 The Loan Agreement shall be amended by deleting the following text appearing as Section 7.7 thereof:
 - "7.7 Investments; Distributions. (a) Directly or indirectly make any Investment other than Permitted Investments, or permit any of its Subsidiaries to do so; (b) pay any dividends or make any distribution or payment or redeem, retire or purchase any capital stock provided that (i) Borrower may convert any of its convertible securities into other securities pursuant to the terms of such convertible securities or otherwise in exchange thereof, (ii) Borrower may pay dividends solely in common stock; and (iii) Borrower may repurchase the stock of existing and former employees or consultants pursuant to board approved stock repurchase agreements so long as (X) an Event of Default does not exist at the time of such repurchase and would not exist after giving effect to such repurchase, and (Y) such repurchase does not exceed Two Hundred Fifty Thousand Dollars (\$250,000.00) in the aggregate in any fiscal year, and (iii) Borrower may make the GSE Earn-Out Payments so long as (X) an Event of Default does not exist at the time of such payment and would not exist immediately after giving effect to such payment and (Y) Borrower provides evidence satisfactory to Bank, in its sole discretion, that Borrower will remain in compliance with the Minimum Cash and Availability financial covenant contained in Section 6.9(a) both immediately after giving effect to such payment and as of the last day of the month in which such payment is made." and inserting in lieu thereof the following:
 - "7.7 Investments; Distributions. (a) Directly or indirectly make any Investment other than Permitted Investments, or permit any of its Subsidiaries to do so; (b) pay any dividends or make any distribution or payment or redeem, retire or purchase any capital

stock provided that (i) Borrower may convert any of its convertible securities into other securities pursuant to the terms of such convertible securities or otherwise in exchange thereof, (ii) Borrower may pay dividends solely in common stock; and (iii) Borrower may repurchase the stock of existing and former employees or consultants pursuant to board approved stock repurchase agreements so long as (X) an Event of Default does not exist at the time of such repurchase and would not exist after giving effect to such repurchase, and (Y) such repurchase does not exceed Two Hundred Fifty Thousand Dollars (\$250,000.00) in the aggregate in any fiscal year, and (iii) Borrower may make the GSE Earn-Out Payments and the NEP Earn-Out Payments so long as, in each case (X) an Event of Default does not exist at the time of such payment and would not exist immediately after giving effect to such payment and (Y) Borrower provides evidence satisfactory to Bank, in its sole discretion, that Borrower will remain in compliance with the Minimum Cash and Availability financial covenant contained in Section 6.9(a) both immediately after giving effect to such payment and as of the last day of the month in which such payment is made."

- 5 The Loan Agreement shall be amended by deleting the following text appearing as Section 7.9 thereof:
 - "7.9 Subordinated Debt. (a) Make or permit any payment on any Subordinated Debt, except under the terms of the subordination, intercreditor, or other similar agreement to which such Subordinated Debt is subject; provided however. Borrower may make scheduled payments under the NES Note so long as (i) an Event of Default does not exist at the time of such payment and would not exist immediately after giving effect to such payment and (ii) Borrower provides evidence satisfactory to Bank, in its sole discretion, that Borrower will remain in compliance with the Minimum Cash and Availability financial covenant contained in Section 6.9(a) both immediately after giving effect to such payment and as of the last day of the month in which such payment is made, or (b) amend any provision in any document relating to the Subordinated Debt which would increase the amount thereof or adversely affect the subordination thereof to Obligations owed to Bank." and inserting in lieu thereof the following:
 - "7.9 Subordinated Debt. (a) Make or permit any payment on any Subordinated Debt, except under the terms of the subordination, intercreditor, or other similar agreement to which such Subordinated Debt is subject; provided however. Borrower may make regularly scheduled payments under the NES Note and the NEP Note so long as, in each case (i) an Event of Default does not exist at the time of such payment and would not exist immediately after giving effect to such payment and (ii) Borrower provides evidence satisfactory to Bank, in its sole discretion, that Borrower will remain in compliance with the Minimum Cash and Availability financial covenant contained in Section 6.9(a) both immediately after giving effect to such payment and as of the last day of the month in which such payment is made, or (b) amend any provision in any document relating to the Subordinated Debt which would increase the amount thereof or adversely affect the subordination thereof to Obligations owed to Bank."

6 The Loan Agreement shall be amended by deleting the following text appearing in Section 10 of the Loan Agreement:

"If to Borrower: World Energy Solutions, Inc.

World Energy Securities Corp. c/o World Energy Solutions, Inc.

446 Main Street

Worcester, Massachusetts 01608

Attn: Jim Parslow, CFO Fax: (508) 459-8101

Email: jparslow@worldenergy.com

with a copy to:

Wilmer Cutler Pickering Hale and Dorr LLP

60 State Street

Boston, Massachusetts

Attn: Mr. Mitchel Appelbaum

Fax: (617) 526-5000

Email: mitchel.appelbaum@wilmerhale.com"

and inserting in lieu thereof the following:

"If to Borrower:

World Energy Solutions, Inc. World Energy Securities Corp. c/o World Energy Solutions, Inc.

100 Front Street

Worcester, Massachusetts 01608

Attn: Jim Parslow, CFO Fax: (508) 459-8101

Email: jparslow@worldenergy.com"

- 7 The Loan Agreement shall be amended by deleting the following definition appearing in Section 13.1thereof:
 - "Credit Extension" is any Advance, Term Loan 2012, or any other extension of credit by Bank for Borrower's benefit.
 - "Fixed Charges" are, for any period of measurement, the sum of Borrower's (i) Interest Expense, <u>plus</u> (ii) any required or optional principal payments on outstanding Indebtedness owed to Bank (including, without limitation, principal amortization and prepayments of the Term Loan 2012, but specifically excluding (A) payments of principal on the Revolving Line and (B) GSE Earn-Out Payments permitted to be paid pursuant to Section 7.7 herein and scheduled payments under the NES Note permitted to be paid pursuant to Section 7.9 herein and paid during such period.
 - "Loan Documents" are, collectively, this Agreement, any Bank Services Agreements, the Perfection Certificate, the Subordination Agreement, if any, any note, or notes or guaranties executed by Borrower, or any other present or future agreement executed by Borrower, if any, and/or for the benefit of Bank in connection with this Agreement and/or Bank Services, all as amended, restated, or otherwise modified.
 - "Revolving Maturity Date" is March 15, 2013.

and inserting in lieu thereof the following:

"Credit Extension" is any Advance, Term Loan 2012A, or any other extension of credit by Bank for Borrower's benefit.

- "Fixed Charges" are, for any period of measurement, the sum of Borrower's (i) Interest Expense, <u>plus</u> (ii) any required or optional principal payments on outstanding Indebtedness owed to Bank (including, without limitation, principal amortization and prepayments of the Term Loan 2012A, but specifically excluding (A) payments of principal on the Revolving Line and (B) (i) GSE Earn-Out Payments and NEP Earn-Out Payments permitted to be paid pursuant to Section 7.7 herein and (ii) regularly scheduled payments under the NES Note and the NEP Note permitted to be paid pursuant to Section 7.9 herein and paid during such period.
- "Loan Documents" are, collectively, this Agreement, any Bank Services Agreements, the Perfection Certificate, the Warrant, any Subordination Agreement, if any, any note, or notes or guaranties executed by Borrower, or any other present or future agreement executed by Borrower, if any, and/or for the benefit of Bank in connection with this Agreement and/or Bank Services, all as amended, restated, or otherwise modified.
- "Revolving Line Maturity Date" is March 14, 2014.
- The Loan Agreement shall be amended by inserting the following definitions in Section 13.1 thereof each in their appropriate alphabetical position:
 - "Fourth Loan Modification Effective Date" is October 3, 2012
 - "NEP Earn-Out Payments" are the regularly scheduled earn-out payments as and when due and payable in accordance with the terms of Section 1.7 of that certain Asset Purchase Agreement by and among World Energy Solutions, Inc. and Northeast Energy Partners, LLC, entered into as of October 3, 2012.
 - "NEP Acquisition" means the transactions contemplated and consummated pursuant to the NEP Acquisition Agreement.
 - "NEP Acquisition Agreement" means that certain Asset Purchase Agreement, by and between World Energy Solutions Inc. in favor of Northeast Energy Partners, LLC, dated as of the Fourth Loan Modification Effective Date.
 - "NEP Note" is that certain Promissory Note, dated October 3, 2012, made by World Energy Solutions Inc. in favor of Northeast Energy Partners, LLC in the original principal amount equal to Two Million Dollars (\$2,000,000).
 - "Term Loan 2012A" is a loan made by Bank pursuant to the terms of Section 2.1.6 hereof.
 - "Term Loan 2012A Amount" is an aggregate amount equal to Four Million Dollars (\$4,000,000).
 - "Term Loan 2012A Maturity Date" is the earliest of (a) March 1, 2016 or (b) the occurrence of an Event of Default.
 - "Term Loan 2012A Payment" is defined in Section 2.1.6(b).
 - "Warrant" is that certain Warrant to Purchase Stock dated as of the Fourth Loan Modification Effective Date, executed by Borrower in favor of Bank.

- 9 The Loan Agreement shall be amended by deleting the following definitions appearing in Section 13.1 thereof:
 - "Term Loan 2012" is a loan made by Bank pursuant to the terms of Section 2.1.5 hereof.
 - "Term Loan 2012 Amount" is an aggregate amount equal to Two Million Five Hundred Thousand Dollars (\$2,500,000).
 - "Term Loan 2012 Maturity Date" is the earliest of (a) February 1, 2016 or (b) the occurrence of an Event of Default.
 - "Term Loan 2012 Payment" is defined in Section 2.1.5(b).
- The Loan Agreement shall be amended by deleting the following text appearing as clause (c) of the definition of "Permitted Indebtedness" in Section 13.1 thereof:
 - "(c) Subordinated Debt, including, without limitation, Indebtedness owed to Northeast Energy Solutions, LLC pursuant to the NES Note;"
 - and inserting in lieu thereof the following:
 - "(c) Subordinated Debt, including, without limitation, Indebtedness owed to Northeast Energy Solutions, LLC pursuant to the NES Note and Indebtedness owed to Northeast Energy Partners, LLC pursuant to the NEP Note;"
- The Loan Agreement shall be amended by deleting the following text appearing as clause (i) of the definition of "Permitted Indebtedness" in Section 13.1 thereof:
 - "(i) the GSE Earn-Out Payments"
 - and inserting in lieu thereof the following:
 - "(i) the GSE Earn-Out Payments and the NEP Earn-Out Payments;"
- The Compliance Certificate appearing as <u>Exhibit B</u> to the Loan Agreement is hereby replaced with the Compliance Certificate attached as <u>Exhibit A</u> hereto.
- 4. <u>CONDITIONS PRECEDENT</u>. Borrower hereby agrees that the following documents shall be delivered to the Bank prior to or concurrently with the execution of this Loan Modification Agreement, each in form and substance satisfactory to the Bank (collectively, the "Conditions Precedent"):
 - A. copies, certified by a duly authorized officer of Borrower, to be true and complete as of the date hereof, of each of (i) the governing documents of Borrower as in effect on the date hereof (but only to the extent modified since last delivered to the Bank), (ii) the resolutions of Borrower authorizing the execution and delivery of this Loan Modification Agreement, the other documents executed in connection herewith and Borrower's performance of all of the transactions contemplated hereby (but only to the extent required since last delivered to Bank), and (iii) an incumbency certificate giving the name and bearing a specimen signature of each individual who shall be so authorized on behalf of Borrower (but only to the extent any signatories have changed since such incumbency certificate was last delivered to Bank);

- B. a good standing certificate of Borrower certified by the Secretary of State of the applicable state of incorporation or organization of Borrower, together with a certificate of foreign qualification from the appropriate authority in each jurisdiction in which Borrower is so qualified, in each case dated as of a date no earlier than thirty (30) days prior to the date hereof;
- C. duly executed original signatures to the Warrant, together with the Borrower's capitalization table in effect as of the Fourth Loan Modification Effective Date:
- D. a Subordination Agreement, executed by Northeast Energy Partners, LLC;
- E. an Acknowledgment and Reaffirmation of Subordination Agreement, executed by Northeast Energy Solutions, LLC;
- F. updated Perfection Certificates executed by Borrower;
- G. <u>updated</u> evidence of insurance (Acord 27 or 28 naming Bank as loss payee and Acord 25 (including excess/umbrella liability) naming Bank as additional insured);
- H. certified copies, dated as of a recent date, of updated financing statement searches (including, without limitation, searches on the assets acquired in connection with the NEP Acquisition), as Bank shall request, accompanied by written evidence (including any UCC termination statements) that the Liens indicated in any such updated financing statements either constitute Permitted Liens or have been or, in connection with any Credit Extension, will be terminated or released;
- I. an officer's certificate, executed by Borrower and attaching copies of the fully-executed NEP Acquisition Agreement; and
- J. such other documents as Bank may reasonably request.
- 5. <u>FEES</u>. Borrower shall pay to Bank a commitment fee equal to Twenty Five Thousand Dollars (\$25,000), which fee shall be due on the date hereof and shall be deemed fully earned as of the date hereof. Borrower shall also reimburse Bank for all legal fees and expenses incurred in connection with the Existing Loan Documents and this Loan Modification Agreement.
- 6. <u>RATIFICATION OF NEGATIVE PLEDGE</u>. Borrower hereby ratifies, confirms and reaffirms, all and singular, the terms and conditions of a certain negative pledge arrangement with respect to Borrower's intellectual property, between Borrower and Bank, and Borrower acknowledges, confirms and agrees that said negative pledge arrangement remains in full force and effect.
- 7. <u>ADDITIONAL COVENANTS: RATIFICATION OF PERFECTION CERTIFICATE</u>. Borrower shall not, without providing the Bank with thirty (30) days prior written notice: (i) relocate its principal executive office or add any new offices or business locations or keep any Collateral, in excess of Ten Thousand Dollars (\$10,000.00) in any additional locations, or (ii) change its jurisdiction of organization, or (iii) change its organizational structure or type, (iv) change its legal name, or (v) change any organizational number (if any) assigned by its jurisdiction of organization. In addition, the Borrower hereby certifies that no Collateral in excess of Ten Thousand Dollars (\$10,000.00) is in the possession of any third party bailee (such as at a warehouse). In the event that Borrower, after the date hereof, intends to store or otherwise deliver the Collateral in excess of Ten Thousand Dollars (\$10,000.00) to such a bailee, then Borrower shall first receive, the prior written consent of Bank and such bailee must acknowledge in writing that the bailee is holding such Collateral for the benefit of Bank. Borrower hereby ratifies, confirms and reaffirms, all and singular, the terms and disclosures contained in a certain Perfection Certificate dated as of the date hereof, and acknowledges, confirms and agrees the disclosures and information above Borrower provided to Bank in the Perfection Certificate has not changed.

- 8. <u>AUTHORIZATION TO FILE</u>. Borrower hereby authorizes Bank to file UCC financing statements without notice to Borrower, with all appropriate jurisdictions, as Bank deems appropriate, in order to further perfect or protect Bank's interest in the Collateral, including a notice that any disposition of the Collateral, by either the Borrower or any other Person, shall be deemed to violate the rights of the Bank under the Code.
- 9. CONSISTENT CHANGES. The Existing Loan Documents are hereby amended wherever necessary to reflect the changes described above.
- 10. <u>RATIFICATION OF LOAN DOCUMENTS</u>. Borrower hereby ratifies, confirms, and reaffirms all terms and conditions of all security or other collateral granted to the Bank, and confirms that the indebtedness secured thereby includes, without limitation, the Obligations.
- 11. NO DEFENSES OF BORROWER. Borrower hereby acknowledges and agrees that Borrower has no offsets, defenses, claims, or counterclaims against Bank with respect to the Obligations, or otherwise, and that if Borrower now has, or ever did have, any offsets, defenses, claims, or counterclaims against Bank, whether known or unknown, at law or in equity, all of them are hereby expressly WAIVED and Borrower hereby RELEASES Bank from any liability thereunder.
- 12. CONTINUING VALIDITY. Borrower understands and agrees that in modifying the existing Obligations, Bank is relying upon Borrower's representations, warranties, and agreements, as set forth in the Existing Loan Documents. Except as expressly modified pursuant to this Loan Modification Agreement, the terms of the Existing Loan Documents remain unchanged and in full force and effect. Bank's agreement to modifications to the existing Obligations pursuant to this Loan Modification Agreement in no way shall obligate Bank to make any future modifications to the Obligations. Nothing in this Loan Modification Agreement shall constitute a satisfaction of the Obligations. It is the intention of Bank and Borrower to retain as liable parties all makers of Existing Loan Documents, unless the party is expressly released by Bank in writing. No maker will be released by virtue of this Loan Modification Agreement.
- 13. <u>RIGHT OF SET-OFF</u>. In consideration of Bank's agreement to enter into this Loan Modification Agreement, Borrower hereby reaffirms and hereby grants to Bank, a lien, security interest and right of set off as security for all Obligations to Bank, whether now existing or hereafter arising upon and against all deposits, credits, collateral and property, now or hereafter in the possession, custody, safekeeping or control of Bank or any entity under the control of Silicon Valley Bank (including a Bank subsidiary) or in transit to any of them. At any time after the occurrence and during the continuance of an Event of Default, without demand or notice, Bank may set off the same or any part thereof and apply the same to any liability or obligation of Borrower even though unmatured and regardless of the adequacy of any other collateral securing the loan. ANY AND ALL RIGHTS TO REQUIRE BANK TO EXERCISE ITS RIGHTS OR REMEDIES WITH RESPECT TO ANY OTHER COLLATERAL WHICH SECURES THE OBLIGATIONS, PRIOR TO EXERCISING ITS RIGHT OF SETOFF WITH RESPECT TO SUCH DEPOSITS, CREDITS OR OTHER PROPERTY OF BORROWER, ARE HEREBY KNOWINGLY, VOLUNTARILY AND IRREVOCABLY WAIVED.
- 14. <u>CONFIDENTIALITY</u>. Bank may use confidential information for the development of databases, reporting purposes, and market analysis, so long as such confidential information is aggregated and anonymized prior to distribution unless otherwise expressly permitted by Borrower. The provisions of the immediately preceding sentence shall survive the termination of the Loan Agreement.
- 15. JURISDICTION/VENUE. Section 11 of the Loan Agreement is hereby incorporated by reference in its entirety.
- 16. <u>COUNTERSIGNATURE</u>. This Loan Modification Agreement shall become effective only when it shall have been executed by Borrower and Bank.

[The remainder of this page is intentionally left blank]

This Loan Modification Agreement is executed as a sealed instrument under the laws of the Commonwealth of Massachusetts as of the date first written above.
BORROWER:
WORLD ENERGY SOLUTIONS, INC.

Ву	/s/ James Parslow
Name:	James Parslow
Title:	CFO
wori	LD ENERGY SECURITIES CORP.
Ву	/s/ James Parslow
Name:	James Parslow
Title:	Treasurer
BANK	: ON VALLEY BANK
_	4/5
By	/s/ Darren Gastrock
	Darren Gastrock
Title:	Relationship Manager

[Signature page to Fourth Loan Modification Agreements]

EXHIBIT B

COMPLIANCE CERTIFICATE

TO: SILICON VALLEY BANK

FROM: WORLD ENERGY SOLUTIONS, INC.

AND WORLD ENERGY SECURITIES

CORP.

Date:

The undersigned authorized officers of World Energy Solutions, Inc., and World Energy Securities Corp. (individually and collectively, jointly and severally, "Borrower"), solely in their capacities as officers of their respective entities, certify that under the terms and conditions of the Loan and Security Agreement between Borrower and Bank (the "Agreement"), (1) Borrower is in complete compliance for the period ending with all required covenants except as noted below, (2) there are no Events of Default, (3) all representations and warranties in the Agreement are true and correct in all material respects on this date except as noted below; provided, however, that such materiality qualifier shall not be applicable to any representations and warranties that already are qualified or modified by materiality in the text thereof; and provided, further that those representations and warranties expressly referring to a specific date shall be true, accurate and complete in all material respects as of such date, (4) Borrower, and each of its Subsidiaries, has timely filed all required tax returns and reports, and Borrower has timely paid all foreign, federal, state and local taxes, assessments, deposits and contributions owed by Borrower except as otherwise permitted pursuant to the terms of Section 5.9 of the Agreement, and (5) no Liens have been levied or claims made against Borrower or any of its Subsidiaries, if any, relating to unpaid employee payroll or benefits of which Borrower has not previously provided written notification to Bank. Attached are the required documents supporting the certification. The undersigned solely in their capacities as officers of their respective entities, certify that these are prepared in accordance with GAAP consistently applied from one period to the next except as explained in an accompanying letter or footnotes. The undersigned solely in their capacities as officers of their respective entities, acknowledge that no borrowings may be requested at any time or date of determination that Borrower is not in compliance with any of the terms of the Agreement, and that compliance is determined not just at the date this certificate is delivered. Capitalized terms used but not otherwise defined herein shall have the meanings given them in the Agreement.

Please indicate compliance status by circling Yes/No under "Complies" column.

Reporting Covenant	Required		plies_
Monthly financial statements with Compliance Certificate	Monthly within 30 days	Yes	No
Annual financial statement (CPA Audited) + CC	FYE within 90 days	Yes	No
10-Q, 10-K and 8-K	Within 5 days after filing with SEC	Yes	No
A/R & A/P Agings, Deferred Revenue report, and schedule of	Monthly within 20 days when there		
expected collections	are outstanding Credit Extensions		
	under the Revolving Line	Yes	No
Borrowing Base Certificate	Monthly within 20 Days	÷	
	during Streamline Period	Yes	No
Transaction Report	Weekly when not Streamline Period		
	when there are outstanding Credit Extensions under the		
	Revolving Line and upon each request for a Credit Extension		
	under the Revolving Line	Yes	No
Board-approved projections	Within 30 days of approval	Yes	Νo

Financial Covenant Minimum Cash and Availability Minimum Fixed Charge Coverage Ratio		Required \$3,000,000 1.25:1.00	Actual \$:1.00	Complies Yes No Yes No
The following financial covenant analyses and infethis Certificate. The following are the exceptions with respect to the				
WORLD ENERGY SOLUTIONS, INC.	BANK USE ONLY			
By: Name:			AUTHORIZ	ED SIGNER
WORLD ENERGY SECURITIES CORP. By:	Verified: Date:		AUTHORIZ	ED SIGNER
Name: Title:			Yes No	

Schedule 1 to Compliance Certificate

Financial Covenants of Borrower

Dated:	
I. Minimum Cash and Availability (Section 6.9(a))	
Required: Minimum Cash and Availability Commencing on the Fouriestricted cash of Borrower at Bank plus unused Availability Ar	ourth Loan Modification Effective Date and at all times thereafter, maintain mount of not less than Three Million Dollars (\$3,000,000);
Actual:	
A. Aggregate value of the unrestricted cash of Borrower at Bank	CAN CALL SOLL SOLL SOLL SOLL SOLL SOLL SOLL S
B. Unused Availability Amount	\$
C: Total (line A <u>plus</u> line B)	\$ames/V
Is line C equal to or greater than \$3,000,000?	
No, not in compliance	Yes, in compliance
	3

Effe	uired: Minimum Fixed Charge Coverage Ratio. Commencing with the first full fiscal quarter following the Fourth Loan Modification ective Date, achieve, on a trailing three month basis, measured on the last day of each monthly period, a Fixed Charge Coverage Ratio of less than 1.25:1.00.
Act	ual: All amounts measured on a trailing three month basis, ending as of the date of measurement.
Ã.	BBITDA:
B.	Taxes actually paid in cash
<u>C</u> .	Unfinanced Capital Expenditures
D.	Interest Expense \$
	The sum of required or principal payments on outstanding Indebtedness owed to Bank (including without limitation) # principal amortization and prepayments of the Term Loan 2012A, but specifically excluding (A) payments of principal # on the Revolving Line and (B) (i) GSE Earn-Out Payments and NEP Earn-Out Payments permitted to be paid pursuant # to Section 757 of the Loan Agreement and (ii) regularly scheduled payments under the NES Note and the NEP Note # permitted to be paid pursuant to Section 7.9 of the Loan Agreement and paid during such period \$
F.	FIXED CHARGE COVERAGE RATIO ((line A minus line B minus line C) divided by (line D plus line E)) :1.00
Is li	ne G greater than or equal to 1.25:1.00?
	No, not in complianceYes, in compliance

II.

Minimum Fixed Charge Coverage Ratio (Section 6.9(b))

WORLD ENERGY SOLUTIONS, INC.

Note Purchase Agreement

Dated as of October 3, 2012

ı

WORLD ENERGY SOLUTIONS, INC.

Note Purchase Agreement

Dated as of October 3, 2012

INDEX

ARTICLE I	Page 1
	1
PURCHASE, SALE AND TERMS OF NOTES	1
1.01. The Notes	1
1.02. Purchase and Sale of Notes	1
(a) The Closing	1
(b) Use of Proceeds	2
1.03. Payments and Endorsements	2
1.04. Redemptions	2
(a) Required Redemptions	2
(b) Optional Redemptions With Premium	2
(c) Notice of Redemptions; Pro rata Redemptions	2
1.05. Payment on Non-Business Days	3
1.06. Registration, etc	3
1.07. Transfer and Exchange of Notes	3
1.08. Replacement of Notes	4
1.09. Subordination	4
(a) Payment of Senior Debt	4
(b) No Payment on Notes Under Certain Conditions	5
(c) Payments Held in Trust	5
(d) Subrogation	5
(e) Scope of Section	5
(f) Survival of Rights	ϵ
(g) Amendment or Waiver	ϵ
(h) Senior Debt Defined	ϵ
1.10. Representations by the Purchaser	ϵ
1.11. Disclosure of Information by the Purchaser	7
ARTICLE II	7
CONDITIONS TO PURCHASER'S OBLIGATION	-
	<i>'</i>
2.01. Representations and Warranties	7
2.02. Documentation at Closing	7
ARTICLE III	8

REPRESENTATIONS AND WARRANTIES	8
3.01. Organization and Standing	8
3.02. Corporate Action	9
3.03. Governmental Approvals	9
3.04. Litigation	9
3.05. Compliance with Other Instruments	9
3.06. Federal Reserve Regulations	10
3.07. Title to Assets, Patents	10
3.08. Financial Information	10
3.09. Taxes	11
3.10. ERISA	11
3.11. Transactions with Affiliates	11
3.12. Assumptions or Guaranties of Indebtedness of Other Persons	11
3.13. Investments in Other Persons	12
3.14. Equal Employment Opportunity	12
3.15. Status of Notes as Qualified Investments	12
3.16. Securities Act	13
3.17. Disclosure	13
3.18. No Brokers or Finders	13
3.19. Other Agreements of Officers	13
3.20. Capitalization; Status of Capital Stock	13
3.21. Labor Relations	14
3.22. Insurance	14
3.23. Books and Records	14
3.24. Foreign Corrupt Practices Act	14
ARTICLE IV	14
COVENANTS OF THE COMPANY	14
4.01. Affirmative Covenants of the Company	14
(a) Punctual Payment	15
(b) Payment of Taxes and Trade Debt	15
(c) Maintenance of Insurance	15
(d) Preservation of Corporate Existence	15
(e) Compliance with Laws	15
(f) Visitation Rights	15
(g) Keeping of Records and Books of Account	16
(h) Maintenance of Properties, etc	16
(i) Compliance with ERISA	16
(j) Maintenance of Interest Coverage	16
(k) Foreign Corrupt Practices Act	16
(I) Equal Employment Opportunity	16
(m) Status of Notes as Qualified Investments	17
4.02. Negative Covenants of the Company	17
(a) Liens	17

 (b) Indebtedness (c) Lease Obligations (d) Assumptions or Guaranties of Indebtedness of Other Persons (e) Mergers, Sale of Assets, etc (f) Investments in Other Persons (g) Distributions (h) Dealings with Affiliates (i) Maintenance of Ownership of Subsidiaries (j) Change in Nature of Business 4.03. Reporting Requirements 	18 19 19 19 20 21 21 21
ARTICLE V	22
EVENTS OF DEFAULT	22
5.01. Events of Default 5.02. Annulment of Defaults	22 24
ARTICLE VI	24
DEFINITIONS AND ACCOUNTING TERMS	24
6.01. Certain Defined Terms 6.02. Accounting Terms	24 27
ARTICLE VII	28
MISCELLANEOUS	28
7.01. No Waiver; Cumulative Remedies 7.02. Amendments, Waivers and Consents 7.03. Addresses for Notices, etc 7.04. Costs, Expenses and Taxes 7.05. Binding Effect; Assignment 7.06. Survival of Representations and Warranties 7.07. Prior Agreements 7.08. Severability 7.09. Governing Law 7.10. Headings 7.11. Sealed Instrument 7.12. Counterparts	28 28 29 29 29 29 29 29 30 30
7.13. Further Assurances	30

EXHIBITS

1.01 Form of Subordinated Notes

SCHEDULES

- 2.02(b) Matters to be Covered by Opinion Letter
 3.01 Schedule of Subsidiaries
 3.04 Schedule of Litigation
 3.05 Schedule of Indebtedness for Money Borrowed
 3.07 Schedule of Mortgages, Pledges, etc.
 3.11 Schedule of Transactions with Affiliates
 3.13 Schedule of Investments in Other Persons
 3.15 Certificate re "Qualified Investments"

WORLD ENERGY SOLUTIONS, INC. 100 Front Street Worcester, Massachusetts 01608

As of October 3, 2012

Massachusetts Capital Resource Company 420 Boylston Street Boston, Massachusetts 02116

Re: Subordinated Notes due 2020

Gentlemen:

World Energy Solutions, Inc., a Delaware corporation (the "Company"), hereby agrees with Massachusetts Capital Resource Company (the "Purchaser") as follows:

ARTICLE I

PURCHASE, SALE AND TERMS OF NOTES

1.01. <u>The Notes</u>. The Company has authorized the issuance and sale to the Purchaser of the Company's Subordinated Notes, due September 30, 2020, in the original principal amount of \$4,000,000. The Subordinated Notes shall be substantially in the form set forth in <u>Exhibit 1.01</u> hereto and are herein referred to individually as a "Note" and collectively as the "Notes", which terms shall also include any notes delivered in exchange or replacement therefor.

1.02. Purchase and Sale of Notes.

(a) The Closing. The Company agrees to issue and sell to the Purchaser, and, subject to and in reliance upon the representations, warranties, terms and conditions of this Agreement, the Purchaser agrees to purchase, the Notes for an aggregate purchase price of \$4,000,000. Such purchase and sale shall take place at a closing (the "Closing") to be held at the Law Office of George W. Thibeault, 60 State Street, Suite 700, Boston, Massachusetts 02109, on October 3, 2012 at 10:00 A.M., or on such other date and at such time as may be mutually agreed upon. At the Closing the Company will initially issue one Note, payable to the order of the Purchaser, in the principal amount of \$4,000,000, against delivery to the Company of a check or a receipt of a wire transfer, in the amount of \$4,000,000, in payment of the full purchase price for the Notes.

- (b) <u>Use of Proceeds</u>. The Company agrees to use the full proceeds from the sale of the Notes solely to assist in acquisitions and agrees that such use would constitute a purpose which increases or maintains equal opportunity employment in the Commonwealth of Massachusetts.
- 1.03. <u>Payments and Endorsements</u>. Payments of principal, interest and premium, if any, on the Notes, shall be made directly by check duly mailed, wired or delivered to the Purchaser at its address referred to in Section 7.03 hereof, without any presentment or notation of payment, except that prior to any transfer of any Note, the holder of record shall endorse on such Note a record of the date to which interest has been paid and all payments made on account of principal of such Note.

1.04. Redemptions.

- (a) Required Redemptions. Beginning on and with October 31, 2016, and on the last day of each calendar month in each year thereafter through and including September 30, 2020, the Company will redeem, without premium, \$83,333.33 in principal amount of the Notes, or such lesser amount as may be then outstanding, together with all accrued and unpaid interest then due on the amount so redeemed. On the stated or accelerated maturity of the Notes, the Company will pay the principal amount of the Notes then outstanding together with all accrued and unpaid interest then due thereon. No optional redemption of less than all of the Notes shall affect the obligation of the Company to make the redemptions required by this subsection.
- (b) Optional Redemptions With Premium. The Company may at any time redeem the Notes in whole or in part (in integral multiples of \$50,000) together with interest due on the amount so redeemed through the date of redemption, and a premium equal to the percentage of the principal amount of the Notes redeemed under this subsection applicable to the period in which such redemption is made, as follows:

Period ending	Premium
October:1; 2013	The Control of an interpretation of the control of
October 1, 2014	3%
October 1, 2015	1%
Thereafter	0%

(c) Notice of Redemptions: Pro rata Redemptions. Notice of any optional redemptions pursuant to subsection 1.04(b) shall be given to all registered holders of the Notes at least ten (10) business days prior to the date of such redemption. Each redemption of Notes pursuant to subsections 1.04(a) or (b) shall be made so that the Notes then held by each holder shall be redeemed in a principal amount which shall bear the same ratio to the total principal amount of Notes being redeemed as the principal amount of Notes then held by such holder bears to the aggregate principal amount of the Notes then outstanding.

- 1.05. <u>Payment on Non-Business Days</u>. Whenever any payment to be made shall be due on a Saturday, Sunday or a public holiday under the laws of the Commonwealth of Massachusetts, such payment may be made on the next succeeding business day, and such extension of time shall in such case be included in the computation of payment of interest due.
- 1.06. Registration, etc. The Company shall maintain at its principal office a register of the Notes and shall record therein the names and addresses of the registered holders of the Notes, the address to which notices are to be sent and the address to which payments are to be made as designated by the registered holder if other than the address of the holder, and the particulars of all transfers, exchanges and replacements of Notes. No transfer of a Note shall be valid unless made on such register for the registered holder or his executors or administrators or his or their duly appointed attorney, upon surrender therefor for exchange as hereinafter provided, accompanied by an instrument in writing, in form and execution reasonably satisfactory to the Company. Each Note issued hereunder, whether originally or upon transfer, exchange or replacement of a Note or Notes, shall be registered on the date of execution thereof by the Company and shall be dated the date to which interest has been paid on such Notes or Note. The registered holder of a Note shall be that Person in whose name the Note has been so registered by the Company. A registered holder shall be deemed the owner of a Note for all purposes of this Agreement and, subject to the provisions hereof, shall be entitled to the principal, premium, if any, and interest evidenced by such Note free from all equities or rights of setoff or counterclaim between the Company and the transferor of such registered holder or any previous registered holder of such Note.
- 1.07. Transfer and Exchange of Notes. The registered holder of any Note or Notes may, prior to maturity or prepayment thereof, surrender such Note or Notes at the principal office of the Company for transfer or exchange, provided that the principal amount of any Note or Notes to be issued in connection with such transfer or exchange shall not be less than \$100,000 unless (y) such transfer or exchange is for the entire principal balance of the Notes then outstanding or (z) such transfer or exchange is to one or more partners of such holder. Within a reasonable time after notice to the Company from a registered holder of its intention to make such exchange and without expense (other than transfer taxes, if any) to such registered holder, the Company shall issue in exchange therefor another Note or Note, in such denominations as requested by the registered holder, for the same aggregate principal amount as the unpaid principal amount of the Note or Notes so surrendered, and having the same maturity and rate of interest, containing the same provisions and subject to the same terms and conditions as the Note or Notes so surrendered. Each new Note shall be made payable to such Person or Persons, or registered assigns, as the registered holder of such surrendered Note or Notes may designate, and such transfer or exchange shall be made in such a manner that no gain or loss of principal or interest shall result therefrom.

- 1.08. Replacement of Notes. Upon receipt of evidence satisfactory to the Company of the loss, theft, destruction or mutilation of any Note and, if requested in the case of any such loss, theft or destruction, upon delivery of an indemnity bond or other agreement or security reasonably satisfactory to the Company, or, in the case of any such mutilation, upon surrender and cancellation of such Note, the Company will issue a new Note, of like tenor and amount and dated the date to which interest has been paid, in lieu of such lost, stolen, destroyed or mutilated Note; provided, however, if any Note of which Massachusetts Capital Resource Company, its nominee, or any of its partners is the registered holder is lost, stolen or destroyed, the affidavit of the President, Treasurer or any Assistant Treasurer of the registered holder setting forth the circumstances with respect to such loss, theft or destruction shall be accepted as satisfactory evidence thereof, and no indemnification bond or other security shall be required as a condition to the execution and delivery by the Company of a new Note in replacement of such lost, stolen or destroyed Note other than the registered holder's written agreement to indemnify the Company.
- 1.09. Subordination. The Company, for itself, its successors and assigns, covenants and agrees, and the Purchaser and each successor holder of the Notes by his or its acceptance thereof, likewise covenants and agrees, that notwithstanding any other provision of this Agreement or the Notes, the payment of the principal of and interest on each and all of the Notes shall be subordinated in right of payment, to the extent and in the manner hereinafter set forth, to the prior payment in full of all Senior Debt (as hereinafter defined) at any time outstanding. The provisions of this Section 1.09 shall constitute a continuing representation to all Persons who, in reliance upon such provisions, become the holders of or continue to hold Senior Debt, and such provisions are made for the benefit of the holders of Senior Debt, and such holders are hereby made obligees hereunder the same as if their names were written herein as such, and they or any of them may proceed to enforce such provisions against the Company or against the holder of any Note without the necessity of joining the Company as a party.
- (a) Payment of Senior Debt. In the event of any insolvency or bankruptcy proceedings, or any receivership, liquidation, reorganization or other similar proceedings in connection therewith, relative to the Company or to its property, or, in the event of any proceedings for voluntary liquidation, dissolution or other winding up of the Company or distribution or marshalling of its assets or any composition with creditors of the Company, whether or not involving insolvency or bankruptcy, then and in any such event all Senior Debt shall be paid in full before any payment or distribution of any character, whether in cash, securities or other property, shall be made on account of the Notes; and any such payment or distribution, except securities which are subordinated and junior in right of payment to the payment of all Senior Debt then outstanding in terms of substantially the same tenor as this Section 1.09, which would, but for the provisions hereof, be payable or deliverable in respect of the Notes shall be paid or delivered directly to the holders of Senior Debt (or their duly authorized representatives), in the proportions in which they hold the same, until all Senior Debt shall have been paid in full, and every holder of the Notes by becoming a holder thereof shall have designated and appointed the holder or holders of Senior Debt (and their duly authorized representatives) as his or its agents and attorney-in-fact to demand, sue for, collect and receive such Senior Debt holder's ratable share of all such payments and distributions and to file any necessary proof of claim therefor and to take all such other action in the name of the holders of the Notes or otherwise, as such Senior Debt holders (or their authorized representatives) may

determine to be necessary or appropriate for the enforcement of this Section 1.09. The Purchaser and each successor holder of the Notes by its or his acceptance thereof agrees to execute, at the request of the Company, a separate agreement with any holder of Senior Debt on the terms set forth in this Section 1.09, and to take all such other action as such holder or such holder's representative may request in order to enable such holder to enforce all claims upon or in respect of such holder's ratable share of the Notes.

- (b) No Payment on Notes Under Certain Conditions. In the event that any default occurs in the payment of the principal of or interest on any Senior Debt (whether as a result of the acceleration thereof by the holders of such Senior Debt or otherwise) and during the continuance of such default for a period up to ninety (90) days and thereafter if judicial proceedings shall have been instituted with respect to such defaulted payment, or (if a shorter period) until such payment has been made or such default has been cured or waived in writing by such holder of Senior Debt then and during the continuance of such event no payment of principal or interest on the Notes shall be made by the Company or accepted by any holder of the Notes who has received notice from the Company or from a holder of Senior Debt of such events.
- (c) <u>Payments Held in Trust</u>. In case any payment or distribution shall be paid or delivered to any holder of the Notes before all Senior Debt shall have been paid in full, despite or in violation or contravention of the terms of this subordination, such payment or distribution shall be held in trust for and paid and delivered ratably to the holders of Senior Debt (or their duly authorized representatives), until all Senior Debt shall have been paid in full.
- (d) <u>Subrogation</u>. Subject to the payment in full of all Senior Debt and until the Notes shall be paid in full, the holders of the Notes shall be subrogated to the rights of the holders of Senior Debt (to the extent of payments or distributions previously made to such holders of Senior Debt pursuant to the provisions of subsections (a) and (c) of this Section 1.09) to receive payments or distributions of assets of the Company applicable to the Senior Debt. No such payments or distributions applicable to the Senior Debt shall, as between the Company and its creditors, other than the holders of Senior Debt and the holders of the Notes, be deemed to be a payment by the Company to or on account of the Notes would be entitled except for the provisions of this Section 1.09 shall, as between the Company and its creditors, other than the holders of Senior Debt and the holders of the Notes, be deemed to be a payment by the Company to or on account of the Senior Debt.
- (e) Scope of Section. The provisions of this Section 1.09 are intended solely for the purpose of defining the relative rights of the holders of the Notes, on the one hand, and the holders of the Senior Debt, on the other hand. Nothing contained in this Section 1.09 or elsewhere in this Agreement or the Notes is intended to or shall impair, as between the Company, its creditors other than the holders of Senior Debt, and the holders of the Notes, the obligation of the Company, which is unconditional and absolute, to pay to the holders of the Notes the principal of and interest on the Notes as and when the same shall

become due and payable in accordance with the terms thereof, or to affect the relative rights of the holders of the Notes and creditors of the Company other than the holders of the Senior Debt, nor shall anything herein or therein prevent the holder of any Note from accepting any payment with respect to such Note or exercising all remedies otherwise permitted by applicable law upon default under such Note, subject to the rights, if any, under this Section 1.09 of the holders of Senior Debt in respect of cash, property or securities of the Company received by the holders of the Notes.

- (f) <u>Survival of Rights</u>. The right of any present or future holder of Senior Debt to enforce subordination of the Notes pursuant to the provisions of this Section 1.09 shall not at any time be prejudiced or impaired by any act or failure to act on the part of the Company or any such holder of Senior Debt, including, without limitation, any forbearance, waiver, consent, compromise, amendment, extension, renewal, or taking or release of security of or in respect of any Senior Debt or by noncompliance by the Company with the terms of such subordination regardless of any knowledge thereof such holder may have or otherwise be charged with.
- (g) <u>Amendment or Waiver</u>. The provisions of this Section 1.09 may not be amended or waived in any manner which is detrimental to any Senior Debt without the consent of the holders of all then existing Senior Debt.
- (h) <u>Senior Debt Defined</u>. The term "Senior Debt" shall mean (i) all Indebtedness of the Company for money borrowed from banks or other institutional lenders, including any extension or renewals thereof, whether outstanding on the date hereof or thereafter created or incurred, which is not by its terms subordinate and junior to or on a parity with the Notes and which is permitted hereby at the time it is created or incurred, and (ii) all guaranties by the Company which are not by their terms subordinate and junior to or on a parity with the Notes and which are permitted hereby at the time they are made, of Indebtedness of any Subsidiary if such Indebtedness would have been Senior Debt pursuant to the provisions of clause (i) of this sentence had it been Indebtedness of the Company. In making any loans which are (or the guaranties of which are) intended to be Senior Debt, the lenders or purchasers shall be entitled to rely as to the fact that such Indebtedness or guaranty is permitted hereby upon a certificate by the Company's chief financial officer purporting to show such Indebtedness or guaranty will not result in the Company's failure to comply with the provisions of Article IV hereof as of the date of the loan or guarantee.
- 1.10. Representations by the Purchaser. The Purchaser represents that it is its present intention to acquire the Notes for its own account and that the Notes are being and will be acquired for the purpose of investment and not with a view to distribution or resale thereof; <u>subject</u>, <u>nevertheless</u>, to the condition that the disposition of the property of the Purchaser shall at all times be within its control. The acquisition by the Purchaser of the Notes shall constitute a confirmation of this representation.

1.11. Disclosure of Information by the Purchaser. The Company understands that the Purchaser is a special purpose limited partnership organized under Chapter 109 of the General Laws of the Commonwealth of Massachusetts and Chapter 816 of the Acts and Resolves of 1977 of the Commonwealth of Massachusetts (the "Capital Resource Company Act"), and as such, in accordance with such provisions, the Purchaser, in order to obtain certain benefits for itself and its partners, is required to file certain reports and otherwise disclose information relating to the business, financial affairs, and future prospects of the Company and its affiliates (as defined in the aforesaid legislation) with the Clerk of the Senate and the Clerk of the House of Representatives of the General Court of the Commonwealth of Massachusetts, the Secretary of Manpower Affairs, the Commissioner of Insurance and the Department of Revenue of the Commonwealth of Massachusetts, and that such reports and other information may constitute "public records" within the purview of Section 7 of Chapter 4 of the General Laws of the Commonwealth of Massachusetts. In addition, information relating to the business, financial affairs and future prospects of the Company and its affiliates must be disclosed to others in order to obtain independent confirmation that financing on substantially similar terms to financing provided pursuant to this Agreement was not elsewhere available to the Company. The Company hereby authorizes the Purchaser to disclose all such information relating to the business, financial affairs and future prospects of the Company and its affiliates as has been or may in the future be presented to the Purchaser to all such persons as the Purchaser in good faith deems necessary or appropriate in order to fulfill its obligations under the Capital Resource Company Act.

ARTICLE II

CONDITIONS TO PURCHASER'S OBLIGATION

The obligation of the Purchaser to purchase and pay for the Notes at the Closing is subject to the following conditions:

- 2.01. <u>Representations and Warranties</u>. Each of the representations and warranties of the Company set forth in Article III hereof shall be true on the date of the Closing.
- 2.02. <u>Documentation at Closing</u>. The Purchaser shall have received prior to or at the Closing all of the following, each in form and substance satisfactory to the Purchaser and its counsel:
- (a) A certified copy of all charter documents of the Company; a certified copy of the resolutions of the Board of Directors and, to the extent required, the stockholders of the Company evidencing approval of this Agreement, the Notes, and other matters contemplated hereby; a certified copy of the By-laws of the Company; and certified copies of all documents evidencing other necessary corporate or other action and governmental approvals, if any, with respect to this Agreement, the Notes.

- (b) A favorable opinion of Mirick, O'Connell, DeMallie & Lougee, LLP, counsel for the Company, as to matters set forth in Schedule 2.02(b), and as to such other matters as the Purchaser, or its counsel, may reasonably request.
- (c) A certificate of the Secretary or an Assistant Secretary of the Company which shall certify the names of the officers of the Company, authorized to sign this Agreement, the Notes, and the other documents or certificates to be delivered pursuant to this Agreement by the Company, or any of its officers, together with the true signatures of such officers. The Purchaser may conclusively rely on such certificates until it shall receive a further certificate of the Secretary or an Assistant Secretary of the Company canceling or amending the prior certificate and submitting the signatures of the officers named in such further certificate.
- (d) A certificate from a duly authorized officer of the Company stating that the representations and warranties of the Company contained in Article III hereof and otherwise made by the Company in writing in connection with the transactions contemplated hereby are true and correct in all material respects and that no condition or event has occurred or is continuing or will result from execution and delivery of this Agreement or the Notes which constitute an Event of Default or would constitute an Event of Default but for the requirement that notice be given or time elapse or both.
- (e) A certificate, in the form attached as <u>Schedule 3.15</u> hereto, shall have been executed and delivered by a duly authorized officer of the Company.
- (f) Payment for the costs and expenses identified in Section 7.04 as to which the Purchaser gives the Company notice prior to the Closing.

ARTICLE III REPRESENTATIONS AND WARRANTIES

The Company represents and warrants as follows:

3.01. Organization and Standing. The Company and each Subsidiary is a duly organized and validly existing corporation in good standing under the laws of the jurisdiction in which it was organized and has all requisite corporate power and authority for the ownership and operation of its properties and for the carrying on of its business as now conducted and as now proposed to be conducted. The Company and each Subsidiary is duly licensed or qualified and in good standing as a foreign corporation authorized to do business in all jurisdictions wherein the character of the property owned or leased, or the nature of the activities conducted, by it makes such licensing or qualification necessary, except where failure to be so licensed or qualified would not have a material adverse effect on the Company or such Subsidiary. Attached hereto as Schedule 3.01 is a schedule which correctly identifies all Subsidiaries of the Company as of the date hereof and shows with respect to each Subsidiary its jurisdiction of incorporation. All of the outstanding capital stock of each Subsidiary has been duly authorized and validly issued, is fully paid and

nonassessable, and is owned beneficially and of record by the Company or by another Subsidiary as indicated in <u>Schedule 3.01</u>, free and clear of any lien, right, encumbrance or restriction of any nature, including, without limitation, any lien, right, encumbrance or restriction on transfer.

- 3.02. Corporate Action. The Company has all necessary corporate power and has taken all corporate action required to make all the provisions of this Agreement, the Notes, and any other agreements and instruments executed in connection herewith and therewith the valid and enforceable obligations they purport to be. The issuance of the Notes is not subject to preemptive or other similar statutory or contractual rights (which have not been waived) and will not conflict with any provisions of any agreement or instrument to which the Company or any Subsidiary is a party or by which it is bound.
- 3.03. Governmental Approvals. No authorization, consent, approval, license, exemption of or filing or registration with any court or governmental department, commission, board, bureau, agency or instrumentality, domestic or foreign, is or will be necessary for, or in connection with, the offer, issuance, sale, execution or delivery by the Company of, or for the performance by it of its obligations under, this Agreement or the Notes.
- 3.04. <u>Litigation</u>. Except as set forth on <u>Schedule 3.04</u>, there is no litigation or governmental proceeding or investigation pending or, to the Knowledge of the Company, threatened against the Company or any Subsidiary affecting any of its properties or assets, or against any officer, key employee or principal stockholder of the Company or any Subsidiary where such litigation, proceeding or investigation, either individually or in the aggregate, would have a material adverse effect on the Company or any Subsidiary, nor, to the Knowledge of the Company, has there occurred any event or does there exist any condition on the basis of which any litigation, proceeding or investigation might properly be instituted which would have a material adverse effect on the Company or any Subsidiary. Neither the Company nor any Subsidiary, nor, to the Knowledge of the Company, any officer or key employee of the Company or any Subsidiary, or principal stockholder of the Company or any Subsidiary, is in default with respect to any order, writ, injunction, decree, ruling or decision of any court, commission, board or other government agency affecting the Company or any Subsidiary. There are no actions or proceedings pending or threatened (or any basis therefor known to the Company) which might result, either in any case or in the aggregate, in any material adverse change in the business, operations, affairs or condition of the Company or any Subsidiary or in any of its properties or assets, or which might call into question the validity of this Agreement, the Notes, or any action taken or to be taken pursuant hereto or thereto.
- 3.05. Compliance with Other Instruments. The Company and each Subsidiary is in compliance in all respects with the terms and provisions of this Agreement and in all material respects with the terms and provisions of its charter and by-laws and mortgages, indentures, leases, agreements and other instruments and of all judgments, decrees, governmental orders, statutes, rules and regulations by which it is bound or to which its properties or assets are subject. There is no term or provision in any of the foregoing documents and instruments which materially adversely affects the business, assets or financial condition of the Company or any Subsidiary. Neither the execution and

delivery of this Agreement or the Notes, nor the consummation of any transactions contemplated hereby or thereby has constituted or resulted in or will constitute or result in a default or violation of any term or provision in any of the foregoing documents or instruments. A schedule of Indebtedness for borrowed money of the Company and each Subsidiary (including lease obligations required to be capitalized in accordance with applicable Statements of Financial Accounting Standards) is attached as Schedule 3.05.

- 3.06. Federal Reserve Regulations. Neither the Company nor any Subsidiary is engaged in the business of extending credit for the purpose of purchasing or carrying margin stock (within the meaning of Regulation G of the Board of Governors of the Federal Reserve System), and no part of the proceeds of the Notes will be used to purchase or carry any margin security or to extend credit to others for the purpose of purchasing or carrying any margin security or in any other manner which would involve a violation of any of the regulations of the Board of Governors of the Federal Reserve System.
- 3.07. Title to Assets, Patents. Except as is set forth in Schedule 3.07, the Company and each Subsidiary has good and clear record and marketable title in fee to such of its fixed assets as are real property, and title to all of its other assets, now carried on its books including those reflected in the most recent consolidated balance sheet of the Company and its Subsidiaries which forms a part of Schedule 3.08 attached hereto, or acquired since the date of such balance sheet (except personal property disposed of since said date in the ordinary course of business) free of any mortgages, pledges, charges, liens, security interests or other encumbrances except for liens permitted hereunder. The Company and each Subsidiary enjoys peaceful and undisturbed possession under all leases under which it is operating, and all said leases are valid and subsisting and in full force and effect. The Company and each Subsidiary owns or to its Knowledge has a valid right to use the patents, patent rights, licenses, permits, trade secrets, trademarks, trademark rights, trade names or trade name rights or franchises, copyrights, inventions and intellectual property rights being used to conduct its business as now operated and as now proposed to be operated; and to its Knowledge the conduct of its business as now operated and as now proposed to be operated, and to its Knowledge the conduct of its business as now operated and as now proposed to be operated, and to its Knowledge the conduct of its business as now operated and as now proposed to be operated, and to its Knowledge the conduct of its business as now operated and as now proposed to be operated, and to its Knowledge the conduct of its business as now operated and as now proposed to be operated, and to its Knowledge the conduct of its business as now operated and as now proposed to be operated, and to its Knowledge the conduct of its business as now operated and as now proposed to be operated, and to its Knowledge the conduct of its business, trademarks, trademarks, trade
- 3.08. Financial Information. The consolidated financial statements of the Company and its Subsidiaries set forth in the Company's Form 10-K for its fiscal year ended December 31, 2011 (the "Form 10-K") and in its Form 10-Q for the quarter ended June 30, 2012 (the "Form 10-Q"), both as filed with the Securities and Exchange Commission (the "SEC"), present fairly in all material respects the consolidated financial position of the Company and its Subsidiaries as at the dates thereof and its results of operations for the periods covered thereby and have been prepared in accordance with GAAP. The consolidated financial statements included: (i) in the Form 10-K are for the two years ended December 31, 2010 and December 31, 2011, as audited by Marcum, LLP, and (ii) in the Form 10-Q

are for the three months and six months ended June 30, 2012, being unaudited and subject to year-end adjustments consisting of normal recurring items which will not be material in the aggregate. Neither the Company nor any Subsidiary has any liability contingent or otherwise not disclosed in the aforesaid financial statements or in the notes thereto that would reasonably be expected, together with all such other liabilities, materially affect the financial condition of the Company or any Subsidiary, nor does the Company have any reasonable grounds to know of any such liability. Since the date of said December 31, 2011 audited financial statements, except as has been disclosed in the Company's reports filed with the SEC and which are publicly available on the SEC's Electronic Data Gathering, Analysis, and Retrieval (EDGAR) system (the "SEC Reports"), (i) there has been no material adverse change in the business, assets or condition, financial or otherwise, operations or prospects, of the Company or any Subsidiary; (ii) neither the business, condition, operations or prospects of the Company or any Subsidiary nor any of their properties or assets has been adversely affected as a result of any legislative or regulatory change, any revocation or change in any franchise, license or right to do business, or any other event or occurrence, whether or not insured against; and (iii) neither the Company nor any Subsidiary has entered into any material transaction or made any distribution on its capital stock except as is set forth in the Form 10-K or Form 10-Q.

- 3.09. Taxes. The Company and each Subsidiary has accurately prepared and timely filed all federal, state and other tax returns required by law to be filed by it, and all taxes shown to be due and all additional assessments have been paid or provision made therefor. The Company knows of no additional assessments or adjustments pending or threatened against the Company or any Subsidiary for any period, nor of any basis for any such assessment or adjustment.
- 3.10. ERISA. No employee benefit plan established or maintained, or to which contributions have been made, by the Company or any Subsidiary, which is subject to part 3 of Subtitle B of Title I of The Employee Retirement Income Security Act of 1974, as amended ("ERISA") had an accumulated funding deficiency (as such term is defined in Section 302 of ERISA) as of the last day of the most recent fiscal year of such plan ended prior to the date hereof, and no material liability to the Pension Benefit Guaranty Corporation has been incurred with respect to any such plan by the Company or any of its Subsidiaries.
- 3.11. <u>Transactions with Affiliates</u>. Except as set forth in <u>Schedule 3.11</u> attached hereto, there are no loans, leases, royalty agreements or other continuing transactions between the Company or any Subsidiary and any Person owning five percent (5%) or more of any class of capital stock of the Company or any Subsidiary or other entity controlled by such stockholder or a member of such stockholder's family.
- 3.12. <u>Assumptions or Guaranties of Indebtedness of Other Persons</u>. Neither the Company nor any Subsidiary has assumed, guaranteed, endorsed or otherwise become directly or contingently liable on (including, without limitation, liability by way of agreement, contingent or otherwise, to purchase, to provide funds for payment, to supply funds to or otherwise invest in the debtor or otherwise to assure the creditor against loss) any Indebtedness of any other Person.

- 3.13. <u>Investments in Other Persons</u>. Except as set forth in <u>Schedule 3.13</u> attached hereto, neither the Company nor any Subsidiary has made any loan or advance to any Person which is outstanding on the date of this Agreement, nor is the Company or any Subsidiary obligated or committed to make any such loan or advance, nor does the Company or any Subsidiary own any capital stock or assets comprising the business of, obligations of, or any interest in, any Person.
- 3.14. Equal Employment Opportunity. The Company has reviewed its employment practices and policies and those of each Subsidiary and, to its Knowledge, the Company and each Subsidiary is in full compliance with (a) all applicable laws of the United States, of the Commonwealth of Massachusetts and of each other applicable jurisdiction, relating to equal employment opportunity (including, without limitation, Title VII of the Civil Rights Act of 1964, as amended (42 U.S.C. §000e-17), the Age Discrimination in Employment Act of 1967, as amended (29 U.S.C. §§621-634), the Equal Pay Act of 1963 (29 U.S.C. §206(d)), and any rules, regulations and administrative orders and Executive Orders relating thereto; Mass. Gen. Laws. c. 151B, Mass. Gen. Laws c. 149 §24A et seq. and §105A et seq., and any rules or regulations relating thereto; and (b) the applicable terms, relating to equal employment opportunity, of any contract, agreement or grant the Company or any Subsidiary has with, from, or relating (by way of subcontract or otherwise) to any other contract, agreement or grant of, any federal or state governmental unit ("Government Contract"), including, without limitation, any terms required pursuant to Federal Executive Order No. 11246 and Massachusetts Executive Order No. 74 (both as amended). To the Company's Knowledge, it and each Subsidiary has kept all records required to be kept, and has filed all reports, affirmative action plans and forms (including, without limitation and where applicable, Form EEO-1) required to be filed pursuant to any such applicable law or the terms of any such Government Contract. Neither the Company nor any Subsidiary has been subject to any adverse final determination or order, with respect to any charge of employment discrimination made against it, by the United States Equal Employment Opportunity Commission, the Massachusetts Commission Against Discrimination or any other governmental unit (including, without limitation, any such governmental unit with which it has a Government Contract), and neither the Company nor any Subsidiary is presently, to the best of the Company's Knowledge, subject to any formal proceedings before, or investigations by, such commissions or governmental units.
- 3.15. Status of Notes as Qualified Investments. The Company has duly authorized the execution and delivery to the Purchaser on behalf of the Company of the certificate attached as Schedule 3.15 hereto, setting forth such statements, information and related data as are necessary to permit the Purchaser to determine and demonstrate that the Notes issued pursuant to this Agreement will constitute "qualified investments" within the meaning of that term as set forth in the Capital Resource Company Act and that the full proceeds of the Notes will be used for purposes which will materially increase or maintain equal opportunity employment in the Commonwealth of Massachusetts. All such statements, information and related data presented in such certificate that are not based on estimates and projections of future events are materially true and correct as of the date of such certificate and all such statements, information and related data based upon estimates or projections of future events have been carefully considered and prepared on behalf of the Company.

- 3.16. Securities Act. Neither the Company nor anyone acting on its behalf has offered any of the Notes, or similar securities, or solicited any offers to purchase or made any attempt by preliminary conversation or negotiations to dispose of the Notes, or similar securities, to any Person other than the Purchaser or the institutions described in Schedule 3.15. Neither the Company nor anyone acting on its behalf has offered or will offer to sell the Notes, or similar securities to, or solicit offers with respect thereto from, or enter into any preliminary conversations or negotiations relating thereto with, any Person, so as to bring the issuance and sale of the Notes under the registration provisions of the Securities Act.
- 3.17. <u>Disclosure</u>. Neither this Agreement, the Form 10-K, Form10-Q, the Schedule 14A, the Certificate set forth as <u>Schedule 3.15</u> hereof, nor any other agreement, or certificate furnished to the Purchaser or its counsel by or on behalf of the Company or any Subsidiary in connection with the transactions contemplated hereby contains any untrue statement of a material fact or omits to state a material fact necessary in order to make the statements contained herein or therein not misleading. Without limiting the foregoing, the Company has no Knowledge or belief that there exists, or there is pending or planned, any patent, invention, device, application or principle or any statute, rule, law, regulation, standard or code which would materially adversely affect the condition, financial or otherwise, or the operations of the Company or any Subsidiary.
- 3.18. No Brokers or Finders. No Person has or will have, as a result of the transactions contemplated by this Agreement, any right, interest or valid claim against or upon the Company or any Subsidiary for any commission, fee or other compensation as a finder or broker because of any act or omission by the Company or any Subsidiary or any agent of the Company or any Subsidiary.
- 3.19. Other Agreements of Officers. To the Knowledge of the Company, no officer or key employee of the Company or any Subsidiary is a party to or bound by any agreement, contract or commitment, or subject to any restrictions, particularly but without limitation in connection with any previous employment of any such person, which materially and adversely affects, or in the future would (so far as the Company can reasonably foresee) materially and adversely affect, the business or operations of the Company or any Subsidiary or the right of any such person to participate in the affairs of the Company or any Subsidiary. To the best of the Knowledge of the Company, no officer or key employee has any present intention of terminating his employment with the Company or any Subsidiary and neither the Company nor any Subsidiary has any present intention of terminating any such agreement.
- 3.20. <u>Capitalization: Status of Capital Stock</u>. The Company capitalization as of June 30, 2012 is as set forth in the Form 10-Q. A complete list of the "principal stockholders" of the Company and the number of shares of capital stock of the Company owed by them is set forth in Schedule 14A" as filed with the SEC in connection with the Company's annual meeting of stockholders

held on May 17, 2012. All the outstanding shares of capital stock of the Company have been duly authorized, are validly issued and are fully paid and nonassessable. Except as set forth in the Form 10-K and Form 10-Q, there are, as of the date of said Forms, no options, warrants or rights to purchase shares of capital stock or other securities of the Company authorized, issued or outstanding, nor is the Company obligated in any other manner to issue shares of its capital stock or other securities. There are no restrictions on the transfer of other shares of capital stock of the Company other than those imposed by relevant state and federal securities laws. The offer and sale of all shares of capital stock and other securities of the Company issued before the Closing complied with or were exempt from all federal and state securities laws.

- 3.21. <u>Labor Relations</u>. To the Knowledge of the Company, no labor union or any representative thereof has made any attempt to organize or represent employees of the Company or any Subsidiary. There are no unfair labor practice charges, pending trials with respect to unfair labor practice charges, pending material grievance proceedings or adverse decisions of a Trial Examiner of the National Labor Relations Board against the Company or any Subsidiary. Furthermore, to the Knowledge of the Company, relations with employees of the Company and each Subsidiary are good and there is no reason to believe that any labor difficulties will arise in the foreseeable future.
- 3.22. <u>Insurance</u>. The Company and each Subsidiary carries insurance covering its properties and business adequate and customary for the type and scope of the properties and business.
- 3.23. <u>Books and Records</u>. The books of account, ledgers, order books, records and documents of the Company and each Subsidiary are materially accurate and complete.
- 3.24. <u>Foreign Corrupt Practices Act</u>. To the Company's Knowledge and belief neither it nor any Subsidiary is engaged, nor has any officer, director, employee or agent of the Company or any Subsidiary engaged, in any act or practice which would constitute a violation of the Foreign Corrupt Practices Act of 1977, or any rules or regulations promulgated thereunder.

ARTICLE IV

COVENANTS OF THE COMPANY

4.01. <u>Affirmative Covenants of the Company</u>. Without limiting any other covenants and provisions hereof, the Company covenants and agrees that, as long as any of the Notes are outstanding, it will perform and observe the following covenants and provisions and will cause each Subsidiary to perform and observe such of the following covenants and provisions as are applicable to such Subsidiary:

- (a) <u>Punctual Payment</u>. Pay the principal of, premium, if any, and interest on each of the Notes at the times and place and in the manner provided in the Notes and herein.
- (b) Payment of Taxes and Trade Debt. Pay and discharge, and cause each Subsidiary to pay and discharge, all taxes, assessments and governmental charges or levies imposed upon it or upon its income or profits or business, or upon any properties belonging to it, prior to the date on which penalties attach thereto, which, if unpaid, would become a material lien or charge upon any properties of the Company or any Subsidiary, provided that neither the Company nor the Subsidiary shall be required to pay any such tax, assessment, charge, levy or claim which is being contested in good faith and by appropriate proceedings if the Company or Subsidiary concerned shall have set aside on its books adequate reserves with respect thereto.
- (c) <u>Maintenance of Insurance</u>. Maintain, and cause each Subsidiary to maintain, insurance with responsible and reputable insurance companies or associations in such amounts and covering such risks as is usually carried by companies engaged in similar businesses and owning similar properties in the same general areas in which the Company or such Subsidiary operates.
- (d) <u>Preservation of Corporate Existence</u>. Preserve and maintain, and cause each Subsidiary to preserve and maintain, its corporate existence, rights, franchises and privileges in the jurisdiction of its incorporation, and qualify and remain qualified, and cause each Subsidiary to qualify and remain qualified, as a foreign corporation in each jurisdiction in which such qualification is necessary or desirable in view of its business and operations or the ownership of its properties; <u>provided</u>, <u>however</u>, that nothing herein contained shall prevent any merger, consolidation or transfer of assets permitted by subsection 4.02(e). Preserve and maintain, and cause each Subsidiary to preserve and maintain, all licenses and other rights to use patents, processes, licenses, trademarks, trade names, inventions, intellectual property rights or copyrights owned or possessed by it and necessary to the conduct of its business.
- (e) <u>Compliance with Laws</u>. Materially comply, and cause each Subsidiary to materially comply, with all applicable laws, rules, regulations and orders of any governmental authority, noncompliance with which would have a material adverse effect on its business or condition, financial or other.
- (f) <u>Visitation Rights</u>. Upon at least fifteen (15) days' notice and from time to time, permit the Purchaser, at its expense, or any agents or representatives thereof, during normal operating hours to examine and make copies of and extracts from the records and books of account of, and visit and inspect the properties of, the Company and any Subsidiary, and to discuss the affairs, finances and accounts of the Company and any Subsidiary with any of their officers or directors and independent accountants.

- (g) <u>Keeping of Records and Books of Account</u>. Keep, and cause each Subsidiary to keep, adequate records and books of account, in which complete entries will be made in accordance with GAAP, reflecting all financial transactions of the Company and such Subsidiary, and in which, for each fiscal year, all proper reserves for depreciation, depletion, obsolescence, amortization, taxes, bad debts and other purposes in connection with its business shall be made.
- (h) <u>Maintenance of Properties, etc.</u> Maintain and preserve, and cause each Subsidiary to maintain and preserve, all of its properties, necessary or useful in the proper conduct of its business, in good repair, working order and condition, ordinary wear and tear excepted.
- (i) <u>Compliance with ERISA</u>. Comply, and cause each Subsidiary to comply, with all minimum funding requirements applicable to any pension or other employee benefit or employee contribution plans which are subject to ERISA or to the Internal Revenue Code of 1986, as amended (the "Code"), and comply, and cause each Subsidiary to comply, in all other material respects with the provisions of ERISA and the Code, and the rules and regulations thereunder, which are applicable to any such plan. Neither the Company nor any Subsidiary will permit any event or condition to exist which could permit any such plan to be terminated under circumstances which would cause the lien provided for in Section 4068 of ERISA to attach to the assets of the Company or any Subsidiary.
- (j) <u>Maintenance of Interest Coverage</u>. Maintain a ratio of Consolidated Net Earnings Available for Interest Charges to Interest Charges of not less than 1 to 1, such ratio to be measured at the end of each fiscal quarter of the Company as an average of the four (4) most recent fiscal quarters of the Company commencing on and with the fiscal quarter ending December 31, 2012.
- (k) Foreign Corrupt Practices Act. Comply, in all material respects, and cause each Subsidiary to comply, in all material respects, and cause each officer, director, employee and agent of the Company and each Subsidiary to comply, in all material respects, at all times with the prohibitions on certain acts and practices set forth in the Foreign Corrupt Practices Act of 1977, and any rules or regulations promulgated thereunder.
- (l) Equal Employment Opportunity. Comply, in all material respects, and cause each Subsidiary to comply, in all material respects, with all applicable laws of the United States, the Commonwealth of Massachusetts, and of each other applicable jurisdiction relating to equal employment opportunity, any rules, regulations, administrative orders and Executive Orders relating thereto and the applicable terms, relating to equal employment opportunity, of any Government Contract; and keep, and cause each Subsidiary to file, all reports, affirmative action plans and forms required to be filed, pursuant to any such applicable law or the terms of any such Government Contract; provided, however, the Company or any Subsidiary shall not be considered to have failed to comply with the foregoing during any period that any matter relating to the Company's or such Subsidiary's employment practices is being contested by the Company or such Subsidiary in appropriate proceedings, or thereafter, if the Company or such Subsidiary complies with any final determination issued in such proceedings.

- (m) Status of Notes as Qualified Investments. In the event that any of the statements, information and related data provided by or on behalf of the Company or any Subsidiary and relied upon by the Purchaser in determining that the Notes constitute "qualified investments" within the meaning of that term in the Capital Resource Company Act shall be put in issue in any formal or informal proceedings initiated or conducted by or on behalf of the Commonwealth of Massachusetts, the Company shall, upon reasonable notice and at its expense, provide, and, cause each Subsidiary to provide, such additional information, witnesses and related data as may be reasonably necessary or appropriate to support the representations and warranties set forth in Article III.
- 4.02. <u>Negative Covenants of the Company</u>. Without limiting any other covenants and provisions hereof, the Company covenants and agrees that, as long as any of the Notes are outstanding, it will comply with and observe the following covenants and provisions, and will cause each Subsidiary to comply with and observe such of the following covenants and provisions as are applicable to such Subsidiary, and will not:
- (a) <u>Liens</u>. Create, incur, assume or suffer to exist, or permit any Subsidiary to create, incur, assume or suffer to exist, any mortgage, deed of trust, pledge, lien, security interest or other charge or encumbrance (including the lien or retained security title of a conditional vendor) of any nature, upon or with respect to any of its properties, now owned or hereinafter acquired, or assign or otherwise convey any right to receive income, except that the foregoing restrictions shall not apply to mortgages, deeds of trust, pledges, liens, security interests or other charges or encumbrances:
 - (i) for taxes, assessments or governmental charges or levies on property of the Company or any Subsidiary if the same shall not at the time be delinquent or thereafter can be paid without penalty, or are being contested in good faith and by appropriate proceedings;
 - (ii) imposed by law, such as carriers', warehousemen's and mechanics' liens and other similar liens arising in the ordinary course of business;
 - (iii) arising out of pledges or deposits under workmen's compensation laws, unemployment insurance, old age pensions, or other social security or retirement benefits, or similar legislation;
 - (iv) securing the performance of bids, tenders, contracts (other than for the repayment of borrowed money), statutory obligations and surety bonds;
 - (v) in the nature of zoning restrictions, easements and rights or restrictions of record on the use of real property which do not materially detract from its value or impair its use;

- (vi) arising by operation of law in favor of the owner or sublessor of leased premises and confined to the property rented;
- (vii) arising from any litigation or proceeding which is being contested in good faith by appropriate proceedings, provided, however, that no execution or levy has been made;
- (viii) described in <u>Schedule 3.07</u> which secure the Indebtedness set forth in <u>Schedule 3.05</u>, provided that no such lien is extended to cover other or different property of the Company or any Subsidiary;
- (ix) arising out of a purchase money mortgage or security interest on personal property to secure the purchase price of such property (or to secure Indebtedness incurred solely for the purpose of financing the acquisition of any such property), provided that such purchase money mortgage or security interest does not extend to any other or different property of the Company or any Subsidiary;
 - (x) securing judgments not to exceed \$100,000 in the aggregate;
 - (xi) arising out of letters of credit up to a maximum amount of \$250,000 in the aggregate;
 - (xii) arising from Senior Debt obligations of the Company; and
 - (xiii) relating to any refinancing, extensions or renewals of the above.
- (b) <u>Indebtedness</u>. Create, incur, assume or suffer to exist, or permit any Subsidiary to create, incur, assume or suffer to exist, any liability with respect to Indebtedness except for:
 - (i) the Notes;
 - (ii) Indebtedness for money borrowed (including without limitation all Senior Debt), provided that such Indebtedness for money borrowed does not result in the Company's failure to comply with all of the provisions of Article IV hereof;
 - (iii) Current Liabilities, other than for borrowed money, which are incurred in the ordinary course of business;
 - (iv) Indebtedness with respect to lease obligations, provided that such lease obligations do not violate subsection 4.02(c);
 - (v) Indebtedness secured by permitted liens under Section 4.02(a); and
 - (vi) Extensions, refinancings, modifications, amendments and restatements of any items of listed in (i) through (v) above, provided that the principal amount thereof is not increased or the terms thereof are not modified to impose more burdensome terms upon the Company or its Subsidiary, as the case may be.

- (c) <u>Lease Obligations</u>. Create, incur, assume or suffer to exist, or permit any Subsidiary to create, incur, assume or suffer to exist, any obligations as lessee for the rental or hire of real or personal property in connection with any sale and leaseback transaction.
- (d) <u>Assumptions or Guaranties of Indebtedness of Other Persons</u>. Assume, guarantee, endorse or otherwise become directly or contingently liable on, or permit any Subsidiary to assume, guarantee, endorse or otherwise become directly or contingently liable on (including, without limitation, liability by way of agreement, contingent or otherwise, to purchase, to provide funds for payment, to supply funds to or otherwise invest in the debtor or otherwise to assure the creditor against loss) any Indebtedness of any other Person, except for guaranties by endorsement of negotiable instruments for deposit or collection in the ordinary course of business.
- (e) Mergers, Sale of Assets, etc. Merge or consolidate with, or sell, assign, lease or otherwise dispose of or voluntarily part with the control of (whether in one transaction or in a series of transactions) all or substantially all of its assets (whether now owned or hereinafter acquired) or sell, assign or otherwise dispose of (whether in one transaction or in a series of transactions) any of its accounts receivable (whether now in existence or hereinafter created) at a discount or with recourse, to, any Person (provided that it may settle in its sole discretion any customer accounts receivable with such customer), or permit any Subsidiary to do any of the foregoing, except for sales or other dispositions of assets in the ordinary course of business and except that (1) any Subsidiary may merge into or consolidate with or transfer assets to any other Subsidiary, (2) any Subsidiary may merge into or transfer assets to the Company, and (3) the Company may merge any Person into it or with or into any Subsidiary or otherwise acquire such Person as long as the Company is the surviving entity, such merger or acquisition does not result in the violation of any of the provisions of this Agreement and no such violation exists at the time of such merger or acquisition, and, provided that such merger or acquisition does not result in the issuance (in one or more transactions) of shares of the voting stock of the Company representing in the aggregate more than thirty percent (30%) of the total outstanding voting stock of the Company, on a fully diluted basis, immediately following the issuance thereof; unless simultaneously with such transaction, the Company repays the outstanding principal, interest and premium on all of the Notes.
- (f) <u>Investments in Other Persons</u>. Make or permit any Subsidiary to make, any loan or advance to any person, or purchase, otherwise acquire, or permit any Subsidiary to purchase or otherwise acquire, the capital stock, assets comprising the business of, obligations of, or any interest in, any Person, except:
 - (i) investments by the Company or a Subsidiary in evidences of indebtedness issued or fully guaranteed by the United States of America and having a maturity of not more than one year from the date of acquisition;

- (ii) investments by the Company or a Subsidiary in certificates of deposit, notes, acceptances and repurchase agreements having a maturity of not more than one year from the date of acquisition issued by a bank organized in the United States having capital, surplus and undivided profits of at least \$100,000,000 and whose parent holding company has long-term debt rated Aa1 or higher, and whose commercial paper (if rated) is rated Prime 1, by Moody's Investors Service, Inc.;
- (iii) investments by the Company or a Subsidiary in the highest-rated commercial paper having a maturity of not more than one year from the date of acquisition;
 - (iv) loans or advances from a Subsidiary to the Company; and
- (v) loans to persons to assist them in becoming a "channel partner" to the Company; provided that all of such loans shall not exceed \$250,000 in any fiscal year of the Company or more than \$1,000,000 at any one time outstanding.
- (g) <u>Distributions</u>. Declare or pay any dividends, purchase, redeem, retire, or otherwise acquire for value any of its capital stock (or rights, options or warrants to purchase such shares) now or hereafter outstanding, return any capital to its stockholders as such, or make any distribution of assets to its stockholders as such, or permit any Subsidiary to do any of the foregoing (such transactions being hereinafter referred to as "Distributions"), <u>except</u> that the Subsidiaries may declare and make payment of cash and stock dividends, return capital and make distributions of assets to the Company; <u>provided</u>, <u>however</u>, that nothing herein contained shall prevent the Company from:
 - (i) effecting a stock split or declaring or paying any dividend consisting of shares of any class of capital stock to the holders of shares of such class of capital stock,
 - (ii) issuance of shares of its capital stock upon a "cashless" exercise of options or warrants for such capital stock,
 - (iii) redeeming any stock of a deceased stockholder out of insurance held by the Company on that stockholder's life, or
 - (iv) repurchase any shares of its Common Stock from former employees or service providers to the Company in connection with their termination of service to the Company; provided, however, that the repurchase price shall not exceed the purchase price paid to the Company for the purchase of such shares,

if in the case of any such transaction there does not exist at the time of such Distribution an Event of Default or an event which, but for the requirement that notice be given or time elapse or both, would constitute an Event of Default and provided that such Distribution can be made in compliance with the other terms of this Agreement.

- (h) <u>Dealings with Affiliates</u>. Except as set forth in <u>Schedule 3.11</u> attached hereto, enter or permit any Subsidiary to enter into any transaction with any holder of 5% or more of any class of capital stock of the Company, or any member of their families or any corporation or other entity in which any one or more of such stockholders or members of their immediate families directly or indirectly holds five percent (5%) or more of any class of capital stock except in the ordinary course of business and on terms not less favorable to the Company or the Subsidiary than it would obtain in a transaction between unrelated parties.
- (i) <u>Maintenance of Ownership of Subsidiaries</u>. Sell or otherwise dispose of any shares of capital stock of any Subsidiary, except to the Company or another Subsidiary, or permit any Subsidiary to issue, sell or otherwise dispose of any shares of its capital stock or the capital stock of any Subsidiary, except to the Company or another Subsidiary, <u>provided</u>, <u>however</u>, that nothing herein contained shall prevent any merger, consolidation or transfer of assets permitted by subsection 4.02(e).
- (j) <u>Change in Nature of Business</u>. Make, or permit any Subsidiary to make, any material change in the nature of its business as carried on at the date hereof.
 - 4.03. Reporting Requirements. The Company will furnish to each registered holder of any Note:
- (a) as soon as possible and in any event within five (5) days after the occurrence of each Event of Default or each event which, with the giving of notice or lapse of time or both, would constitute an Event of Default, the statement of the chief financial officer of the Company setting forth details of such Event of Default or event and the action which the Company proposes to take with respect thereto;
- (b) as soon as available and in any event within forty-five (45) days after the end of each of the first three quarters of each fiscal year of the Company, consolidated and consolidating balance sheets of the Company and its Subsidiaries as of the end of such quarter and consolidated and consolidating statements of income and retained earnings and of changes in financial position of the Company and its Subsidiaries for the period commencing at the end of the previous fiscal year and ending with the end of such quarter, setting forth in each case in comparative form the corresponding figures for the corresponding period of the preceding fiscal year;
- (c) as soon as available and in any event within one hundred twenty (120) days after the end of each fiscal year of the Company, a copy of the annual audit report for such year for the Company and its Subsidiaries, including therein consolidated and consolidating balance sheets of the Company and its Subsidiaries as of the end of such fiscal year and consolidated statements of income and retained earnings and of changes in financial position of the Company and its Subsidiaries for such fiscal year, setting forth in each case in comparative form the corresponding figures for the preceding fiscal year, all duly certified by independent public accountants;

- (d) at the time of delivery of each quarterly and annual statement, a certificate, executed by the chief financial officer, stating that such officer has caused this Agreement and the Notes, to be reviewed and has no knowledge of any default by the Company or any Subsidiary in the performance or observance of any of the provisions of this Agreement or the Notes or, if such officer or accountant has such knowledge, specifying such default and the nature thereof. Each such certificate shall set forth computations in reasonable detail demonstrating compliance with the provisions of subsection 4.01(j) and subsections 4.02(b) and (c);
- (e) promptly upon receipt thereof, any written report submitted to the Company by independent public accountants in connection with an annual or interim audit of the books of the Company and its Subsidiaries made by such accountants;
- (f) within sixty (60) days after the start of each fiscal year, consolidated capital and operating expense budgets, cash flow projections and income and loss projections for the Company and its Subsidiaries in respect of such fiscal year, all itemized in reasonable detail and prepared on a monthly basis, and, promptly after preparation, any revisions to any of the foregoing;
- (g) promptly after the commencement thereof, notice of all actions, suits and proceedings before any court or governmental department, commission, board, bureau, agency or instrumentality, domestic or foreign, affecting the Company or any Subsidiary of the type described in Section 3.04; and
- (h) promptly after sending, making available, or filing the same, such reports and financial statements as the Company or any Subsidiary shall send or make available to the stockholders of the Company or the Securities and Exchange Commission and such other information respecting the business, properties or the condition or operations, financial or otherwise, of the Company or any of its Subsidiaries as the Purchaser may from time to time reasonably request.

ARTICLE V

EVENTS OF DEFAULT

- 5.01. Events of Default. If any of the following events ("Events of Default") shall occur and be continuing:
- (a) The Company shall fail to pay any installment of principal of any of the Notes when due and such failure shall continue for five (5) business days; or
- (b) The Company shall fail to pay any interest or premium on any of the Notes when due and such failure shall continue for five (5) business days; or

- (c) The Company shall default in the performance of any covenant contained in subsection 4.01(j) or shall default for thirty (30) days in the performance of any covenant contained in Section 4.02, which default is not cured within thirty (30) days; or
- (d) Any material representation or warranty made by the Company or any Subsidiary in this Agreement or by the Company or any Subsidiary (or any officers of the Company or any Subsidiary) in any certificate or instrument delivered pursuant to this Agreement, shall prove to have been incorrect when made in any material respect; or
- (e) The Company or any Subsidiary shall fail to perform or observe any other term, covenant or agreement contained in this Agreement or the Notes on its part to be performed or observed and any such failure remains unremedied for thirty (30) days after written notice thereof shall have been given to the Company by the registered holders of at least fifty percent (50%) of the principal amount of the Notes then outstanding; or
- (f) The Company or any Subsidiary shall fail to pay any Indebtedness for borrowed money (other than as evidenced by the Notes) owing by the Company or such Subsidiary (as the case may be), or any interest or premium thereon, when due (or, if permitted by the terms of the relevant document, within any applicable grace period), whether such Indebtedness shall become due by scheduled maturity, by required prepayment, by acceleration, by demand or otherwise, or shall fail to perform any term, covenant or agreement on its part to be performed under any agreement or instrument (other than this Agreement or the Notes) evidencing or securing or relating to any Indebtedness owing by the Company or any Subsidiary, as the case may be, when required to be performed (or, if permitted by the terms of the relevant document, within any applicable grace period), if the effect of such failure to pay or perform is to accelerate, or to permit the holder or holders of such Indebtedness, or the trustee or trustees under any such agreement or instrument to accelerate, the maturity of such Indebtedness, unless such failure to pay or perform shall be waived by the holder or holders of such Indebtedness or such trustee or trustees; or
- (g) The Company or any Subsidiary shall (i) admit in writing its inability to pay its debts generally as they become due; (ii) commence a voluntary case under Title 11 of the United States Code as from time to time in effect, or by its authorizing, by appropriate proceedings of its Board of Directors or other governing body, the commencement of such a voluntary case; (iii) file an answer or other pleading admitting or fail to deny the material allegations of a petition filed against it commencing an involuntary case under said Title 11, or seek, consent to or acquiesce in the relief therein provided, or fail to controvert timely the material allegations of any such petition; (iv) have entered against it an order for relief in any involuntary case commenced under said Title 11; (v) seek relief as a debtor under any applicable law, other than said Title 11, of any jurisdiction relating to the liquidation or reorganization of debtors or to the modification or alteration of the rights of creditors, or consent to or acquiesce in such relief; (vi) have entered against it an order by a court of competent jurisdiction (a) finding it to be bankrupt or insolvent, (b) ordering or approving its liquidation, reorganization or any modification or alteration of the rights of its creditors, or (c) assuming custody of, or appointing a receiver or other custodian for, all or a substantial part of its property; or (vii) make an assignment for the benefit of, or enter into a composition with, its creditors, or appoint or consent to the appointment of a receiver or other custodian for all or a substantial part of its property; or

- (h) Any judgment, writ, warrant of attachment or execution or similar process shall be issued or levied against all or substantially all of the property of the Company or any Subsidiary and such judgment, writ, or similar process shall not be released, vacated or fully bonded within ninety (90) days after its issue or levy; or
- (i) Any acquisition, reorganization, business combination, transfer, sale of capital stock or other transaction or series of transactions, in which the stockholders of the Company prior to such acquisition, reorganization, business combination, sale of capital stock or other transaction or series of transactions own less than fifty percent (50%) of the voting securities outstanding after such transaction (a "Change of Control");

then, and in any such event, the Purchaser or any other holder of the Notes may, by notice to the Company, declare the entire unpaid principal amount of the Notes, all interest accrued and unpaid thereon and all other amounts payable under this Agreement to be forthwith due and payable, whereupon the Notes, all such accrued interest and all such amounts shall become and be forthwith due and payable (unless there shall have occurred an Event of Default under subsection 5.01(g) in which case all such amounts shall automatically become due and payable), without presentment, demand, protest or further notice of any kind, all of which are hereby expressly waived by the Company.

5.02. Annulment of Defaults. Section 5.01 is subject to the condition that, if at any time after the principal of any of the Notes shall have become due and payable, and before any judgment or decree for the payment of the moneys so due, or any thereof, shall have been entered, all arrears of interest upon all the Notes and all other sums payable under the Notes and under this Agreement (except the principal of the Notes which by such declaration shall have become payable) shall have been duly paid, and every other default and Event of Default shall have been made good or cured, then and in every such case the holders of fifty percent (50%) or more in principal amount of all Notes then outstanding may, by written instrument filed with the Company, rescind and annul such declaration and its consequences; but no such rescission or annulment shall extend to or affect any subsequent default or Event of Default or impair any right consequent thereon.

ARTICLE VI

DEFINITIONS AND ACCOUNTING TERMS

6.01. <u>Certain Defined Terms</u>. As used in this Agreement, the following terms shall have the following meanings (such meanings to be equally applicable to both the singular and plural forms of the terms defined):

"Agreement" means this Note Purchase Agreement as from time to time amended and in effect between the parties.

"Capital Resource Company Act" shall have the meaning assigned to that term in Section 1.12.

"Change of Control" shall have the meaning assigned to that term in Section 5.01(i).

"Code" shall have the meaning assigned to that term in Section 4.01(i).

"Company" means and shall include World Energy Solutions, Inc. and its successors and assigns.

"Common Stock" includes the Company's Voting Common Stock, \$0.0001 par value per share, as authorized on the date of this Agreement and any other securities into which or for which any of such Voting Common Stock may be converted or exchanged pursuant to a plan of recapitalization, reorganization, merger, sale of assets or otherwise.

"Consolidated" and "consolidating" when used with reference to any term defined herein mean that term as applied to the accounts of the Company and its Subsidiaries consolidated in accordance with generally accepted accounting principles.

"Consolidated Net Earnings Available for Interest Charges" means, for any period, Consolidated Net Income for such period plus
(a) interest paid or accrued by the Company and its Subsidiaries with respect to all Indebtedness for such period, (b) income and excess profit taxes for such period and all other taxes for such period which are imposed on or measured by income after deduction of interest charges and (c) non-recurring transaction costs incurred in connection with acquisitions.

"Consolidated Net Income" means, for any period, the net income (or net deficit) of the Company and its Subsidiaries for such period, after all expenses, taxes and other proper charges, determined in accordance with generally accepted accounting principles eliminating (i) all intercompany items, (ii) all earnings attributable to equity interests in Persons that are not Subsidiaries unless actually received by the Company or its Subsidiaries, (iii) all income arising from the forgiveness, adjustment or negotiated settlement of any Indebtedness, and (iv) any increase or decrease of income arising from any change in the method of accounting for any item from that employed in the preparation of the financial statements attached hereto as Schedule 3.08.

"Consolidated Net Worth" means, at any dates, the sum of (a) the par value of all of the stock of the Company issued and outstanding, (b) the amount of any additional paid-in-capital, and (c)

- (i) the positive retained earnings, if any, of the Company and its Subsidiaries, or
- (ii) less, the amount of any deficit in the retained earnings of the Company and its Subsidiaries

as the same appears on a consolidated balance sheet of the Company and its Subsidiaries prepared in accordance with GAAP as of such date, after eliminating all intercompany items and all amounts properly attributable to (1) any write-up in the book value of any asset resulting from a revaluation thereof after the date of this Agreement; (2) the amount of any intangible assets including patents, trademarks, unamortized debt discount and expense, goodwill, covenants and agreements and the excess of the purchase price paid for assets or stock acquired over the value assigned thereto on the books of the Company or of the Subsidiary which shall have acquired the same; (3) earnings attributable to any other Person unless actually received by the Company or its Subsidiaries; and (4) changes in the method of accounting.

"Current Assets" means all assets of any corporation which would, in accordance with GAAP, be classified as current assets of a corporation conducting a business the same as or similar to that of such corporation, excluding, however, (1) any assets which have been pledged, assigned, mortgaged, hypothecated or otherwise voluntarily encumbered other than as permitted by the terms of this Agreement to secure any Indebtedness which is not included in Current Liabilities and (2) any receivables uncollected after ninety (90) days from the revenue recognition date.

"Current Liabilities" means all liabilities of any corporation which would, in accordance with GAAP, be classified as current liabilities of a corporation conducting a business the same as or similar to that of such corporation, including, without limitation, all rental payments due under leases required to be capitalized in accordance with applicable Statements of Financial Accounting Standards and fixed prepayments of, and sinking fund payments with respect to, Indebtedness (including Indebtedness evidenced by the Notes), which payments are required to be made within one year from the date of determination.

"Distribution" shall have the meaning assigned to that term in Section 4.02(g).

"ERISA" shall have the meaning assigned to that term in Section 3.10.

"Events of Default" shall have the meaning assigned to that term in Section 5.01.

"Exchange Act" means the Securities Exchange Act of 1934 or any similar federal statute, and the rules and regulations of the Securities and Exchange Commission (or of any other Federal Agency then administering the Exchange Act) thereunder, all as the same shall be in effect at the time.

"Form 10-Q" shall have the meaning assigned to that term in Section 3.08.

"Form 10-K" shall have the meaning assigned to that term in Section 3.08.

"GAAP" shall mean generally accepted accounting principles in the United States as in effect on the Closing and applied on a basis consistent with those used in the financial statements included in the Form 10-K.

"Government Contract" shall have the meaning assigned to that in Section 3.14.

"Indebtedness" means all obligations, contingent and otherwise, which should, in accordance with GAAP, be classified upon the obligor's balance sheet as liabilities, but in any event including, without limitation, liabilities secured by any mortgage on property owned or acquired subject to such mortgage, whether or not the liability secured thereby shall have been assumed, and also including, without limitation, (i) all guaranties, endorsements and other contingent obligations, in respect of Indebtedness of others, whether or not the same are or should be so reflected in said balance sheet, except guaranties by endorsement of negotiable instruments for deposit or collection or similar transactions in the ordinary course of business and (ii) the present value of any lease payments due under leases required to be capitalized in accordance with applicable Statements of Financial Accounting Standards, determined in accordance with applicable Statements of Financial Accounting Standards; but in no event will Indebtedness include leases which are not capital leases under current standards.

"Interest Charges" means the interest expense of the Company and its Subsidiaries on Indebtedness (including the current portion thereof).

"Knowledge" means the actual knowledge of the Company Chief Executive Officer or Chief Financial Officer.

"Notes" shall have the meaning assigned to that term in Section 1.01.

"Person" means an individual, corporation, partnership, joint venture, trust, or unincorporated organization, or a government or any agency or political subdivision thereof.

"Purchaser" means and shall include not only the Massachusetts Capital Resource Company but also any other holder or holders of any of the Notes.

"Schedule 14A" shall have the meaning assigned to that term in Section 3.20.

"Securities Act" means the Securities Act of 1933 or any similar Federal statute, and the rules and regulations of the Securities and Exchange Commission (or of any other Federal agency then administering the Securities Act) thereunder, all as the same shall be in effect at the time.

"Senior Debt" shall have the meaning assigned to that term in Section 1.09(h).

"Subsidiary" or "Subsidiaries" means any corporation or trust of which the Company and/or any of its other Subsidiaries (as herein defined) directly or indirectly owns at the time all of the outstanding shares of every class of such corporation or trust other than directors' qualifying shares.

6.02. Accounting Terms. All accounting terms not specifically defined herein shall be construed in accordance with generally accepted accounting principles consistent with those applied in preparation of the financial statements attached hereto as Schedule 3.08, and all financial data submitted pursuant to this Agreement and all financial tests to be calculated in accordance with this Agreement shall be prepared and calculated in accordance with such principles.

ARTICLE VII

MISCELLANEOUS

- 7.01. No Waiver; Cumulative Remedies. No failure or delay on the part of the Purchaser, or any other holder of the Notes in exercising any right, power or remedy hereunder shall operate as a waiver thereof; nor shall any single or partial exercise of any such right, power or remedy preclude any other or further exercise thereof or the exercise of any other right, power or remedy hereunder. The remedies herein provided are cumulative and not exclusive of any remedies provided by law.
- 7.02. Amendments, Waivers and Consents. Any provision in this Agreement or the Notes to the contrary notwithstanding, changes in or additions to this Agreement may be made, and compliance with any covenant or provision herein or therein set forth may be omitted or waived, if the Company shall, in the case of the Notes, obtain consent thereto in writing from the holder or holders of at least fifty percent (50%) in principal amount of all Notes then outstanding; provided that no such consent shall be effective to reduce or to postpone the date fixed for the payment of the principal (including any required redemption) or interest payable on any Note, without the consent of the holder thereof, or to reduce the percentage of the Notes the consent of the holders of which is required under this Section. Any waiver or consent may be given subject to satisfaction of conditions stated therein and any waiver or consent shall be effective only in the specific instance and for the specific purpose for which given. Written notice of any waiver or consent effected under this subsection shall promptly be delivered by the Company to any holders who did not execute the same.
- 7.03. <u>Addresses for Notices, etc.</u> All notices, requests, demands and other communications provided for hereunder shall be in writing (including telegraphic communication) and mailed or telegraphed or delivered to the applicable party at the addresses indicated below:

If to the Company:

World Energy Solutions, Inc. 100 Front Street Worcester, MA 01608 Attention: Chief Executive Officer

If to the Purchaser:

Payments should be mailed to:

Massachusetts Capital Resource Company P. O. Box 3707 Boston, Massachusetts 02241

and all other deliveries and other communications made at or sent to:

Massachusetts Capital Resource Company 420 Boylston Street Boston, Massachusetts 02116

Attention: President

If to any other holder of the Notes: at such holder's address for notice as set forth in the register maintained by the Company, or, as to each of the foregoing, at such other address as shall be designated by such Person in a written notice to the other party complying as to delivery with the terms of this Section. All such notices, requests, demands and other communications shall, when mailed or telegraphed, respectively, be effective when deposited in the mails or delivered to the telegraph company, respectively, addressed as aforesaid.

- 7.04. Costs, Expenses and Taxes. The Company agrees to pay on demand the reasonable fees and out-of-pocket expenses of the Law Office of George W. Thibeault, counsel for the Purchaser in connection with the preparation, execution and delivery of this Agreement, the Notes and other instruments and documents to be delivered hereunder, as well as the reasonable fees and out-of-pocket expenses of legal counsel, independent public accountants and other outside experts reasonably retained by the Purchaser in connection with the enforcement of this Agreement, the Notes, and other instruments and documents to be delivered hereunder or thereunder.
- 7.05. <u>Binding Effect: Assignment</u>. This Agreement shall be binding upon and inure to the benefit of the Company and the Purchaser and their respective successors and assigns, except that the Company shall not have the right to assign its rights hereunder or any interest herein without the prior written consent of the Purchaser.
- 7.06. <u>Survival of Representations and Warranties</u>. All representations and warranties made in this Agreement, the Notes, or any other instrument or document delivered in connection herewith or therewith, shall survive the execution and delivery hereof or thereof and the making of the loans.
- 7.07. <u>Prior Agreements</u>. This Agreement constitutes the entire agreement between the parties and supersedes any prior understandings or agreements concerning the subject matter hereof.
- 7.08. Severability. The invalidity or unenforceability of any provision hereof shall in no way affect the validity or enforceability of any other provision.
- 7.09. Governing Law. This Agreement shall be governed by, and construed in accordance with, the laws of the Commonwealth of Massachusetts.

- 7.10. <u>Headings</u>. Article, Section and subsection headings in this Agreement are included herein for convenience of reference only and shall not constitute a part of this Agreement for any other purpose.
 - 7.11. Sealed Instrument. This Agreement is executed as an instrument under seal.
- 7.12. <u>Counterparts</u>. This Agreement may be executed in any number of counterparts, all of which taken together shall constitute one and the same instrument, and each of the parties hereto may execute this Agreement by signing any such counterpart.
- 7.13. <u>Further Assurances</u>. From and after the date of this Agreement, upon the request of the Purchaser, the Company and each Subsidiary shall execute and deliver such instruments, documents and other writings as may be necessary or desirable to confirm and carry out and to effectuate fully the intent and purposes of this Agreement and the Notes.

BALANCE OF PAGE INTENTIONALLY LEFT BLANK

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed by their respective officers thereunto duly authorized, as of the date first above written.

WORLD ENERGY SOLUTIONS, INC.

By /s/ Philip Adams

Philip Adams, Chief Executive Officer

MASSACHUSETTS CAPITAL RESOURCE COMPANY

By /s/ Kenneth J. Lavery

Kenneth J. Lavery, Vice President

Exhibit 1.01

WORLD ENERGY SOLUTIONS, INC.

SUBORDINATED NOTE DUE 2020

\$4,000,000 October 3, 2012

For value received, World Energy Solutions, Inc., a Delaware corporation (the "Company"), hereby promises to pay to Massachusetts Capital Resource Company or registered assigns (hereinafter referred to as the "Payee"), on or before September 30, 2020, the principal sum of Four Million Dollars (\$4,000,000) or such part thereof as then remains unpaid pursuant to the terms set forth in that certain Note Purchase Agreement, dated as of October 3, 2012, between the Company and Massachusetts Capital Resource Company (as the same may be amended from time to time, hereinafter referred to as the "Agreement"), and to pay interest from the date hereof on the whole amount of said principal sum remaining from time to time unpaid at the rate of ten and one-half percent (10.5%) per annum, such interest to be payable monthly on the last day of each calendar month in each year, the first such payment to be due and payable on October 31, 2012, until the whole amount of the principal hereof remaining unpaid shall become due and payable, and to pay interest at the rate of fourteen percent (14%) (so far as the same may be legally enforceable) on all overdue principal (including any overdue required redemption), premium and interest. All or a portion of the principal amount of this Note must be redeemed in the amounts and at the times set forth in Section 1.04 of the Agreement. Principal, premium, if any, and interest shall be payable in lawful money of the United States of America, in immediately available funds, at the principal office of the Payee or at such other place as the legal holder may designate from time to time in writing to the Company. Interest shall be computed on the basis of a 360-day year and a 30-day month.

This Note is issued pursuant to and is entitled to the benefits of the Agreement, and each holder of this Note, by his acceptance hereof, agrees to be bound by the provisions of the Agreement, including, without limitation, that (i) this Note is subject to prepayment, in whole or in part, as specified in said Agreement, (ii) the principal of and interest on this Note is subordinated to Senior Debt, as defined in the Agreement and (iii) in case of an Event of Default, as defined in the Agreement, the principal of this Note may become or may be declared due and payable in the manner and with the effect provided in the Agreement.

As further provided in the Agreement, upon surrender of this Note for transfer or exchange, a new Note or new Notes of the same tenor dated the date to which interest has been paid on the surrender Note and in an aggregate principal amount equal to the unpaid principal amount of the Note so surrendered will be issued to, and registered in the name of, the transferee or transferees. The Company may treat the person in whose name this Note is registered as the owner hereof for the purpose of receiving payment and for all other purposes.

In case any payment herein provided for shall not be paid when due, the Company promises to pay all cost of collection, including all reasonable attorney's fees.

This Note shall be governed by, and construed in accordance with, the laws of the Commonwealth of Massachusetts and shall have the effect of a sealed instrument.

The Company and all endorsers and guarantors of this Note hereby waive presentment, demand, notice of nonpayment, protest and all other demands and notices in connection with the delivery, acceptance, performance or enforcement of this Note.

WORLD ENERGY SOLUTIONS, INC.

By Philip Adams, Chief Executive Officer

SUBORDINATION AGREEMENT

This Subordination Agreement (the "Agreement") is made as of October 3, 2012, by and between NORTHEAST ENERGY PARTNERS, LLC, a Connecticut limited liability company (the "Creditor"), and SILICON VALLEY BANK, a California-chartered bank, with its principal place of business at 3003 Tasman Drive, Santa Clara, California 95054 and with a loan production office located at 275 Grove Street, Suite 2-200, Newton, Massachusetts 02466 ("Bank").

Recitals

- A. WORLD ENERGY SOLUTIONS, INC., a Delaware corporation ("World Energy"), with offices located at 446 Main Street, Worcester, Massachusetts 01608, and WORLD ENERGY SECURITIES CORP., a Massachusetts securities corporation with offices located at 446 Main Street, Worcester, Massachusetts 01608 (together with World Energy, individually and collectively, jointly and severally, the "Borrower") has requested and/or obtained certain loans or other credit accommodations from Bank which are or may be from time to time secured by assets and property of Borrower.
- B. Creditor has extended loans or other credit accommodations to World Energy, and/or may extend loans or other credit accommodations to Borrower from time to time.
- C. To induce Bank to continue to extend credit to Borrower and, at any time or from time to time, at Bank's option, to make such further loans, extensions of credit, or other accommodations to or for the account of Borrower, or to purchase or extend credit upon any instrument or writing in respect of which Borrower may be liable in any capacity, or to grant such renewals or extension of any such loan, extension of credit, purchase, or other accommodation as Bank may deem advisable, Creditor is willing to subordinate: (i) all of Borrower's indebtedness to Creditor (including, without limitation, principal, premium (if any), interest, fees, charges, expenses, costs, professional fees and expenses, and reimbursement obligations), plus any dividends and/or distributions or other payments pursuant to call, put, or conversion features in connection with equity securities of Borrower issued to or held by Creditor, whether presently existing or arising in the future (the "Subordinated Debt") to all of Borrower's indebtedness and obligations to Bank; and (ii) all of Creditor's security interests, if any, to all of Bank's security interests in the Borrower's property.

NOW, THEREFORE, THE PARTIES AGREE AS FOLLOWS:

- 1. Creditor subordinates to Bank any security interest or lien that Creditor may have in any property of Borrower. Notwithstanding the respective dates of attachment or perfection of the security interests of Creditor and the security interests of Bank, all now existing and hereafter arising security interests of Bank in any property of Borrower and all proceeds thereof (the "Collateral"), including, without limitation, the "Collateral", as defined in a certain Loan and Security Agreement dated as of September 8, 2008, between Borrower and Bank, as amended by a certain First Loan Modification Agreement, dated as of September 30, 2009, as further amended by a certain Second Loan Modification Agreement, dated as of March 8, 2011, as further amended by a certain Third Loan Modification and Waiver Agreement, dated as of March 2, 2012 and as further amended by a certain Fourth Loan Modification Agreement, dated as of the date hereof (as may be further amended, modified, restated, replaced or supplemented from time to time, the "Loan Agreement"), shall at all times be senior to the security interest of Creditor.
- 2. All Subordinated Debt is subordinated in right of payment to all obligations of Borrower to Bank now existing or hereafter arising, together with all costs of collecting such obligations (including attorneys' fees), including, without limitation, all interest accruing after the commencement by or against Borrower of any bankruptcy, reorganization or similar proceeding, and all obligations under the Loan Agreement and/or any other agreement in connection with the provision by Bank to Borrower of certain products and/or credit services facilities, including, without limitation, any letters of credit, guidance facilities, cash management services (including, without limitation, merchant services, direct deposit of payroll, business credit cards, and check cashing services), interest rate swap arrangements, and foreign exchange services (such obligations, collectively, the "Senior Debt").

- 3. Creditor will not demand or receive from Borrower (and Borrower will not pay to Creditor) all or any part of the Subordinated Debt, by way of payment, prepayment, setoff, lawsuit or otherwise, nor will Creditor exercise any remedy with respect to the Collateral, nor will Creditor accelerate the Subordinated Debt, or commence, or cause to commence, prosecute or participate in any administrative, legal or equitable action against Borrower, until such time as (i) the Senior Debt is fully paid in cash, and (ii) Bank has no commitment or obligation to lend any further funds to Borrower, and (iii) all financing agreements between Bank and Borrower are terminated. The foregoing notwithstanding, provided that an Event of Default, as defined in the Loan Agreement, has not occurred and is not continuing and would not exist immediately after such payment, Creditor shall be entitled to receive (i) each regularly scheduled, non-accelerated payment of non-default interest or principal as and when due and payable in accordance with the terms of the Promissory Note made by World Energy, in the original principal amount equal to Two Million Dollars (\$2,000,000), dated as of the date hereof and attached as Exhibit A hereto, as in effect on the date hereof or as modified with the written consent of the Bank; and (ii) each regularly scheduled Earnout Payment as and when due and payable in accordance with the terms of Section 1.7 of that certain Asset Purchase Agreement by and among World Energy and Creditor, entered into as of October 3, 2012, an executed copy of which is attached as Exhibit B hereto. Nothing in the foregoing paragraph shall prohibit Creditor from converting all or any part of the Subordinated Debt into equity securities of Borrower; provided that, if such securities have any call, put or other conversion features that would obligate Borrower to declare or pay dividends, make distributions, or otherwise pay any money or deliver any other securities or consideration to the holder. Creditor hereby agrees that Borrower may not declare, pay or make such dividends, distributions or other payments to Creditor, and Creditor shall not accept any such dividends, distributions or other payments except as may be permitted in the Loan Agreement.
- 4. Creditor shall promptly deliver to Bank in the form received (except for endorsement or assignment by Creditor where required by Bank) for application to the Senior Debt any payment, distribution, security or proceeds received by Creditor with respect to the Subordinated Debt other than in accordance with this Agreement.
- 5. In the event of Borrower's insolvency, reorganization or any case or proceeding under any bankruptcy or insolvency law or laws relating to the relief of debtors, including, without limitation, any voluntary or involuntary bankruptcy, insolvency, receivership or other similar statutory or common law proceeding or arrangement involving Borrower, the readjustment of its liabilities, any assignment for the benefit of its creditors or any marshalling of its assets or liabilities (each, an "Insolvency Proceeding"), (a) this Agreement shall remain in full force and effect in accordance with Section 510(a) of the United States Bankruptcy Code, (b) the Collateral shall include, without limitation, all Collateral arising during or after any such Insolvency Proceeding, and (c) Bank's claims against Borrower and the estate of Borrower shall be paid in full before any payment is made to Creditor.
- 6. Creditor shall give Bank prompt written notice of the occurrence of any default or event of default under any document, instrument or agreement evidencing or relating to the Subordinated Debt, and shall, simultaneously with giving any notice of default to Borrower, provide Bank with a copy of any notice of default given to Borrower. Creditor acknowledges and agrees that any default or event of default under the Subordinated Debt documents shall be deemed to be a default and an event of default under the Senior Debt documents.
- 7. Until the Senior Debt has been fully paid in cash and Bank's agreements to lend any funds to Borrower have been terminated, Creditor irrevocably appoints Bank as Creditor's attorney-in-fact, and grants to Bank a power of attorney with full power of substitution, in the name of Creditor or in the name of Bank, for the use and benefit of Bank, without notice to Creditor, to perform at Bank's option the following acts in any Insolvency Proceeding involving Borrower:
 - a) To file the appropriate claim or claims in respect of the Subordinated Debt on behalf of Creditor if Creditor does not do so prior to thirty (30) days before the expiration of the time to file claims in such Insolvency Proceeding and if Bank elects, in its sole discretion, to file such claim or claims; and

b) To accept or reject any plan of reorganization or arrangement on behalf of Creditor and to otherwise vote Creditor's claims in respect of any Subordinated Debt in any manner that Bank deems appropriate for the enforcement of its rights hereunder.

In addition to and without limiting the foregoing: (x) until the Senior Debt has been fully paid in cash and Bank's agreements to lend any funds to Borrower have been terminated, Creditor shall not commence or join in any involuntary bankruptcy petition or similar judicial proceeding against Borrower, and (y) if an Insolvency Proceeding occurs: (i) Creditor shall not assert, without the prior written consent of Bank, any claim, motion, objection or argument in respect of the Collateral in connection with any Insolvency Proceeding which could otherwise be asserted or raised in connection with such Insolvency Proceeding, including, without limitation, any claim, motion, objection or argument seeking adequate protection or relief from the automatic stay in respect of the Collateral, (ii) Bank may consent to the use of cash collateral on such terms and conditions and in such amounts as it shall in good faith determine without seeking or obtaining the consent of Creditor as (if applicable) holder of an interest in the Collateral, (iii) if use of cash collateral by Borrower is consented to by Bank, Creditor shall not oppose such use of cash collateral on the basis that Creditor's interest in the Collateral (if any) is impaired by such use or inadequately protected by such use, or on any other ground, and (iv) Creditor shall not object to, or oppose, any sale or other disposition of any assets comprising all or part of the Collateral, free and clear of security interests, liens and claims of any party, including Creditor, under Section 363 of the United States Bankruptcy Code or otherwise, on the basis that the interest of Creditor in the Collateral (if any) is impaired by such sale or inadequately protected as a result of such sale, or on any other ground (and, if requested by Bank, Creditor shall affirmatively and promptly consent to such sale or disposition of such assets), if Bank has consented to, or supports, such sale or disposition of such assets.

- 8. Creditor represents and warrants that Creditor has provided Bank with true and correct copies of all of the documents evidencing or relating to the Subordinated Debt. Creditor shall immediately affix a legend to the instruments evidencing the Subordinated Debt stating that the instruments are subject to the terms of this Agreement. By the execution of this Agreement, Creditor hereby authorizes Bank to amend any financing statements filed by Creditor against Borrower as follows: "In accordance with a certain Subordination Agreement by and among the Secured Party, the Debtor and Silicon Valley Bank, the Secured Party has subordinated any security interest or lien that Secured Party may have in any property of the Debtor to the security interest of Silicon Valley Bank in all assets of the Debtor, notwithstanding the respective dates of attachment or perfection of the security interest of the Secured Party and Silicon Valley Bank."
- 9. No amendment of the documents evidencing or relating to the Subordinated Debt shall directly or indirectly modify the provisions of this Agreement in any manner which might terminate or impair the subordination of the Subordinated Debt or the subordination of the security interest or lien that Creditor may have in any property of Borrower. By way of example, such instruments shall not be amended to (a) increase the rate of interest with respect to the Subordinated Debt, or (b) accelerate the payment of the principal or interest or any other portion of the Subordinated Debt. Bank shall have the sole and exclusive right to restrict or permit, or approve or disapprove, the sale, transfer or other disposition of property of Borrower except in accordance with the terms of the Senior Debt. Upon written notice from Bank to Creditor of Bank's agreement to release its lien on all or any portion of the Collateral in connection with the sale, transfer or other disposition thereof by Bank (or by Borrower with consent of Bank), Creditor shall be deemed to have also, automatically and simultaneously, released its lien on the Collateral, and Creditor shall upon written request by Bank, immediately take such action as shall be necessary or appropriate to evidence and confirm such release. All proceeds resulting from any such sale, transfer or other disposition shall be applied first to the Senior Debt until payment in full thereof, with the balance, if any, to the Subordinated Debt, or to any other entitled party. If Creditor fails to release its lien as required hereunder, Creditor hereby appoints Bank as attorney in fact for Creditor with full power of substitution to release Creditor's liens as provided hereunder. Such power of attorney being coupled with an interest shall be irrevocable.
- 10. All necessary action on the part of Creditor, its officers, directors, partners, members and shareholders, as applicable, necessary for the authorization of this Agreement and the performance of all obligations of Creditor hereunder has been taken. This Agreement constitutes the legal, valid and binding obligation of Creditor, enforceable against Creditor in accordance with its terms. The execution, delivery and performance of and compliance with this Agreement by Creditor will not (a) result in any material violation or default of any term of any of Creditor's charter, formation or other organizational documents (such as Articles or Certificate of Incorporation, bylaws, partnership agreement, operating agreement, etc.) or (b) violate any material applicable law, rule or regulation.

- 11. If, at any time after payment in full of the Senior Debt any payments of the Senior Debt must be disgorged by Bank for any reason (including, without limitation, any Insolvency Proceeding), this Agreement and the relative rights and priorities set forth herein shall be reinstated as to all such disgorged payments as though such payments had not been made and Creditor shall immediately pay over to Bank all payments received with respect to the Subordinated Debt to the extent that such payments would have been prohibited hereunder. At any time and from time to time, without notice to Creditor, Bank may take such actions with respect to the Senior Debt as Bank, in its sole discretion, may deem appropriate, including, without limitation, terminating advances to Borrower, increasing the principal amount, extending the time of payment, increasing applicable interest rates, renewing, compromising or otherwise amending the terms of any documents affecting the Senior Debt and any collateral securing the Senior Debt, and enforcing or failing to enforce any rights against Borrower or any other person. No such action or inaction shall impair or otherwise affect Bank's rights hereunder. Creditor waives the benefits, if any, of any statutory or common law rule that may permit a subordinating creditor to assert any defenses of a surety or guarantor, or that may give the subordinating creditor the right to require a senior creditor to marshal assets, and Creditor agrees that it shall not assert any such defenses or rights.
- 12. This Agreement shall bind any successors or assignees of Creditor and shall benefit any successors or assigns of Bank; provided, however, Creditor agrees that, prior and as conditions precedent to Creditor assigning all or any portion of the Subordinated Debt: (a) Creditor shall give Bank prior written notice of such assignment, and (b) such successor or assignee, as applicable, shall execute a written agreement whereby such successor or assignee expressly agrees to assume and be bound by all terms and conditions of this Agreement with respect to Creditor. This Agreement shall remain effective until terminated in writing by Bank. This Agreement is solely for the benefit of Creditor and Bank and not for the benefit of Borrower or any other party. Creditor further agrees that if Borrower is in the process of refinancing any portion of the Senior Debt with a new lender, and if Bank makes a request of Creditor, Creditor shall agree to enter into a new subordination agreement with the new lender on substantially the terms and conditions of this Agreement.
- 13. Creditor hereby agrees to execute such documents and/or take such further action as Bank may at any time or times reasonably request in order to carry out the provisions and intent of this Agreement, including, without limitation, ratifications and confirmations of this Agreement from time to time hereafter, as and when requested by Bank.
- 14. This Agreement may be executed in two or more counterparts, each of which shall be deemed an original and all of which together shall constitute one instrument.
- 15. This Agreement shall be governed by and construed in accordance with the laws of the Commonwealth of Massachusetts, without giving effect to conflicts of laws principles. Creditor and Bank submit to the exclusive jurisdiction of the state and federal courts located in Boston, Massachusetts in any action, suit, or proceeding of any kind, against it which arises out of or by reason of this Agreement. CREDITOR AND BANK WAIVE THEIR RESPECTIVE RIGHTS TO A JURY TRIAL OF ANY CLAIM OR CAUSE OF ACTION BASED UPON OR ARISING OUT OF THIS AGREEMENT OR ANY OF THE TRANSACTIONS CONTEMPLATED HEREIN.
- 16. This Agreement represents the entire agreement with respect to the subject matter hereof, and supersedes all prior negotiations, agreements and commitments. Creditor is not relying on any representations by Bank or Borrower in entering into this Agreement, and Creditor has kept and will continue to keep itself fully apprised of the financial and other condition of Borrower. This Agreement may be amended only by written instrument signed by Creditor and Bank.

[Signature page follows.]

IN WITNESS WHEREOF, the undersigned have executed this Agreen of Massachusetts, as of the date first above written.	nent as a se	aled instrument under the laws of the Commonwealth
"Creditor"	"Bank	.39
NORTHEAST ENERGY PARTNERS, LLC	SILICO	N VALLEY BANK
By: /s/ John Hardy Name: John Hardy Title: A Member	By: Name: Title:	/s/ Darren Gastrock Darren Gastrock Relationship Manager
The undersigned acknowledge and approve the terms of this Agreement.		
"Borrower"		
WORLD ENERGY SOLUTIONS, INC.		
By /s/ James Parslow Name: James Parslow Title: CFO		
WORLD ENERGY SECURITIES CORP.		
By /s/ James Parslow Name: James Parslow		
Title: Treasurer		

Exhibit A

(See attached Subordinated Debt documents.)

Exhibit B

(See attached Asset Purchase Agreement.)



For Immediate Release

World Energy Solutions Acquires Northeast Energy Partners

Deal Broadens Company's Reach in Key "Mid-Market" Segment, Adds Significant Cross-Sell Opportunities for Efficiency Services

Hartford, CT and Worcester, MA – October 4, 2012 – World Energy Solutions, Inc. (NASDAQ: XWES), a leading energy management services firm, today announced it has acquired Northeast Energy Partners, LLC (NEP), a privately-held Enfield, CT-based energy management and procurement company, for approximately \$7.9 million in cash plus an additional \$5.2 million in seller notes and potential earn-outs. The acquisition expands World Energy's national footprint in the mid-market – addressing the energy management needs of small and mid-sized businesses – while increasing the Company's market share in New England.

The deal also creates a significant cross-sell opportunity for World Energy, opening the door for the Company to deliver its energy efficiency services to NEP's 2,000+ customers in Connecticut and Massachusetts. World Energy is an approved contractor of CL&P and United Illuminating in the CT Small Business Energy Advantage Program and in NSTAR's small business and municipal programs in MA. These states have some of the largest utility incentive programs in the country – nearly \$1 billion combined – which help fund energy efficiency projects.

"NEP is a perfect strategic fit for us," said Phil Adams, CEO of World Energy Solutions. "The acquisition advances our goal of becoming the national leader in energy procurement for the mid-market segment, building on our purchase last year of GSE Consulting. The deal also brings us a roster of new customers in utility territories with active incentive programs where we can cross-sell our energy efficiency services. By helping clients lower the price they pay for energy, reduce the amount they consume, and maximize available incentives, we are providing a winning formula for lowering total energy cost."

Added Jim Parslow, CFO of World Energy Solutions: "With the purchase of NEP, we are continuing to execute our growth strategy, supplementing strong organic growth with the acquisition of profitable, cash-generating companies that have predictable revenues and cash flows. Buying companies with these attributes allows us to use their cash flows to fund earn-outs and other considerations. We are financing this deal primarily with debt to minimize shareholder dilution, and we expect our purchase to be accretive and to have a positive impact on our top-line revenue, EBITDA and backlog."

Founded in 2000 in response to natural gas and electricity deregulation in the Connecticut and Massachusetts markets, NEP established itself as one of the top energy brokerages in New England, specializing in the needs of small and mid-size businesses. With annual revenues topping \$5 million, NEP currently employs 17, all of whom will be retained by World Energy. The Company will continue to operate in Enfield.

"Joining forces with a national energy management leader like World Energy will enable us to attract new customers, better serve our existing base, and provide our employees with exciting growth opportunities," said Russ Monroe, President, Northeast Energy Partners. "World Energy's approach to lowering a customer's total energy cost is a compelling one, and our combined capability to execute it is going to make a big impact on how energy management gets done here and across the country."

About World Energy Solutions, Inc.

World Energy Solutions, Inc. (NASDAQ: XWES) is an energy management services firm that brings together the passion, processes and technologies to take the complexity out of energy management and turn it into bottom-line impact for the businesses, institutions and governments we serve. To date, the Company has transacted more than \$30 billion in energy, demand response and environmental commodities on behalf of its customers, creating more than \$1 billion in value for them. World Energy is also a leader in the global carbon market, where its World Energy Exchange *supports the ground-breaking Regional Greenhouse Gas Initiative's (RGGI) cap and trade program for CO 2 emissions. For more information, please visit www.worldenergy.com.

This press release contains forward-looking statements that are subject to risks and uncertainties that could cause actual results to differ from those indicated in the forward-looking statements. Such risks and uncertainties include, but are not limited to the following: our revenue and backlog are dependent on actual future energy purchases pursuant to completed procurements; the demand for our services is affected by changes in regulated prices or cyclicality or volatility in competitive market prices for energy; the effect of acquisitions on revenue, growth, EBITDA and backlog; and there are factors outside our control that affect transaction volume in the electricity market. Additional risk factors are identified in our Annual Report on Form 10-K and subsequent reports filed with the Securities and Exchange Commission.

For additional information, contact:

Dan Mees World Energy Solutions, Inc. (508) 459-8156 <u>dmees@worldenergy.com</u> Susan Forman
The Investor Relations Group
(212) 825-3210
sforman@dgicomm.com

Wellublicigy

Exhibit C-6 "Credit Rating"

World Energy Solutions, Inc. is not rated by any of the large rating agencies other than Dun & Bradstreet. The Dun and Bradstreet report on World Energy Solutions, Inc. is attached in Exhibit C-7.

World Energy Solutions, Inc.'s D&B DUNS Number is: 01-114-1343

Werlutinersy

Exhibit C-7 "Credit Report"

See the enclosed Dun & Bradstreet report for World Energy Solutions, Inc.



Live Report: WORLD ENERGY SOLUTIONS, INC.

D-U-N-S® Number: 01-114-1343

Trade Names: WORLD ENERGY SOLUTIONS

Endorsement/Billing Reference: athomas@worldenergy.com

Address 100 Front St FI 20 Location Type Headquarters Worcester,MA - 01608

Phone 508 459-8100 Fax 508-459-8101

Web www.worldenergy.com

Added to Portfolio: 03/20/2007

Last View Date: 02/05/2014

Endorsement: athomas@worldenergy.c

Company Summary

Currency: Shown in USD unless otherwise indicated

Score Bar

PAYDEX®	0	71	Paying 14 days past due		
Commercial Credit Score Class	Ø.	2	Low to Moderate Risk of severe payment delinquency.		
Financial Stress Score Class		3	Moderate Risk of severe financial stress.		
Credit Limit - D&B Conservative		80,000.00	Based on profiles of other similar companies.		
D&B Rating		Unavailable			

Commercial Credit Score Class
Commercial Credit Score Class: 2
D&B PAYDEX®
D&B PAYDEX® 71 🛇
Lowest Risk:100;Highest Risk:1
When weighted by amount, Payments to suppliers average 14 days beyond terms
Financial Stress Score Class
Financial Stress Score Class: 3 😍

Latest Financial Statement	12-31-2012 / Annual Statement
otal Current Assets	12,699,144
Total Current Liabilities	15,163,862
rotal Assets	55,130,662
otal Liabilities	28,420,535
Current Ratio	0.84
Net worth	26,710,127
Annual Sales	31,778,837
BITDA	-1,760,895
ree Cash Flow	4,190,454
nterest Coverage	-3
Fotal Liabilities to Net Worth Ratio (%)	106.4
Sales to Working Capital Ratio	-12.89

Financial Overview

Lowest Risk:1; Highest Risk:5

Branches (Domestic)

Company	City , State	D-U-N-S® NUMBER
WORLD ENERGY SOLUTIONS, INC.	DUBLIN , Ohio	80-285-8675
WORLD ENERGY SOLUTIONS, INC.	FORT WORTH , Texas	07-829-5195
WORLD ENERGY SOLUTIONS, INC.	DALLAS , Texas	07-829-5200
WORLD ENERGY SOLUTIONS, INC.	HOUSTON, Texas	07-829-5184
WORLD ENERGY SOLUTIONS, INC.	WORCESTER, Massachusetts	07-860-5838

Predictive Scores

Currency: Shown in USD unless otherwise indicated



D&B Viability Rating Summary

The D&B Viability Rating uses D&B's proprietary analytics to compare the most predictive business risk indicators and deliver a highly reliable assessment of the probability that a company will go out of business, become domant/inactive, or file for bankruptcy within the next 12 months.



Viability Score

Lowest Risk:1

Highest Risk:9

Compared to All US Businesses within the D&B Database:

- · Level of Risk: Low Risk
- Businesses ranked 2 have a probability of becoming no longer viable: 2 %
- · Percentage of businesses ranked 2: 4 %
- . Across all US businesses, the average probability of becoming no longer viable: 14 %

6

Portfolio Comparison

Lowest Risk:1

Highest Risk:9

Compared to All US Businesses within the same MODEL SEGMENT:

- Model Segment : Available Financial Data
- · Level of Risk: Moderate Risk
- Businesses ranked 6 within this model segment have a probability of becoming no longer viable: 0.7 %
- Percentage of businesses ranked 6 with this model segment: 8 %
- Within this model segment, the average probability of becoming no longer viable: 0.6 %



Data Depth Indicator

Predictive Data: A Descriptive Data: G

Data Depth Indicator:

- ✓ Rich Firmographics
- ✓ Extensive Commercial Trading Activity
- ✓ Comprehensive Financial Attributes

i	T. 1500000 10 10 25 10 10	ï	
		H	
		٠	
		ı	
	* /AA		ì
	/ = .		,
	₹. ;=;	ľ	
		ı	
		۱	
٠			

Company Profile

Financial Data	Trade Payments	Company Size	Years in Business		
Available	Available (3+ Trade)	Large	Established		

Company Profile Details:

- · Financial Data: Available
- Trade Payments: Available (3+ Trade)
- Company Size: Large (Employees: 50+ or Sales: \$500K+)
- Years in Business: Established (5+)

Credit Capacity Summary

This credit rating was assigned because of D&B's assessment of the company's creditworthiness. For more information, see the D&B Rating Key

D&B Rating: -

The Rating was changed on April 2, 2013 because of changes in financial information, payment information, or other information about this business.

The blank rating symbol should not be interpreted as indicating that credit should be denied. It simply means that the information available to D&B does not permit us to classify the company within our rating key and that further enquiry should be made before reaching a decision. Some reasons for using a "-" symbol include: deficit net worth, bankruptcy proceedings, insufficient payment information, or incomplete history information.

Number of Employees Total:

Below is an overview of the companys rating

history since 04-19-2005

Sales:

31,778,837.00 (Up by 54.8% from last year)

126 (Undetermined here)

D&B Rating	Date Applied
_	04-02-2013
4A2	08-07-2012
4A3	03-13-2012
4A2	04-11-2011
4A3	02-22-2011
4A4	04-03-2009
1R2	11-13-2007

26,710,127

Worth: (Un h

(Up by 29.5% (As of 30-Sep-13)

12-05-2006

11-15**-**2006 04-19**-**2005

from last year)

Working Capital:

1R3

1R2

1R3

(\$2,464,718)

(As of 31-Dec-12)

Payment Activity:

(based on 46 experiences)

16,330

Highest Credit:

200,000

Total Highest Credit:

573,750

D&B Credit Limit Recommendation

Conservative credit Limit

80.000

Aggressive credit Limit:

200,000

Risk category for this business:

LOW

The Credit Limit Recommendation (CLR) is intended to serve as a directional benchmark for all businesses within the same line of business or industry, and is not calculated based on any individual business. Thus, the CLR is intended to help guide the credit limit decision, and must be balanced in combination with other elements which reflect the individual company's size, financial strength, payment history, and credit worthiness, all of which can be derived from D&B reports.

Risk is assessed using D&Bs scoring methodology and is one factor used to create the recommended limits. See Help for details.

Financial Stress Class Summary

The Financial Stress Score predicts the likelihood of a firm ceasing business without paying all creditors in full, or reorganization or obtaining relief from creditors under state/federal law over the next 12 months. Scores were calculated using a statistically valid model derived from D&Bs extensive data files.

The Financial Stress Class of 3 for this company shows that firms with this class had a failure rate of 0.24% (24 per 10,000), which is lower than the average of businesses in D & B's database

Financial Stress Class: 3 9 (Lowest Risk:1; Highest Risk:5)

Moderately lower than average risk of severe financial stress, such as a bankruptcy or going out of business with unpaid debt, over the next 12 months.

Probability of Failure:

Risk of Severe Financial Stress for Businesses with this Class: 0.24 % (24 per 10,000)

Financial Stress National Percentile: 44 (Highest Risk: 1; Lowest Risk: 100)

Financial Stress Score: 1470 (Highest Risk: 1,001; Lowest Risk: 1,875)

Average Risk of Severe Financial Stress for Businesses in D&B database: 0.48 % (48 per 10,000)

The Financial Stress Class of this business is based on the following factors:

UCC Filings reported.

Negative change in net worth.

Low proportion of satisfactory payment experiences to total payment experiences.

High number of inquiries to D & B over last 12 months.

High proportion of past due balances to total amount owing.

Notes:

The Financial Stress Class indicates that this firm shares some of the same business and financial characteristics of other companies with this classification. It does not mean the firm will necessarily experience financial stress.

The Probability of Failure shows the percentage of firms in a given Class that discontinued operations over the past year with loss to creditors. The Probability of Failure - National Average represents the national failure rate and is provided for comparative purposes. The Financial Stress National Percentile reflects the relative ranking of a company among all scorable companies in D&Bs file. The Financial Stress Score offers a more precise measure of the level of risk than the Class and Percentile. It is especially helpful to customers using a scorecard approach to determining overall business performance.

Norms	National %
This Business	44
Region: NORTHEAST	48
Industry: BUSINESS, LEGAL AND ENGINEERING SERVICES	52
Employee range: 100-499	75
Years in Business: 11-25	68

This Business has a Financial Stress Percentile that shows:

Higher risk than other companies in the same region.

Higher risk than other companies in the same industry.

Higher risk than other companies in the same employee size range.

Higher risk than other companies with a comparable number of years in business.

Credit Score Summary

The Commercial Credit Score (CCS) predicts the likelihood of a business paying its bills in a severely delinquent manner (91 days or more past terms), obtaining legal relief from its creditors or ceasing operations without paying all creditors in full over the next 12 months. The Credit Score class of 2 for this company shows that 2.5% of firms with this class paid one or more bills severely delinquent, which is lower than the average of businesses in D & B's database.

Credit Score Class: 2 C Lowest Risk:1; Highest Risk:5

Incidence of Delinquent Payment

Among Companies with this Classification: 2.50 %

Average compared to businesses in D&Bs database: 10.20 % Credit Score Percentile: 75 (Highest Risk: 1; Lowest Risk: 100) Credit Score: 537 (Highest Risk: 101; Lowest Risk:670)

The Credit Score Class of this business is based on the following factors:

Higher risk industry based on delinquency rates for this industry Recent amount past due
Higher risk region based on delinquency rates for this region
Limited business activity signals reported in the past 12 months
Variable Paydex over last 12 months

Notes:

The Commercial Credit Score Risk Class indicates that this firm shares some of the same business and financial characteristics of other companies with this classification. It does not mean the firm will necessarily experience severe delinquency.

The Incidence of Delinquent Payment is the percentage of companies with this classification that were reported 91 days past due or more by creditors. The calculation of this value is based on D&B's trade payment database.

The Commercial Credit Score percentile reflects the relative ranking of a firm among all scorable companies in D&B's file.

The Commercial Credit Score offers a more precise measure of the level of risk than the Risk Class and Percentile. It is especially helpful to customers using a scorecard approach to determining overall business performance.

Norms	National %
This Business	75
Region: NORTHEAST	62
Industry: BUSINESS, LEGAL AND ENGINEERING SERVICES	43

Employee range: 100-499 89 Years in Business: 11-25 66

This business has a Credit Score Percentile that shows:

Lower risk than other companies in the same region.

Lower risk than other companies in the same industry.

Higher risk than other companies in the same employee size range.

Lower risk than other companies with a comparable number of years in business.

Trade Payments

Currency: Shown in USD unless otherwise indicated

D&B PAYDEX®

The D&B PAYDEX is a unique, weighted indicator of payment performance based on payment experiences as reported to D&B by trader references. Learn more about the D&B PAYDEX

Timeliness of historical payments for this company.

Current PAYDEX is

71 Equal to 14 days beyond terms (Pays more slowly than the average for its industry of 3 days beyond terms)

Industry Median is

78 Equal to 3 days beyond terms

Payment Trend currently is 🛑 Unchanged, compared to payments three months ago

Indications of slowness can be the result of dispute over merchandise, skipped invoices etc. Accounts are sometimes placed for collection even though the existence or amount of the debt is disputed.

Total payment Experiences in D&Bs File (HQ)	46
Payments Within Terms (not weighted)	88 %
Trade Experiences with Slow or Negative Payments(%)	17.39%
Total Placed For Collection	0
High Credit Average	16,330
Largest High Credit	200,000
Highest Now Owing	100,000
Highest Past Due	40,000

D&B PAYDEX®: 71 (Lowest Risk:100; Highest Risk:1)

When weighted by amount, payments to suppliers average 14 days beyond terms

3-Month D&B PAYDEX®: 70 O (Lowest Risk:100; Highest Risk:1)

Based on payments collected over last 3 months.

When weighted by amount, payments to suppliers average 15 days beyond terms

D&B PAYDEX® Comparison

Current Year

PAYDEX® of this Business compared to the Primary Industry from each of the last four quarters. The Primary Industry is Business services, based on SIC code 7389.

Shows the trend in D&B PAYDEX scoring over the past 12 months.

4/13 5/13 6/13 7/13 8/13 9/13 10/13

This Business	77	76	75	69	72	70	69	71	74	73	73	71
Industry Quartiles												
Upper	•		80			80			80			
Median		•	78		•	78			78			
Lower		•	71			71			71			

Current PAYDEX for this Business is 71, or equal to 14 days beyond terms The 12-month high is 77, or equal to 5 DAYS BEYOND terms

The 12-month low is 69, or equal to 16 DAYS BEYOND terms

Previous Year

Shows PAYDEX of this Business compared to the Primary Industry from each of the last four quarters. The Primary Industry is Business services , based on SIC code 7389.

Previous Year	03/12 Q1'12	06/12-1 Q212	.09/12 Q3'12	12/12 Q4'12
This Business	UN	73	76	76
Industry Quartiles				
Upper	80	80	80	80
Median	78	78	78	78
Lower	70	70	70	70

Based on payments collected over the last 4 quarters.

Current PAYDEX for this Business is 71, or equal to 14 days beyond terms. The present industry median Score is 78, or equal to 3 days beyond terms. Industry upper quartile represents the performance of the payers in the 75th percentile. Industry lower quartile represents the performance of the payers in the 25th percentile.

Payment Habits

For all payment experiences within a given amount of credit extended, shows the percent that this Business paid within terms. Provides number of experiences to calculate the percentage, and the total credit value of the credit extended.

\$ Cre	dit Extended	# Payment Experiences	Total Amount		% of Payr	nents WithIn	Terms	
	Over 100,000	1	200,000	100%				
	50,000-100,000	3 -	205,000	39%				
	15,000-49,999	5	125,000	62%				
	5,000-14,999	4	25,000	90%				
	1,000-4,999	9	13,500	78%				
4	Under 1,000	13	3,050	92%				

Based on payments collected over last 24 months.

All Payment experiences reflect how bills are paid in relation to the terms granted. In some instances, payment beyond terms can be the result of disputes over merchandise, skipped invoices etc.

Payment Summary

There are 46 payment experience(s) in D&Bs file for the most recent 24 months, with 26 experience(s) reported during the last three month period.

The highest Now Owes on file is 100,000. The highest Past Due on file is 40,000

Below is an overview of the companys currency-weighted payments, segmented by its suppliers primary industries:

	Total Revd (#)	Total Larg Amts Cred	est High] It	Within Terms (%)	Day <31 (%)	/s Slow 31-60 6 (%)	1-90	90>
Top Industries								
Telephone communictns	6	4,200	2,500	100	0	0	0	0

Public finance	· 4	11,550	10,000	100	0	0	0	0
Misc business credit	. 4	4,500	2,500	89	11	0	0	0
Mfg refrig/heat equip	3	180,000	80,000	44	56	0	0	0
Whol electrical equip	2	200,000	200,000	100	0	0	0	0
Radiotelephone commun	2	35,000	20,000	100	0	0	0	0
Nonclassified	2	10,000	5,000	100	0	0	0	0
Misc business service	2	5,750	5,000	57	43	0	0	0
Whol heating/ac equip	1	55,000	55,000	0	100	0	0	0
Gravure printing	1	35,000	35,000	50	0	50	0	0
Whol computers/softwr	1	25,000	25,000	100	0	. 0	0	0
Computer system desgn	1	2,500	2,500	0	100	0	0	0
Short-trm bush credit	1	1,000	1,000	100	0	0	0	0
Misc equipment rental	1	1,000	1,000	100	0	0	0	0
Data processing svcs	1	500	500	100	0	0	0	0
Trucking non-local	1	250	250	0	100	0	0	0
Whol durable goods	1	100	100	100	0	0	, 0	0
Reg misc coml sector	1	100	100	100	0	0	0	0
Electric services	1	50	50	100	0	0	0	0
Lithographic printing	1	50	50	100	0	0	0	0
Other payment categories								
Cash experiences	9	2,200	1,000					
Payment record unknown	0	0	0		•			
Unfavorable comments	0	0	0					
Placed for collections	0	N/A	0					
Total in D&B's file	46	573,750	200,000					

Accounts are sometimes placed for collection even though the existence or amount of the debt is disputed.

Indications of slowness can be result of dispute over merchandise, skipped invoices etc.

Detailed payment history for this company

Date Reported (mm/yy)	Paying Record	High Credit N	low Owes	Past Due Sellin Terms	Last Sale Within (month)
02/14	Ppt	25,000	25,000	0 N30	1 mo
	Ppt	5,000	0	0	6-12 mos
	Ppt	2,500	1,000	0	1 mo
	Ppt	2,500	2,500	0	1 mo
	Ppt	1,000	250	0	1 mo
,	Ppt	1,000	250	Lease Agreemn	t
	Ppt	1,000	750	0 Lease Agreemn	1 mo t
	Ppt-Slow 30	5,000	1,000	0	1 mo
	Ppt-Slow 30	1,000	100	0	1 mo
	Ppt-Slow 60	35,000	10,000	0	1 mo
	Slow 5	250	0	0	4-5 mos
	Slow 15	2,500	0	0	2-3 mos
	Slow 30	70,000	40,000	40,000	1 mo
	Slow 30	55,000	55,000	30,000	1 mo
01/14	Ppt	200,000	100,000	250	1 mo
	Ppt	20,000	Ò	. 0	2-3 mos

	Ppt	15,000	10,000	0	1 mo
	Ppt	5,000	0	0	2-3 mos
	Ppt	1,000	0	0	4-5 mos
	Ppt	750	500	0	1 mo
	Ppt	500	0	0	4-5 mos
	Ppt	500	0	0	1 mo
	Ppt	100	50	0	4-5 mos
	Ppt	50	0	0	1 mo
	Ppt	50	0	0	4-5 mos
	Ppt	50	0	0	4-5 mos
08/13	Ppt .	10,000			1 mo
07/13	Ppt	000,08	30,000	0 N30	1 mo
	(029)	50		Cash account	1 mo
06/13	Ppt	500	•	•	1 mo
	Slow 30	30.000	0	0 N30	6-12 mos
05/13	Ppt	50		•	1 mo
	Ppt	0	0	0 N30	6-12 mos
	(034)	750		Cash account	1 mo
	(035) Satisfactory.	100			1 mo
i.	(036) Cash own option ,	100		Cash account	1 mo
	(037) Cash own option .	100		Cash account	1 mo
04/13	Ppt	1,000 .	0	0	1 mo
03/13	(039)	1,000		Cash account	1 mo
02/13	Ppt	100	0	0	6-12 mos
11/12	(041)	50		Cash account	1 mo
	(042)	50		Cash account	6-12 mos
08/12	(043)	50		Cash account	1 mo
.05/12	Ppt	50			1 mo
02/12	(045)	50		Cash account	1 mo
01/12	Ppt		. 0	. 0	6-12 mos

Payments Detail Key: # 30 or more days beyond terms

Payment experiences reflect how bills are paid in relation to the terms granted. In some instances payment beyond terms can be the result of disputes over merchandise, skipped invoices, etc. Each experience shown is from a separate supplier. Updated trade experiences replace those previously reported.

Public Filings

Currency: Shown in USD unless otherwise indicated

Summary

The following data includes both open and closed filings found in D&B's database on this company.

Record Type # of Records | Most Recent Filing Date |

Bankruptcy Proceedings | 0 | - |

Judgments | 0 | - |

Liens | 0 | - |

Suits | 0 | - |

UCCs 10 12/31/13

The following Public Filing data is for information purposes only and is not the official record. Certified copies can only be obtained from the official source.

UCC Filings

Collateral All Assets
Type Original

Sec. Party SILICON VALLEY BANK, SANTA CLARA, CA

pebtor WORLD ENERGY SOLUTIONS, INC.

Filing No. 2008 3018593

Filed With SECRETARY OF STATE/UCC DIVISION, DOVER, DE

 Date Filed
 2008-09-08

 Latest Info Received
 10/09/08

Type Amendment

Sec. Party SILICON VALLEY BANK

Debtor WORLD ENERGY SOLUTIONS, INC.

Filing No. 2012 4460012

Filed With SECRETARY OF STATE/UCC DIVISION, DOVER, DE

 Date Filed
 2012-11-19

 Latest Info Received
 12/13/12

 Original Filing No.
 2008 3018593

Type Continuation

sec. Party SILICON VALLEY BANK, SANTA CLARA, CA

Debtor WORLD ENERGY SOLUTIONS, INC.

Filing No. 2013 1920736

Filed With SECRETARY OF STATE/UCC DIVISION, DOVER, DE

 Date Filed
 2013-05-20

 Latest Info Received
 08/09/13

 Original UCC Filed Date
 2008-09-08

 Original Filing No.
 2008 3018593

Type Termination

Sec. Party SILICON VALLEY BANK, SANTA CLARA, CA

Debtor WORLD ENERGY SOLUTIONS, INC.

Filing No. 2013 5179842

Filed With SECRETARY OF STATE/UCC DIVISION, DOVER, DE

Date Filed 2013-12-31

Latest Info Received 02/18/14

Original UCC Filed Date 2008-09-08

Original Filing No. 2008 3018593

Collateral Negotiable instruments including proceeds and products - Accounts receivable including proceeds

and products - Inventory including proceeds and products - Account(s) including proceeds and

products - and OTHERS

Type Original

Sec. Party COMMERCE BANK & TRUST COMPANY, WORCESTER, MA

Debtor WORLD ENERGY SOLUTIONS, INC.

Filing No. 2013 5160867

Filed With SECRETARY OF STATE/UCC DIVISION, DOVER, DE

Date Filed 2013-12-30
Latest Info Received 01/24/14

Collateral Negotiable instruments and proceeds - Inventory and proceeds - Account(s) and proceeds - Assets

and proceeds - and OTHERS

Type Original

Sec. Party COMMERCE BANK & TRUST COMPANY, WORCESTER, MA

Debtor WORLD ENERGY SECURITIES CORP.

Filing No. 201309018130

Filed With SECRETARY OF STATE/UCC DIVISION, BOSTON, MA

 Date Filed
 2013-12-31

 Latest Info Received
 01/07/14

Collateral Computer equipment including proceeds and products - PATENTS/TRADEMARKS including

proceeds and products

Type Original

Sec. Party COMMERCE BANK & TRUST COMPANY, WORCESTER, MA

Debtor WORLD ENERGY SOLUTIONS, INC.

Filing No. 2013 5160982

Filed With SECRETARY OF STATE/UCC DIVISION, DOVER, DE

 Date Filed
 2013-12-30

 Latest Info Received
 01/24/14

Type Continuation

Sec, Party SILICON VALLEY BANK, SANTA CLARA, CA

Debtor WORLD ENERGY SECURITIES CORP.

Filing No. 201304087230

Filed With SECRETARY OF STATE/UCC DIVISION, BOSTON, MA

 Date Filed
 2013-05-20

 Latest Info Received
 05/27/13

 Original UCC Filed Date
 2008-09-08

 Original Filing No.
 200868174350

Type Termination

Sec. Party SILICON VALLEY BANK, SANTA CLARA, CA

Debtor WORLD ENERGY SECURITIES CORP.

Filing No. 201309045550

Filed With SECRETARY OF STATE/UCC DIVISION, BOSTON, MA

Date Filed 2013-12-31 Latest Info Received 01/07/14 Original UCC Filed Date 2008-09-08

Original Filing No. 200868174350

Type Original

Sec. Party CIT COMMUNICATIONS FINANCE CORPORATION, LIVINGSTON, NJ

Debtor WORLD ENERGY SOLUTIONS, INC.

Filing No. 200971734890

Filed With SECRETARY OF STATE/UCC DIVISION, BOSTON, MA

Date Filed 2009-03-10 Latest Info Received 03/18/09

Government Activity

Activity summary

Borrower (Dir/Guar) NO Administrative Debt NO Contractor YES NO Grantee Party excluded from federal program(s) NO

Possible candidate for socio-economic program consideration

Labour Surplus Area N/A Small Business N/A N/A 8(A) firm

The details provided in the Government Activity section are as reported to Dun & Bradstreet by the federal government and other sources.

Special Events

Currency: Shown in USD unless otherwise indicated

Special Events

11/11/2013 - EARNINGS UPDATE:

According to published reports, comparative operating results for the 9 months ended September 30, 2013: Revenue of \$25,331,696, Net Income of (\$3,203,338); compared to Revenue of \$21,584,460, Net Income of (\$2,245,241) for the comparable period in the prior year.

08/09/2013 -EARNINGS UPDATE:

According to published reports, comparative operating results for the 6 months ended June 30, 2013: Revenue of \$16,593,139, Net Income of (\$2,611,040); compared to Revenue of \$13,978,606, Net Income of (\$1,700,086) for the comparable period in the prior year.

ĺ	S.	

Company (Overview
-----------	----------

Company Name: WORLD ENERGY SOLUTIONS, INC.

Doing Business As: WORLD ENERGY SOLUTIONS

Street Address: 100 Front St FI 20 Worcester, MA 01608

Phone: 508 459-8100 Fax: 508-459-8101

URL: http://www.worldenergy.com

Stock Symbol: **XWES** History Is clear Operations Profitable Present management control 18 years **Annual Sales** 31,778,837

History

The following information was reported: 12/17/2013

PHILIP ADAMS, PRES-CEO+

JAMES PARSLOW, CFO-TREAS-SEC Officer(s):

MARTHA DANLY, COO EDWARD LIBBEY, CHB+

DIRECTOR(S):

The officers identified by (+) and John Wellard and Thad Wolfe.

The company was incorporated in the State of Delaware on June 22, 1999.

Business started 1996.

The company commenced operations through an entity named Oceanside Energy, Inc., or Oceanside, which was incorporated under the laws of the State of Delaware on September 3, 1996. It was incorporated World Energy Solutions, Inc. under the laws of the State of Delaware under the name World Energy Exchange, Inc. on June 22, 1999 and on October 31, 1999, Oceanside became a wholly-owned subsidiary of World Energy Solutions, Inc. and was subsequently dissolved. On December 21, 2006, the company incorporated a 100% owned subsidiary, World Energy Securities Corp., under the laws of the Commonwealth of Massachusetts.

.....

The company's common stock is traded on the NASDAQ Capital Market under the symbol "XWES". As of April 9, 2013, there were 74 shareholders of record. As of April 2, 2013, those shareholders identified by the company as beneficially owning 5% or more of the outstanding shares were:

Richard Domaleski/ Roman Holdings Trust/ RD Holdings (13.9%); Royce & Associates (9.7%); GSE Consulting, L.P. reporting group (8.3%); Ardsley Advisory Partners (6.6%) and Austin W Marxe and David M Greenhouse (7.7%). As of the same date, officers and directors as a group beneficially owned 5.7% of the outstanding shares.

RECENTS EVENTS:

On October 3, 2012, the company acquired substantially all of the assets of and certain obligations of Northeast Energy Partners, LLC, Enfield, CT.

On November 1, 2011, the company purchased GSE Consulting LP, Houston, TX. GSE Consulting LP's offices in Dallas, Fort Worth and Houston will operate as branches of World Energy Solutions Inc.

On October 17, 2011, the company acquired all of the assets of Northeast Energy Solutions, LLC, Cromwell, CT.

On September 14, 2011, the company purchased the energy procurement business of Co-Exprise, Inc., Wexford, PA.

PHILIP V ADAMS. Director since 2012. He was appointed CEO of the company on June 6, 2012. He has served as its President since October 2007 and held the position of COO from October 2003 to June 2012.

JAMES PARSLOW. He joined the company in May 2006 and serves as the company's CFO, Treasurer and Secretary, He is a CPA in Massachusetts with over 20 years experience serving private and public companies in the alternative energy, biomedical products and services, online auction services and high technology manufacturing industries. Since April 2004 until joining the company in 2006, he was the CFO and Treasurer for Spire Corporation. Prior to joining Spire, he was an independent financial consultant from January 2003 to March 2004. MARTHA DANLY. She was appointed as the COO of the company on April 17, 2013. She has been with the company since September 2012 and most recently served as its Chief Product Officer. She has a B.A. in linguistics from Macalester College, where she graduated summa cum laude and a Ph.D. in experimental psychology from Harvard University.

EDWARD LIBBEY. Director since 1999. He is the company's CHB of Directors and has served in this position since 2007. He holds a First Class Degree and Doctorate in Chemistry from Cambridge and is a Fellow of the Energy Institute.

JOHN WELLARD. Director since 2006. From March 1996 to April 2005, he was employed with Union Gas Limited in various capacities, including as its President from May 2003 to December 2004.

THAD WOLFE, Director since 2007, Since September 2008, he has been the Owner of Polaris Results LLC and an independent contractor working with QinetiQ-North America, Thomas Group, Inc.

Business Registration

CORPORATE AND BUSINESS REGISTRATIONS REPORTED BY THE SECRETARY OF STATE OR OTHER OFFICIAL SOURCE AS OF Jun 01 2013

Registered Name:

WORLD ENERGY SOLUTIONS, INC.

Business type:

CORPORATION

Corporation type:

NOT AVAILABLE

Date incorporated:

Jun 22 1999

State of incorporation:

DELAWARE

Filing date:

Jun 22 1999

Registration ID:

3059369

Status:

STATUS NOT AVAILABLE

Where filed:

SECRETARY OF STATE/CORPORATIONS DIVISION, DOVER, DE

Registered agent:

NATIONAL REGISTERED AGENTS, INC., 160 GREENTREE DR STE

101, DOVER, DE, 199040000

Operations

12/17/2013

The company provides a range of energy management solutions. The company operates in two segments, Energy Procurement and Energy Efficiency Services. The Energy Procurement segment delivers energy procurement services. The Energy Efficiency Services segment provides turn-key electrical, mechanical and lighting energy efficiency services. The company operates World Energy Exchange, an online auction platform that enables energy consumers to negotiate for the purchase or sale of electricity, natural gas and other energy resources from competing energy suppliers who have agreed to participate on its auction platform in a given event; and enables buyers and

Description:

sellers to negotiate for the purchase or sale of environmental commodities, such as renewable energy certificates,

verified emissions reductions and certified emissions reductions.

Terms are on contract basis. Sells to commercial concerns. Territory: International.

Season peaks weather conditions.

Employees:

126 which includes officer(s). Undetermined employed here.

Facilities:

Leases 12,400 sq. ft. in a building.

Branches:

This business has multiple branches, detailed branch information is available in D & B's linkage or family tree products.

Subsidiaries:

This business has one subsidiary, detailed subsidiary information is available in D & B's linkage or family tree products.

SIC & NAICS

SIC:

Based on information in our file, D&B has assigned this company an extended 8-digit SIC. D&B's use of 8-digit SICs enables us to be more specific about a company's operations than if we use the standard 4-digit code.

The 4-digit SIC numbers link to the description on the Occupational Safety & Health Administration (OSHA) Web site. Links open in a new browser window.

7389 1400 Auction, appraisal, and exchange services NAICS:

561499 All Other Business Support Services

Financials

Currency: Shown in USD unless otherwise indicated

Company Financials: D&B

05/08/2013

Three-year Statement Comparative:

Fiscal Consolidated

Fiscal Consolidated

Fiscal Consolidated Dec 31 2010

Current Liabilities 15,163,862 27,739,082 3,5	13,577
Current Ratio 0.84 0.24	1.97
Working Capital (2,464,718) (21,163,605) 3,3	99,147
Other Assets 42,431,518 27,739,082 7,8	14,070
Net Worth 26,710,127 20,619,101 11,2	12,012
Sales 31,778,837 20,524,567 17,9	84,662
Long Term Liab 13,256,673 3,123,291	1,205
Net Profit (Loss) 5,290,692 (46,477) (5	99,087)

Company Financial: EDGAR (Annual Statements)

Balance Sheet

Period Ending	Fiscal Consolidated Dec 31 2012	Fiscal Consolidated Dec 31 2011	Fiscal Consolidated Dec 31 2010
ASSETS			
Current Assets			
Cash and Cash Equivalents	3,307,822	1,837,801	3,559,288
Short Term Investments	UN	UN	UN
Net Trade Receivables	8,874,883	3,603,634	3,124,328
Inventory	154,626	288,174	0
Other Current Assets	361,813	392,287	229,108
Total Current Assets	12,699,144	6,121,896	6,912,724
Property, Plant and Equipment	639,839	426,403	287,191
Fixed Assets	639,839	426,403	287,191
Long Term Investments	0	716,936	433,333
Deferred Long Term Asset Charges	5,844,980	0	UN
Other Assets	685,867	109,516	191,238
Goodwill	16,167,834	12,307,255	3,178,701
Total Assets	55,130,662	33,860,978	14,726,794
Accumulated Amortization	UN	UN	UN
Intangible Assets	19,092,998	14,178,972	3,723,607
LIABILITIES			
Current Liabilities			
Accounts Payable	10,221,556	8,755,638	2,421,482
Short Term And Current Long Term Debt	3,012,929	970,185	862,556
Other Current Liabilities	1,929,377	392,763	229,539
Total Current Liabilities	15,163,862	10,118,586	3,513,577
Long Term Debt	8,410,286	0	1,205
Deferred Long Term Liability Charges	3,379,635	526,592	0
Negative Goodwill	UN	UN	UN
Minority Interest Expense	UN	UN	UN
Other Liabilities	1,466,752	2,596,699	0
Misc Stocks Options Warrant	UN	UN	UN
Total Liabilities	28,420,535	13,241,877	3,514,782
SHAREHOLDER'S EQUITY			
Preferred Stock Amount	0	0	0
Common Stock	1,195	1,185	916
Retained Earnings	-16,836,823	-22,127,515	-22,081,038

Treasury Stock	224,353	221,603	209,940
Capital Surplus	43,770,108	42,967,034	33,502,074
Other Equity	UN	UN	UN
Total Equity	26,710,127	20,619,101	11,212,012

Income Statement

Period Ending	Fiscal Consolidated Dec 31 2012	Fiscal Consolidated Dec 31 2011	Fiscal Consolidated Dec 31 2010
Sales (Revenue)	31,778,837	20,524,567	17,984,662
Cost of Goods Sold	10,069,357	4,009,995	3,715,869
Gross Profit	21,709,480	16,514,572	14,268,793
Sales And General Admin	23,410,612	16,421,299	14,376,562
Research and Development Expense	UN	UN	UN
Non-Recurring Charges	UN	UN	UN
Other Operating Items	UN	UN	UN
Operating Income	-1,701,132	93,273	-107,769
Net Total Other Income and Expenses	59,763	51,245	11,042
Earnings Before Interest and Taxes	-1,641,369	144,518	-96,727
Interest Expense	547,075	52,771	2,360
Earning Before Tax	-2,188,444	91,747	-99,087
Income Tax Expense	-7,479,136	138,224	0
Equity Earnings or Loss	UN	UN	UN
Minority Interest Expense	UN	UN	UN
Net Income From Continuing Operations	5,290,692	-46,477	-99,087
Discontinued Operations	UN	UN	UN
Effect of Accounting Changes	ŲN	UN	UN
Extraordinary Items	UN	UN	UN
Other Non-Operating Expenses	0	0	0
Net Income	5,290,692	-46,477	-99,087
Preferred Stocks and Other Adjustments	UN	UN	UN
Net Income Applicable to Common Shares	5,290,692	-46,477	-99,087

Statement of Cash Flow

Period Ending	Fiscal Consolidated Dec 31 2012	Fiscal Consolidated Dec 31 2011	Fiscal Consolidated Dec 31 2010
Depreciation	3,291,589	1,621,034	1,434,103
Net Income Adjustments	-7,051,989	644,027	657,913
Changes in Liabilities	5,396,069	1,936,619	-105,026
Changes in Accounts Receivables	-2,685,195	-66,698	-215,304
Changes in Inventories	133,548	-286,299	-0
Changes in Other Operating Activities	-588,166	-154,479	-16,075
Net Cash Flows - Operating Activities	3,786,548	3,647,727	1,656,524
Capital Expenditures	-403,906	-17,540	-55,770
Investments	770,042	-216,666	-433,334
Other Cash Flows from Investing Activities	-8,110,959	-10,447,995	-20,500
Net Cash Flows - Investing Activities	-7,744,823	-10,682,201	-509,604
Dividends Paid	UN	UN	UN
Sale and Purchase of Stock	194,9 44	5,381,624	381,634
Net Borrowings	7,483,352	-68,637	-16,175

Other Cash Flows from Financing Activities	-2,250,000	0	0
Net Cash Flows - Financing Activities	5,428,296	5,312,987	365,459
Effect of Exchange Rate	UN	UN	UN
Change in Cash and Cash Equivalents	1.470.021	-1.721.487	1.512.379

Financial Ratios

Period Ending	Fiscal Consolidated Dec 31 2012	Fiscal Consolidated Dec 31 2011	Fiscal Consolidated Dec 31 2010
Solvency Ratios			
Current Ratio	0.84	0.61	1.97
Quick Ratio	UN	UN	UN
Current Liabilities to Net Worth (%)	56.77	49.07	31.34
Total Liabilities to Net worth (%)	106.40	64.22	31.35
Current Liabilities to Inventory (%)	9806.80	3511.28	UN
Fixed Assets to Net Worth (%)	2,40	2.07	2.56
Cash Ratio	UN	UN	UN
Efficiency Ratios			
Accounts Payable to Sales Ratio	0.32	0.43	0.13
Sales to Working Capital Ratio	-12.89	-5.14	5.29
Sales to Inventory (%)	20552.07	7122.28	UN
Assets to Sales (%)	173.48	164.98	81.89
ROCE (%)	-4.11	0.61	-0.86
Profitability Ratios			
Return on Net Worth (%)	19.81	-0.23	-0.88
Return on Assets (%)	9.60	-0.14	-0.67
Return on Sales (%)	-5.16	0.70	-0.54
Gross Profit Margin (%)	68.31	80.46	79.34
Operating Margin (%)	-5.35	0.45	-0.60
Pre-Tax Profit Margin (%)	-6.89	0.45	-0.55
Profit Margin (%)	16.65	-0.23	-0.55
Pre-Tax Return on Equity (%)	-8.19	0.44	-0.88
After Tax Return on Equity (%)	19.81	-0.23	-0.88
Operating Income to Interest Ratio	-3.11	1.77	-45.66
Financial Leverage Ratios			
EBITDA to EBIT Ratio	3.01	-10.22	15.83
Debt to Income Ratio	5.37	-284.91	~35.47
Debt to Equity Ratio	1.06	0.64	0.31
Equity Ratio (%)	48.45	60.89	76.13
Interest Coverage Ratio	-3.00	2.74	-40.99
Interest Coverage by EBITDA Ratio	-9.02	-27.98	-648.66

Note:UN=Unavailable

Powered By EDGAROnline

This Data is Provided as per License from EDGAR online

Company Financial: EDGAR (Quarterly Statements)

Balance Sheet

Quarter Q3 Q2 Q1 Q4

Quarter Ending	Sep 30 2013	Jun 30 2013	Mar 31 2013	Dec 31 2012
ASSETS				
Current Assets				
Cash and Cash Equivalents	2,518,377	2,191,478	2,103,246	3,307,822
Short Term Investments	UN	UN	UN	UN
Net Trade Receivables	8,641,291	8,571,882	8,845,637	8,874,883
Inventory	501,406	454,984	432,000	154,626
Other Current Assets	439,721	682,800	432,762	361,813
Total Current Assets	12,100,795	11,901,144	11,813,645	12,699,144
Property, Plant and Equipment	603,183	604,828	604,425	639,839
Fixed Assets	603,183	604,828	604,425	639,839
Long Term Investments	UN	UN	UN	0
Deferred Long Term Asset Charges	5,617,565	5,690,370	5,769,175	5,844,980
Other Assets	626,269	596,155	664,285	685,867
Goodwill	16,167,834	16,167,834	16,167,834	16,167,834
Total Assets	51,284,369	52,103,813	53,137,604	55,130,662
Accumulated Amortization	UN	UN	UN	UN
Intangible Assets	16,168,723	17,143,482	18,118,240	19,092,998
LIABILITIES				
Current Liabilities				
Accounts Payable	7,449,733	7,803,403	9,108,456	10,221,556
Short Term And Current Long Term Debt	5,452,690	5,403,259	3,314,671	3,012,929
Other Current Liabilities	3,378,824	2,421,488	2,167,799	1,929,377
Total Current Liabilities	16,281,247	15,628,150	14,590,926	15,163,862
Long Term Debt	6,940,191	7,430,223	7,920,254	8,410,286
Deferred Long Term Liability Charges	4,026,184	4,564,103	4,142,703	3,379,635
Negative Goodwill	UN	UN	UN	UN
Minority Interest Expense	UN	UN	UN	UN
Other Liabilities	15,747	16,702	517,634	1,466,752
Misc Stocks Options Warrant	UN	UN	UN	UN
Total Liabilities	27,263,369	27,639,178	27,171,517	28,420,535
SHAREHOLDER'S EQUITY				
Preferred Stock Amount	0	0	0	0
Common Stock	1,201	1,201	1,197	1,195
Retained Earnings	-20,040,161	-19,447,863	-17,793,708	-16,836,823
Treasury Stock	259,191	258,928	224,394	224,353
Capital Surplus	44,319,151	44,170,225	43,982,992	43,770,108
Other Equity	UN	UN	UN	UN
Total Equity	24,021,000	24,464,635	25,966,087	26,710,127
Income Statement				
Quarter	Q3	Q2	Q1	Q4
Quarter Ending	Sep 30 2013	Jun 30 2013	Mar 31 2013	Dec 31 2012
Sales (Revenue)	8,738,557	7,935,657	8,657,482	10,194,377
Cost of Goods Sold	2,243,875	2,144,455	2,233,151	3,454,546
Gross Profit	6,494,682	5,791,202	6,424,331	6,739,831
Sales And General Admin	6,693,981	7,057,368	7,039,754	6,489,740
Research and Development Expense	UN	UN	UN	UN
Non-Recurring Charges	UN	UN	UN	UN
Other Operating Items	UN	UN	UN	UN

Operating Income	-199,299	-1,266,166	-615,423	250,091
Net Total Other Income and Expenses	3,378	20,713	-7,420	6,657
Earnings Before Interest and Taxes	-195,921	-1,245,453	-622,843	256,748
Interest Expense	253,822	277,397	202,737	272,451
Earning Before Tax	-449,743	-1,522,850	-825,580	-15,703
Income Tax Expense	142,555	131,305	131,305	-7,551,636
Equity Earnings or Loss	UN	UN	UN	UN
Minority Interest Expense	UN	UN	UN	UN
Net Income From Continuing Operations	-592,298	-1,654,155	-956,885	7,535,933
Discontinued Operations	UN	UN	UN	UN
Effect of Accounting Changes	UN	UN	UN	UN
Extraordinary Items	UN	UN	UN	UN
Other Non-Operating Expenses	0	0	0	0
Net Income	-592,298	-1,654,155	-956,885	7,535,933
Preferred Stocks and Other Adjustments	UN	UN	UN	UN
Net Income Applicable to Common Shares	-592,298	-1,654,155	-956,885	7,535,933
Statement of Cash Flow				
Quarter	Q3	Q2	Q1	Q4
Quarter Ending	Sep 30 2013	Jun 30 2013	Mar 31 2013	Dec 31 2012
Depreciation	1,039,639	1,038,334	1,038,934	1,026,203
Net Income Adjustments	261,986	278,870	284,999	-7,393,572
Changes in Liabilities	115,561	950,946	650,426	1,220,800
Changes in Accounts Receivables	-102,409	240,755	-3,754	-313,001
Changes in Inventories	-46,422	-22,984	-277,374	587,574
Changes in Other Operating Activities	243,079	-250,038	-70,949	-243,098
Net Cash Flows - Operating Activities	919,136	581,728	665,397	2,420,839
Capital Expenditures	-54,728	-65,023	-9,216	-57,483
Investments	-0	-0	-0	-0
Other Cash Flows from Investing Activities	-38,621	68,374	13,075	-8,110,959
Net Cash Flows - Investing Activities	-93,349	3,351	3,859	-8,168,442
Dividends Paid	UN	UN	UN	UN
Sale and Purchase of Stock	5,738	6,380	66,859	44,826
Net Borrowings	-504,626	-503,227	-505,143	5,996,509
Other Cash Flows from Financing Activities	0	0	-1,435,548	0
Net Cash Flows - Financing Activities	-498,888	-496,847	-1,873,832	6,041,335
Effect of Exchange Rate	UN	UN	UN	UN
Change in Cash and Cash Equivalents	326,899	88,232	-1,204,576	293,732
Financial Ratios				
Quarter	Q3	Q2	Q1	Q4
Quarter Quarter Ending	Sep 30 2013	Jun 30 2013	Mar 31 2013	Dec 31 2012
Solvency Ratios	00p 40 20 13	Juli 60 20 10	01 2010	
Current Ratio	0.74	0.76	0.81	0.84
Quick Ratio	UN	UN	UN	UN
Current Liabilities to Net Worth (%)	67.78	63.88	56.19	56.77
	113.50	112,98	104.64	106.40
Total Liabilities to Net worth (%)	3247.12	3434.88	3377.53	9806.80
Current Liabilities to Inventory (%) Fixed Assets to Net Worth (%)	2.51	2,47	2,33	2,40
Liven vesers in their Motrii (-1/4)	2.31	A-10*T7	2,00	_,

Cash Ratio	UN	UN	UN	UN
Efficiency Ratios				
Accounts Payable to Sales Ratio	0.85	0.98	1.05	1.00
Sales to Working Capital Ratio	-2.09	-2.13	-3.12	-4.14
Sales to Inventory (%)	1742.81	1744.16	2004.05	6592.93
Assets to Sales (%)	586.87	656.58	613.78	540.79
ROCE (%)	-0.56	-3.41	-1.62	0.64
Profitability Ratios				
Return on Net Worth (%)	-2.47	-6.76	-3,69	28.21
Return on Assets (%)	-1.15	-3.17	-1.80	13.67
Return on Sales (%)	-2.24	-15.69	-7.19	2.52
Gross Profit Margin (%)	74.32	72.98	74.21	66.11
Operating Margin (%)	-2.28	-15.96	-7.11	2.45
Pre-Tax Profit Margin (%)	-5.15	-19.19	-9.54	-0.15
Profit Margin (%)	-6.78	-20.84	-11.05	73.92
Pre-Tax Return on Equity (%)	-1.87	-6.22	-3.18	-0.06
After Tax Return on Equity (%)	-2.47	-6.76	-3.69	28,21
Operating Income to Interest Ratio	-0.79	-4.56	-3.04	0.92
Financial Leverage Ratios				
EBITDA to EBIT Ratio	6.31	1.83	2.67	-3.00
Debt to Income Ratio	-46.03	-16.71	-28.40	3.77
Debt to Equity Ratio	1.13	1.13	1.05	1.06
Equity Ratio (%)	46.84	46.95	48.87	48.45
Interest Coverage Ratio	-0.77	-4.49	-3.07	0.94
Interest Coverage by EBITDA Ratio	-4.87	-8.23	-8.20	-2.82

Note:UN=Unavailable

Powered By EDGAROnline

This Data is Provided as per License from EDGAR online

Stock Performance

· 🗓

11/12/2013

Interim Consolidated statement dated SEP 30 2013:

Assets		Liabilities	
Cash	2,518,377	Accts Pay	1,621,439
Accts Rec	7,108,011	Accrued Exps & Other Current Liabs	707,827
Inventory	501,406	Accruals	6,613,030
Deferred Tax Asset	1,533,280	L.T. Liab-(1yr)	1,960,127
Prepaid Exps & Other Current Asset	ts 439,721	Deferred Revenue/Customer Advances	3,378,824
		Rel Party Subordinated Notes Pay	2,000,000
Curr Assets	12,100,795	Curr Liabs	16,281,247
Fixt & Equip	603,183	Long-Term Debt-Net	2,940,191
Intangibles-Net	16,168,723	Subordinated Notes Payable	4,000,000
Goodwill	16,167,834	L.T. Liab-Other	4,041,931
Deferred Tax Assets-Net	5,617,565	COMMON STOCK	1,201
Other Assets	626,269	ADDIT. PDIN CAP	44,319,151
		TREASURY STOCK	(259,191)
		RETAINED EARNINGS	(20,040,161)
Total Assets	51,284,369	Total Liabilities + Equity	51,284,369

From JAN 01 2013 to SEP 30 2013 sales \$25,331,696; cost of goods sold \$6,621,481. Gross profit \$18,710,215; operating expenses \$20,791,103. Operating income \$(2,080,888); other income \$16,671; other expenses \$733,956; net income before taxes \$(2,798,173); Federal income tax \$405,165; (net loss) \$3,203,338.

Statement obtained from Securities And Exchange Commission. Prepared from books without audit.

Accounts receivable shown net less \$275,179 allowance.

Explanations

The net worth of this company includes intangibles; Other Long Term Liabilities consist of def revenue/customer advances-net and other liabs.

Additional Financial Data

Fiscal Consolidated statement dated DEC 31 2012

Assets		Liabilities	
Cash	3,307,822	Accts Pay	1,044,459
Accts Rec	7,242,603	Accrued Exps & Other Current Liabs	1,390,188
Inventory	154,626	Accruals	7,339,711
Deferred Tax Asset	1,632,280	L,T, Liab-(1yr)	1,960,127
Prepaid Exps & Other Current Assets	361,813	Deferred Revenue/Customer Advances	1,929,377
		Related Party Subordinated Note Pay	1,500,000
Curr Assets	12,699,144	Curr Liabs	15,163,862
Fixt & Equip	639,839	Long-Term Debt-Net	4,410,286
Intangible Assets-Net	19,092,998	Subordinated Note Payable	4,000,000
Goodwill	16,167,834	L.T. Liab-Other	4,846,387
Deferred Tax Asset-Net	5,844,980	COMMON STOCK	1,195
Other Assets	685,867	ADDIT. PDIN CAP	43,770,108
		TREASURY STOCK	(224,353)
		RETAINED EARNINGS	(16,836,823)
Total Assets	55,130,662	Total Liabilities + Equity	55,130,662

From JAN 01 2012 to DEC 31 2012 annual sales \$31,778,837; cost of goods sold \$10,069,357. Gross profit \$21,709,480; operating expenses \$23,410,612. Operating income \$(1,701,132); other income \$59,763; other expenses \$547,075; net income before taxes \$(2,188,444); Federal income tax \$(7,479,136). Net income \$5,290,692.

Statement obtained from Securities And Exchange Commission. Prepared from statement(s) by Accountant: Marcum LLP, Boston, Massachusetts,

ACCOUNTANTS OPINION

A review of the accountant's opinion indicated that the financial statement meets generally accepted accounting principles and the audit contains no qualifications.

Explanations

The net worth of this company includes intangibles.

Other long term liabilities consist of def rev/customer advances-net, accrued contingent consideration-net and related party subordinated notes payable-net.

The report was updated using information the company filed with the Securities and Exchange Commission.

16

Key Business Ratios

Statement Date Dec 31 2012

Based on this Number of Establishments

Industry Norms Based On 16 Establishments

<u>=</u>		and the second court court is the second	galanna ann gan gan an an an An
This Business	Industry Median	Industry Quartile	
	, , , , , , , , , , , , , , , , , , ,		3-4 (1420 4 3 4
16.6	4.9	1	
19.8	12.1	2	
0.8	1.8	4	
0.7	1.3	3	
173.5	98.6	3	
(12.9)	7,2	4.	
106.4	111.9	2	
	This Business 16.6 19.8 0.8 0.7 173.5 (12.9)	16.6 4.9 19.8 12.1 0.8 1.8 0.7 1.3 173.5 98.6 (12.9) 7.2	This Business Industry Modian Industry Quartie 16.6 4.9 1 19.8 12.1 2 0.8 1.8 4 0.7 1.3 3 173.5 98.6 3 (12.9) 7.2 4

View Snapshots

View Snapshots

Detailed Trade Risk Insight™

Detailed Trade Risk Insight provides detailed updates on over 1.5 billion commercial trade experiences collected from more than 260 million unique supplier/purchaser relationships.

Days Beyond Terms - Past 3 & 12 Months

3 months from Jan 14 to Mar 14



Dollar-weighted average of 15 payment experiences reported from 15 companies

Dollar-weighted average of 22 payment experiences reported from 21 companies

Derogatory Events Last 12 Months from Mar 13 to Feb 14

No Derogatory trade Event has been reported on this company for the past 13 Months

Total Amount Current and Past Due - 12 month trend from Mar 13 to Feb 14

Status	Mar-13	Apr-13	May-13	Jun-13	Jul-13	Aug-13	Sep-13	Oct-13	Nov-13	Dec-13	Jan-14	Feb-14
Total	84,628	87,914	131,833	135,836	199,709	250,422	327,160	278,481	228,798	183,700	158,724	0
Current	51,218	84,827	108,568	103,447	151,721	189,071	265,917	248,146	226,314	172,751	105,462	-
1-30 Days Past Due	33,138	2,630	23,066	17,599	47.838	48,915	23,267	23,181	1,901	9,939	46,775	-
31-60 Days Past Due	272	309	199	14,790	21	12,307	37,801	-	583	1,010	6,487	-
61-90 Days Past Due	-	148	-		44	44	90	7,069	-	-		-
90+ Days Past Due	-	-	-	j	85	85	85	85	-		_	

This information may not be reproduced in whole or in part by any means of reproduction.

© 2005-2014 Dun & Bradstreet, Inc.

WeildFileigy

Exhibit C-8 "Bankruptcy Information"

World Energy Solutions, Inc. has not filed for protections from creditors, reorganization or any other form of bankruptcy filings.

worldenergy.com

Werlübnergy

Exhibit C-9 "Merger Information"

Acquisitions are an important component of our business strategy. Our focus is on both our core procurement business as well as new product lines within the energy management services industry such as energy efficiency services. During 2011 we acquired Co-eXprise, Inc. ("Co-eXprise"), Northeast Energy Solutions, LLC ("NES") and GSE Consulting, LP ("GSE"). During 2012 we acquired Northeast Energy Partners, L.P. ("NEP"). These acquisitions round out the Company's capabilities in the energy efficiency arena, enable the company to enter the growing small- and medium-sized customer marketplaces, and consolidate the auction space.

Co-eXprise:

On September 13, 2011, the Company acquired certain contracts and assumed certain liabilities of the Co-eXprise, Inc. ("Co-eXprise") energy procurement business for \$4.0 million. Co-eXprise, located in Wexford Pennsylvania, is a leading provider of enterprise sourcing software solutions for discrete manufacturers, enabling companies to effectively manage sourcing activities for direct material and complex indirect spend categories.

NES:

On October 13, 2011, the Company acquired substantially all of the assets and certain obligations of Northeast Energy Solutions, LLC ("NES") for a maximum purchase price of \$4.8 million. NES, located in Cromwell Connecticut, focuses on turn-key electrical and mechanical energy efficiency measures serving commercial, industrial and institutional customers.

GSE:

On October 31, 2011, the Company acquired substantially all of the assets and certain obligations of GSE Consulting, L.P. ("GSE") for a maximum purchase price of \$13.1 million. GSE is a Texas based energy management and procurement company. The purchase price was \$8.6 million, consisting of \$3.9 million in cash, \$1.5 million in cash to pay off GSE debt, and 1.0 million shares of our Common Stock valued at \$3.2 million.

NEP:

On October 3, 2012, the Company acquired substantially all of the assets and certain obligations of Northeast Energy Partners, L.P. ("NEP") for a maximum purchase price of \$13,090,959. NEP is a Connecticut based energy management and procurement company. The purchase price was \$13,090,959, consisting of \$7,910,595 in cash, \$2 million by a promissory note of World Energy Solutions, Inc., and up to an additional \$3,180,000 in cash and shares of World Energy Solutions, Inc. common stock comprising the Earn Out.

Wellatinergy

Exhibit D-1 "Operations"

World Energy Solutions, Inc. ("World Energy" or the "Company") is an energy management services firm that applies a combination of people, process and technology to help listers manage energy as a strategic asset. The Company has developed three online auction platforms, the World Energy Exchange®, the World Green Exchange® and the World DR Exchange®. On the World Energy Exchange®, retail energy consumers (commercial, industrial and government) and wholesale energy participants (utilities, electricity retailers and intermediaries) in the United States ("listers") are able to negotiate for the purchase or sale of electricity and other energy resources from competing energy suppliers ("bidders") which have agreed to participate on the Company's auction platform. The World Energy Exchange® is used by the energy management staff to conduct auctions, analyze results, guide energy consumers through contracting, and track their contracts, sites, accounts and usage history. The team also uses the exchange's sophisticated monitoring, triggering and messaging tools to develop, support and implement comprehensive risk management strategies for our more sophisticated clients.

The Company adapted its World Energy Exchange[®] auction platform to accommodate the brokering of green power in 2003 (i.e., electricity generated by renewable resources), wholesale electricity in 2004 and certain other energy-related products in 2005. In 2007, the Company created the World Green Exchange[®] based on the World Energy Exchange[®] technology and business process. On the World Green Exchange[®], bidders and listers negotiate for the purchase or sale of environmental commodities such as Renewable Energy Certificates, Verified Emissions Reductions, Certified Emissions Reductions and Regional Greenhouse Gas Initiative ("RGGI") allowances.

In January 2010, the Company launched the World DR Exchange® to create an efficient, transparent and liquid marketplace that benefits customers and suppliers alike in the demand response ("DR") industry. The World DR Exchange® creates the industry's first online marketplace for demand response, enabling customers to source DR more efficiently and effectively, bringing together curtailment service providers and energy consumers in highly-structured auction events designed to yield price transparency, heighten competition, and maximize the energy consumers' share of demand response revenues.

In 2011, the Company unveiled the Seven Levers of Energy Management[™] - Planning, Sourcing, Risk Management, Efficiency, Sustainability, Incentives and Monitoring to help customers optimize the equation of E=P*Q-I where energy costs are a function of commodity price times quantity used minus any incentives realized.

In late 2011, World Energy Solutions, purchased Northeast Energy Solutions, LLC (NES), located in Cromwell, Connecticut. NES focuses on turn-key electrical and mechanical energy efficiency measures serving commercial, industrial and institutional customers. With the acquisition of NES, World Energy Solutions has continued to

Werlübnergy

develop and build solutions to assist our customers in managing their energy as a strategic asset.

Also, in 2011, World Energy Solutions acquired GSE Consulting, LP, a mid-market broker principally serving the Texas market and in 2012 acquired Northeast Energy Partners, L.P., a mid-market broker principally serving the Connecticut market. These acquisitions will provide us with a strong base in the growing small- and medium-sized marketplace.

World Energy Solutions experience:

- World Energy Solutions, Inc. does not trade or ever take title or ownership of natural gas, electricity or other energy commodity. World Energy Solutions, Inc. is an energy broker.
- World Energy Solutions, Inc., to date, has transacted more than \$48 billion in energy, demand response, and environmental commodities on behalf of its customers.
- World Energy Solutions, Inc. runs a procurement exchange that is supplierneutral. Our mission is to get the consumer the lowest price for their energy needs.
- World Energy Solutions, Inc. provides a toll free number (1-800-578-0718) for all customer inquiries.

Werlübnergy

Exhibit D-2 "Operations Expertise"

See the enclosed Excel spreadsheet. This spreadsheet summarizes World Energy Solutions, Inc. experience in natural gas procurement.

WendEnergy

	1						
	Operations Expertise 1/1/2002 to 7/31/	2014					
			Decatherms				
Service State	Utility	Number of Deals	Contracted				
AR	Arkansas Western Gas Company (GAS)	2	135,721				
AR	CenterPoint Energy - Arkansas (GAS)	3	1,168,800				
AZ	Southwest Gas Corporation (GAS)	3	68,489				
AZ	Unisource Energy (GAS)	2	103,654				
CA	City of Vernon (GAS)	1	27,966				
CA	Pacific Gas and Electric Company (GAS)	1	110,224				
CA	Pacific Gas and Electric Company (PGE) (29	7,573,983				
CA	San Diego Gas and Electric Co. / Sempra	5	202,603				
CA	Southern California Gas (GAS)	15	20,562,958				
CA	Southern California Gas (GAS): SOCAL Bo	4	1,658,804				
CA	Southern California Gas (GAS): Socal Ci	8	1,639,874				
CA	Wholesale- SMUD (GAS)	68	76,661,000				
CA	Wholesale- SMUD (GAS): Wholesale	5	8,502,500				
со	Atmos Energy Corporation (Greeley Gas) (1	218,410				
со	Colorado Interstate Gas Pipeline	2	9,125,000				
СО	Colorado Interstate Gas Pipeline: Cheyen	21	61,117,500				
СО	Colorado Interstate Gas Pipeline: Elk Ba	1	1,510,000				
со	Colorado Interstate Gas Pipeline: Uintah	4	12,775,000				
со	Colorado Interstate Gas Pipeline: Wind R	1	7,300,000				
со	Wholesale (GAS)	38	102,247,500				
CO	Xcel Energy/Public Service Co. Colorado	10	1,057,839				
СТ	Connecticut Natural Gas (GAS)	30	1,278,382				
СТ	Multiple CT (CNG & SCG) (GAS)	2	5,847				
СТ	Southern Connecticut Gas Co (GAS)	21	476,247				
СТ	Yankee Gas (GAS)	209	8,867,271				
DC	Washington Gas Light Company (GAS)	24	12,920,396				
DE	Delmarva Power Company (GAS)	1	187,368				
FL	Central Florida Gas (GAS)	2	77,403				
FL	Florida City Gas (GAS)	1	5,912				
FL.	TECO Peoples Gas (GAS)	16	389,804				
FL	TECO Peoples Gas (GAS): TECO	1	73,810				
GA	Albany Water Gas & Light Commission (GAS	3	9,062,346				
GA	Atlanta Gas & Light Company (GAS)	13	3,649,844				
GA GA	Atlanta Gas & Light Company (GAS): Atlan	3	161,688				
GA	Atlanta Gas & Light Company (GAS): Augus	9	418,187				
GA GA	Sonat Zone 3 Direct Connect (GAS)	6	2,187,200				
IA	Alliant Energy & Mid American Energy (G	2	3,212,535				
IA	Aquila Inc. (GAS)	1	20,451				
IA	MidAmerican Energy Company (GAS)	10	1,603,376				
IA	Northern Border Pipeline: Will County	10	2,675,000				
IA	Northern Natural Gas Pipeline: Vent POI#	3	9,027,000				
IA IA	Wholesale- Xcel NSP (GAS)	8	14,699,800				

Weilublicigy

		JJ	
IA	Wholesale- Xcel NSP (GAS): Wholesale	4	1,519,000
ID	Intermountain Gas Company (GAS)	1	519,098
ID	Wholesale (GAS)	1	930,000
<u>IL</u>	Ameren CILCO (GAS)	1	215,745
<u>IL</u>	Ameren CIPS (GAS)	2	198,837
IL	Ameren IP (GAS)	2	352,456
IL	Ameren IP (GAS): Delivery Zone 1	1	6,630
<u>IL</u>	NGPL: Chicago Citygate NSG's CDP 9254	1	155,000
	NGPL: Gulf Mainline PIN 10558	_ 1	1,510,000
IL	NGPL: Midcontinent PIN 25078	7	8,909,000
L	NGPL: Texok Gulf Coast PIN 10560	2	3,020,000
IL	Nicor Gas (GAS)	30	10,438,630
IL.	Northern Border Pipeline: Ventura (DRN 1	14	9,708,154
IL.	Peoples Gas Light and Coke Company (GAS)	18	5,082,698
IL	Wholesale - Ameren IL (GAS)	1	755,000
ĮN	Citizens Gas and Coke Utility (GAS)	1	50,372
IN	NIPSCO (GAS)	24	1,385,526
1N	Vectren Energy Delivery - Indiana (GAS)	42	9,546,430
KS	Black Hills Energy (GAS)	3	568,260
KS	Kansas Gas Service (GAS)	12	2,037,291
ΚΥ	Atmos Energy (GAS)	1	13,618
KY	Columbia Gas of Kentucky Inc (GAS)	2	1,091,674
Кү	Columbia Gas of Kentucky Inc (GAS): OP6-	26	13,848
КҮ	Delta Natural Gas Company, Inc. (GAS)	3	419,453
Кү	Duke Energy (GAS)	4	1,083,162
КУ	Multiple Ohio (GAS)	1	260,555
КҮ	Scottsville Water & Gas Dept. (GAS)	1	190,862
КҮ	Tennesse Z2 - Mavity (GAS)	100	9,303,290
LA	Crosstex (GAS)	3	1,261,580
LA	Florida Gas Transmission (FGT) Pipeline:	9	10,351,980
LA	Wholesale- OUC (GAS)	5	4,905,180
MA	Berkshire Gas Company (GAS)	5	491,044
MA	Boston Gas Company-Keyspan (NGrid) (GAS)	9	4,149,424
MA	Colonial Gas-Keyspan (NGrid) (GAS)	1	21,438
MA	Columbia Gas of Massachusetts (GAS)	161	6,349,491
MA	Duquesne Light (ELE)	4	13,259
MA	Essex County Gas -Keyspan (NGrid) (GAS)	1	2,838
MA	Keyspan Energy Delivery (National Grid)	105	7,772,784
MA	Multiple MA-gas (GAS)	1	87,678
MA	National Grid - Massachusetts Electric (2	3,636
MA	New England Gas Company (NEGASCO) (GAS)	1	41,086
MA	NSTAR Electric and Gas Corporation (ELE)	2	8,671
MA	NSTAR Electric and Gas Corporation (GAS)	53	1,476,350
MA	NU-Western Massachusetts Electric Co (EL	5	48,066
MA	PPL Utilities (ELE)	1	1,947
MA	UI & CL&P (ELE)	9	82,295
MA	Unitil - Fitchburg Gas & Electric (GAS)	2	331,997
MA	Wholesale - MBTA (GAS)	1	784,471
MD	Baltimore Gas & Electric Company (GAS)	19	2,160,477

WerldEnergy

MD	Columbia Gas of Maryland Inc. (GAS)	1	269,910
MD	Washington Gas & Light Company (GAS)	12	9,688,317
ME	Unitil - Northern Utilities (GAS)	2	279,180
МІ	Consumers Energy Company (GAS)	27	3,247,019
MI	Detroit Edison (GAS)	2	534,780
MI	DTE - MichCon (GAS)	7	772,388
MI	Michigan Consolidated Gas Company (GAS)	4	500,000
MI	Michigan Gas Utilities (MGU (GAS)	2	96,924
MI	Panhandle Pipeline Direct Connect (GAS)	_ 1	5,131,304
МІ	SEMCO ENERGY (GAS)	4	1,108,828
MN	Aquila Networks(MN Energy Resources Corp	3	244,124
MN	Minnesota Gas Co (GAS)	1	13,840
MN	Northern Border Pipeline: Welcome (POI #	1	186,000
MN	Northern Natural Gas Pipeline: Aberdeen	1	781,727
MN	Northern Natural Gas Pipeline: Carlton (3	9,362,000
MN	Northern Natural Gas Pipeline: Demarc PO	15	10,970,000
MN	Northern Natural Gas Pipeline: Vent POI#	6	2,488,000
MN	Northern Natural Gas Pipeline: Welcome (6	1,818,321
MN	Viking Gas Transmission Pipeline: Emerso	5	5,286,800
MN	Wholesale (GAS)	6	1,860,000
MN	Wholesale (GAS): Wholesale	6	8,778,000
MN	Wholesale- CNPT Minnegasco (GAS)	18	21,222,755
MO	Empire District Electric Co (GAS): Empir	2	269,388
MO	Laclede Gas Company (GAS)	7	1,038,857
МО	Liberty Utilities (GAS)	2	134,537
MO	Missouri Gas Energy (GAS)	4	1,343,331
MO	NGPL Direct Connect (GAS)	3	10,517,614
МО	Wholesale- Ameren MO (GAS)	2	1,290,000
MS	Atmos Energy (GAS)	3	3,453,937
MT	Montana-Dakota Utilites, Inc. (GAS)	1	42,123
MT	Northwestern Energy (GAS)	1	112,440
NC	Frontier Energy (GAS)	2	375,638
NC	Piedmont Natural Gas (GAS)	10	8,261,587
NC	Public Service Company of NC (GAS)	6	1,284,246
NE	Source Gas (GAS)	4	1,032,877
NH	Liberty Utilities (GAS)	1	75,947
NJ	Elizabeth Town Gas (GAS)	2	1,290,641
NJ	New Jersey Natural Gas (NJNG) (GAS)	5	1,478,776
NJ	NUI Elizabethtown Gas- NJ (GAS)	1	17,782
NJ	Public Service Electric and Gas (PSEG) (47	4,903,637
NJ	South Jersey Gas Company (SJG) (GAS)	8	6,312,114
NJ	Transcontinental Gas Pipeline: South Jer	1	3,775,000
NJ	Wholesale (GAS)	1	912,500
NM	New Mexico Gas Company (GAS)	2	133,004
NV	NV Energy (GAS)	1	69,472
NY	Bath Electric, Gas & Water Systems (GAS)	2	274,437
NY	Central Hudson Gas & Electric Corp. (GAS	2	141,720
NY	Consolidated Edison Company of New York	13	3,350,085
NY	Dominion South Point (GAS)	1	604,000

WendEnergy

		<u> </u>	
NY	Keyspan Energy Delivery , LI (NGrid) (GA	1	28,431
NY	Keyspan Energy Delivery, NY (N Grid) (GA	3	1,050,311
NY	National Fuel Gas Distribution Corp. (GA	17	1,389,203
NΥ	National Grid - Niagara Mohawk (GAS)	21	6,629,185
NY	New York State Electric & Gas Corp. (GAS	12	5,515,910
NΥ	Orange and Rockland Utilities, Inc. (GA	10	8,621,620
NY	Rochester Gas and Electric Corporation	1	1,206
NY	Tennessee Gas Pipeline: Niagara (NY) / Z	1	1,359,000
NY	Transcontinental Gas Pipeline: Leidy (PA	1	1,661,000
NY	Wholesale- ConEd ORU (GAS)	14	22,719,500
ОН	AK Steel - Middletown Works (GAS)	204	9,740,302
Он	ANR Glen Karn - PEPL GAs City (GAS)	23	11,393,647
ОН	Columbia Gas of Ohio (GAS)	3	44,320
ОН	Columbia Gas of Ohio Inc. (GAS)	103	15,344,843
ОН	Columbia Gas of Ohio Inc. (GAS): Columbi	1	6,629
ОН	Columbia Gas of Ohio Inc. (GAS): OP5-MA2	7	398,848
ОН	Columbia Gas of Ohio Inc. (GAS): OP5-MA7	4	650,160
ОН	Columbia Gas of Ohio Inc. (GAS): OP7-MA1	94	2,370,923
OН	Columbia Gas of Ohio Inc. (GAS): OP7-MA3	2	132,675
ОН	Columbia Gas of Ohio Inc. (GAS): OP7-MA5	29	1,840,857
ОН	Columbia Gas of Ohio Inc. (GAS): OP7-MA6	2	202,014
ОН	Columbia Gas of Ohio Inc. (GAS): OP7-MA8	64	3,275,769
ОН	Columbia Gas of Ohio Inc. (GAS): OP7-MA9	86	2,370,018
ОН	Dominion East Ohio (GAS)	117	11,936,484
ОН	Dominion East Ohio (GAS): Dominion East	8	606,230
ОН	Dominion West Ohio Gas (GAS)	11	2,533,917
ОН	Duke Energy-Ohio (GAS)	121	4,631,923
OН	Multiple Ohio (GAS)	29	63,423,374
ОН	Ohio Gas Company (GAS)	4	2,887,950
ОН	Pike Natural Gas Company (GAS)	1	560,391
ОН	VECTREN (Formerly Dayton Power & Light)	3	1,161,780
ОН	Vectren Energy Delivery of Ohio (GAS)	15	708,109
ОН	Wholesale- COH (GAS)	1	35,000
ОН	Wholesale- Dominion (GAS): Wholesale	1	35,000
ОН	Wholesale- Vectren (GAS): Wholesale	1	35,000
Ок	Oklahoma Natural Gas Company (GAS)	24	752,238
QN	Enbridge Gas Distribution (GAS)	2	132,130
QN	Union Gas North (GAS)	2	421,940
OR	Northwest Natural Gas Company (GAS)	1	1,490,844
PΑ	Columbia Gas of Pennsylvania Inc. (GAS)	19	662,135
PΑ	Columbia Gas of Pennsylvania Inc. (GAS):	1	262,848
PΑ	Dominion Peoples (GAS)	5	282,718
PΑ	Equitable Gas Company (GAS)	2	671,258
PA	Equitrans Direct Connect (GAS)	1	2,620,968
PΑ	National Fuel Gas Distribution Corporati	2	177,126
PA	PECO (GAS)	32	2,247,262
PΑ	PECO (GAS): PECO	1	53,370
PΑ	Philadelphia Gas Works (GAS)	8	1,216,890
PA	PPL Utilities (GAS)	2	291,490

WendEnergy

PA	UGI Central Penn Gas (GAS)	3	62,281
PA	UGI Penn Natural Gas (GAS)	16	81,348,473
PA	UGI Utilities, Inc. (GAS)	14	1,695,851
PA	Wholesale - TOTAL G&P (GAS)	2	2,140,000
RI	National Grid (Rhode Island) (GAS)	_45	16,975,292
SC	Piedmont Natural Gas (GAS)	1	235,722
TN	Atmos Energy (GAS)	1	57,600
TN	Chattanooga Gas Company (GAS)	2	62,010
TN	Jackson Energy Authority (GAS)	1	1,005,002
TN	Memphis Light Gas & Water (GAS)	4	256,075
TN	Piedmont (GAS)	1	10,650
TX	Atmos Energy (GAS)	14	3,963,722
TX	CenterPoint Energy - Texas (GAS)	21	36,913,266
TX	Texas Gas Service- Oneok (GAS)	4	593,456
TX	Wholesale (GAS)	19	26,593,800
TX	Wholesale- CNPT South (GAS)	40	33,979,000
UT	Questar Gas (GAS)	13	3,251,094
VA	Columbia Gas of Virginia Inc. (GAS)	1	251,619
VA	Virginia Natural Gas (GAS)	1	360,477
VA	Washington Gas Light Company (GAS)	5	85,820
WA	Avista Utilities (GAS)	1	111,938
WA	Cascade Natural Gas Company (GAS)	10	815,891
WA	Puget Sound (GAS)	2	146,300
WI	Alliant/Wisconsin Power & Light (GAS)	4	344,718
WI	Multiple WI (GAS)	_1	325,551
WI	WE Energies (GAS)	8	1,812,000
WI	Wisconsin Gas Company (GAS)	1	284,586
WI	Wisconsin Public Service Corp (GAS)	9	11,622,369
WV	Dominion Hope Gas Inc (GAS)	4	699,361
WV	Mountaineer Gas (GAS)	4	1,217,050
Grand Total		2,931	1,055,339,600

Wendenergy

Exhibit D-3 "Key Technical Personnel"

Daniel P. Pugh, CEP – Dan Pugh is Vice President of Supply Management for World Energy's commercial and industrial business. He is responsible for all new and existing supplier relationships, customer-specific energy purchasing and daily operational processes to manage all supply contracts. Dan has over 20 years of experience in the energy industry, providing natural gas marketing and management services for large end users. Dan's experience includes major account procurement, contracting, transportation, risk management and storage arbitrage. Dan was an Energy Trader for AEP Energy Services for two years and prior to that was Market Coordinator with Enron Energy Services. Dan graduated Magna Cum Laude with a Bachelor's degree in Business Administration from The Ohio State University.

(614) 553-6735 dpugh@worldenergy.com

Sean Zurbrick – Sean Zurbrick is Supply Director within World Energy's Supply Management group. He is responsible for supplier offer coordination along with daily procurement and operational processes for consumers. In addition, all natural gas analysts, as well as the procurement services group, report to Sean. Prior to World Energy, Sean spent two years at Integrys Energy Services and 5 years at Clinton Energy Services. In these positions he was responsible for procurement, scheduling, tariff analysis, procurement and management of pipeline capacity, least cost routing analysis and storage and capacity management. Sean holds a Bachelor's degree from The Ohio State University.

(614) 553-6737 szubrick@worldenergy.com

Bill Yeager, CEP — Bill Yeager is a Supply Director within World Energy's Supply Management group and has more than 27 years' experience in the energy industry, with focused knowledge on risk management, procurement, pipeline and utility logistics, tariffs and customer service. At World Energy, Bill's responsibilities include daily procurement, supplier offer coordination and operational processes for consumers. Bill works with customers to explain the effects of energy prices and the ways to get out in front of the risks associated with purchasing energy. Prior to World Energy, Bill worked for FirstEnergy Solutions American Electric Power Energy Services Enron Energy Services, and Ohio Gas Marketing. Bill holds a Bachelor of Science in Agricultural Economics from The Ohio State University.

Roberto Martinez, CFA — Rob Martinez is a Supply Director within World Energy's Supply Management group. He has over fifteen years' experience in the energy industry. He began his career with Enron Corporation in 1998. Rob was in the Analyst and Associate programs and held positions in various groups including Enron Capital and Trade and Enron Energy Services Power Structuring. In early 2002, Rob joined the Mergers and Acquisitions group at Marathon Oil Corporation. Rob worked on buying and selling upstream oil and gas assets and companies. In late 2005, Rob joined Accent Energy as Power and Gas Structuring Manager. Rob priced and structured Mid-Market Power and Gas deals. Rob joined World Energy in December of 2008. Rob holds a Bachelor of Business Administration degree from the University of Notre Dame

Wellariergy

Mendoza College of Business. Rob is also a CFA Charterholder® (CFA) and a member of the CFA Institute and the CFA Society of Columbus (OH). (614) 553-6733 rmartinez@worldenergy.com

Tony Barnhart, Senior Vice President of Energy Sales - As Senior Vice President of Energy Sales, Tony is responsible for leading World Energy's Commercial, Industrial, Institutional, Wholesale, and Mid-Market energy sales efforts through both direct and partner channels. Since joining World Energy in 2009, he has held various sales and sales management positions. During this time, the Company has tripled its core procurement business, significantly broadened its product and service offerings to the marketplace, and successfully integrated three procurement related acquisitions.

Tony has over 27 years of sales and sales management experience and has spent 23 years in the energy industry. Prior to World Energy, he worked with Access Energy, Clinton Energy Management, Enron Energy Services, and Accent Energy. With these companies, Tony has held senior roles in sales management, business development, business start-up, national accounts, regulatory, and mergers & acquisitions. Tony holds a B.A. in Transportation & Logistics from The Ohio State University.

(614) 553-6713 tbarnhart@worldenergy.com